# EIGHTH AMENDED AND RESTATED LIMITED LIABILITY COMPANY AGREEMENT

**OF** 

MEMX HOLDINGS LLC

Dated as of [\_\_\_\_]

# **TABLE OF CONTENTS**

		Page
ARTICLE I	. DEFINITIONS	2
1.1	Definitions	2
1.2	Interpretation	
ARTICLE I	I. ORGANIZATION	27
2.1	Formation; Agreement.	27
2.2	Name	
2.3	Principal Office	28
2.4	Registered Office; Registered Agent	28
2.5	Purpose; Powers	
2.6	Term	29
2.7	No State-Law Partnership	
ARTICLE I	II. UNITS; OWNERSHIP LIMITATIONS	29
3.1	Units Generally	29
3.2	Authorization and Issuance of Class A Units, Class C Units, Class D U	
	and Common Units	
3.3	Service Provider Equity Pool; Class B Units	31
3.4	Other Issuances	
3.5	Limitations on Ownership	
3.6	Notices of Ownership	34
3.7	Certain Prohibited Actions	35
3.8	Certain Special Transfer Rights	36
3.9	Restricted Economic Elections.	36
3.10	Specification of Maximum Voting Percentage	37
3.11		42
3.12	Certification of Units	43
ARTICLE I	V. MEMBERS	44
4.1	Admission of New Members	44
4.2	Representations and Warranties of Members	44
4.3	No Personal Liability	
4.4	No Withdrawal	45
4.5	Death	46
4.6	No Voting Rights or Power to Bind the Company	46
4.7	Voting Rights of Members; Meetings of Members	46
4.8	No Interest in Company Property	

ARTICLE V	. CAPITAL CONTRIBUTIONS; CAPITAL ACCOUNTS	50
5.1	Capital Contributions and Unit Issuances of the Class A Members, Class C	
	Members and Class D Members.	50
5.2	Maintenance of Capital Accounts	51
5.3	Succession Upon Transfer	51
5.4	Negative Capital Accounts	51
5.5	No Withdrawal	51
5.6	Treatment of Loans From Members	52
5.7	Modifications	52
ARTICLE V	I. ALLOCATIONS	52
6.1	Allocation of Net Income and Net Loss	52
6.2	Regulatory and Special Allocations	52
6.3	Tax Allocations	53
6.4	Allocations in Respect of Transferred Units	54
6.5	Curative Allocations	54
ARTICLE V	II. DISTRIBUTIONS	54
7.1	Tax Advances	
7.2	General	
7.3	Priority of Distributions	
7.4	Tax Withholding; Withholding Advances; Tax Election	
7.5	Distributions in Kind	58
ARTICLE V	III. MANAGEMENT	59
8.1	The Board	59
8.2	Authority of Board	59
8.3	Board Composition; Vacancies	59
8.4	Removal; Resignation	60
8.5	Meetings	61
8.6	Quorum; Manner of Acting	61
8.7	Action By Written Consent	63
8.8	Compensation; No Employment	
8.9	Committees	64
8.10	Loss or Transfer of Right to Nominate a Director; Grant of Right to Nominate a Director	61
8.11	Waiver of Right to Nominate a Director	
8.12	Alternate Directors	
8.13	Board Observers	
8.14	Officers	
8.15	No Personal Liability	
(), 1.)	1 10 1 VIOUINI L/INVIIII V	

8.16	Company Related Party Transactions	68
8.17	Combination of Nominating Members.	
8.18	Governance of Company Subsidiaries; Certain Agreements Related to the	
	Exchange Board	69
8.19	Industry Advisory Board	76
ARTICLE I	X. PRE-EMPTIVE RIGHTS	77
9.1	Pre-emptive Right	77
ARTICLE X	T. TRANSFER	80
10.1	General Restrictions on Transfer	80
10.2	Permitted Transfers	82
10.3	Right of First Offer	
10.4	Drag-along Rights	
10.5	Tag-along Rights	
10.6	Regulatory Hardship Transfers; Surrender Right	
10.7	Transfers of Class B Units	95
ARTICLE X	II. ADDITIONAL AGREEMENTS OF THE PARTIES	95
11.1	Confidentiality	95
11.2	Other Activities of Members	
11.3	Compliance with Law	98
11.4	Antitrust Practices and Guidelines	
11.5	Initial Public Offering	
11.6	Use of Name and Trade Marks	
11.7	Certain Regulatory Matters	104
ARTICLE X	III. INFORMATION RIGHTS; ACCOUNTING; TAX MATTERS	104
12.1	Information Rights	104
12.2	Inspection Rights; Books and Records	106
12.3	Partnership Representative	
12.4	Annual Budget	
12.5	Tax Returns	
12.6	Company Funds	109
ARTICLE X	III. DISSOLUTION AND LIQUIDATION	110
13.1	Events of Dissolution	
13.2	Effectiveness of Dissolution	110
13.3	Liquidation	
13.4	Cancellation of Certificate	113

13.5	Survival of Rights, Duties and Obligations	113
13.6	Recourse for Claims	113
ARTICLE XI	V. EXCULPATION AND INDEMNIFICATION	114
14.1	Exculpation of Covered Persons	114
14.2	Liabilities and Duties of Certain Persons	
14.3	Indemnification	
14.4	Survival	118
ARTICLE XV	V. MISCELLANEOUS	118
15.1	Expenses	118
15.2	Further Assurances	
15.3	Notices	118
15.4	Headings	119
15.5	Severability	
15.6	Entire Agreement	119
15.7	Successors and Assigns	
15.8	No Third Party Beneficiaries	
15.9	Amendments	119
15.10	Waiver	120
15.11	Governing Law	120
15.12	Submission to Jurisdiction	
15.13	Waiver of Jury Trial	121
15.14		
15.15	•	
15.16	Remedies Cumulative	
15.17	Counterparts	122
15.18	Restrictions on Non-U.S. Operations	

EXHIBIT A – FORM OF ADHERENCE AGREEMENT	. A-1
EXHIBIT B – DIRECTORS AND OBSERVERS SCHEDULE	B-1
EXHIBIT C – BOARD MATTERS	C-1
EXHIBIT D – ANTITRUST GUIDELINES	. D-1
EXHIBIT E – FORM OF RESTRICTED ECONOMIC ELECTION NOTICE	E-1
EXHIBIT F – RESTRICTED VOTING ELECTION NOTICE	F-1
EXHIBIT G - CONVERSION RIGHTS OF CLASS C UNITS AND CLASS D UNITS	. G-1
EXHIBIT H – NOMINATION WAIVER	. H-1
EXHIBIT I-1 –MEMX LLC EXCHANGE DIRECTOR NOMINATION ROTATION	I-1
EXHIBIT I-2 –MX2 LLC EXCHANGE DIRECTOR NOMINATION ROTATION	I-1

# EIGHTH AMENDED AND RESTATED LIMITED LIABILITY COMPANY AGREEMENT OF MEMX HOLDINGS LLC

This Eighth Amended and Restated Limited Liability Company Agreement of MEMX Holdings LLC, a Delaware limited liability company (the "Company"), is entered into as of [\_\_\_\_\_\_] (the "Effective Date"), by and among the Company, the Members and each other Person who after the date hereof becomes a Member of the Company and becomes a party to this Agreement by executing an Adherence Agreement.

#### RECITALS

WHEREAS, (a) the Company was formed under the laws of the State of Delaware as MembersX Holdings LLC by the filing of a certificate of formation with the Secretary of State of the State of Delaware on September 6, 2018, (b) an amended and restated certificate of formation (the "Certificate of Formation") changing the name of the Company to MEMX Holdings LLC was filed with the Secretary of State of the State of Delaware on January 22, 2019, and (c) on September 17, 2018, the original members of the Company entered into a Limited Liability Company Agreement of the Company effective as of September 17, 2018 (the "Original LLC Agreement");

WHEREAS, the Original LLC Agreement was amended and restated as set forth in the First Amended and Restated Limited Liability Agreement of the Company dated December 14, 2018 (the "<u>First Amended LLC Agreement</u>"); and

WHEREAS, the First Amended LLC Agreement was amended and restated as set forth in the Second Amended and Restated LLC Agreement of the Company dated May 7, 2019 (the "Second Amended LLC Agreement"); and

WHEREAS, the Second Amended LLC Agreement was amended and restated as set forth in the Third Amended and Restated LLC Agreement of the Company dated September 5, 2019 (the "Third Amended LLC Agreement"), which became effective on October 31, 2019; and

WHEREAS, the Third Amended LLC Agreement was amended and restated as set forth in the Fourth Amended and Restated LLC Agreement of the Company dated February 19, 2020, which was subsequently amended by the Amendment No. 1 to the Fourth Amended and Restated LLC Agreement, which became effective on July 17, 2020 (as amended, the "Fourth Amended LLC Agreement"); and

WHEREAS, the Fourth Amended LLC Agreement was amended and restated as set forth in the Fifth Amended and Restated LLC Agreement of the Company, which became effective on April 5, 2021 (the "Fifth Amended LLC Agreement"); and

WHEREAS, the Fifth Amended LLC Agreement was amended and restated as set forth in the Sixth Amended and Restated LLC Agreement of the Company which became effective as of December 3, 2021 (the "Sixth Amended LLC Agreement");

WHEREAS, the Sixth Amended LLC Agreement was amended and restated as set forth in the Seventh Amended and Restated LLC Agreement of the Company; which became effective on March 17, 2023; and

WHEREAS, pursuant to Section 15.9(b) of the Seventh Amended LLC Agreement, and subject to the provisions of Section 15.9(a) of the Seventh Amended LLC Agreement, the Board (as defined below) desires to amend and restate the Seventh Amended LLC Agreement as of the Effective Date to, among other things, provide for the creation of additional Regulated Securities Exchange Subsidiaries (as defined below).

NOW, THEREFORE, in consideration of the mutual covenants and agreements hereinafter set forth and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto agree as follows:

# ARTICLE I. DEFINITIONS

- 1.1 <u>Definitions</u>. Capitalized terms used herein and not otherwise defined shall have the meanings set forth in this <u>Section 1.1</u>:
  - "Acceptance Notice" has the meaning set forth in Section 9.1(e).
  - "Acting Member" has the meaning set forth in Section 11.7.
- "Adherence Agreement" means the adherence agreement to this Agreement substantially in the form attached hereto as Exhibit A.
- "Adjusted Capital Account Deficit" means, with respect to any Member, the deficit balance, if any, in such Member's Capital Account as of the end of the relevant Fiscal Year, after giving effect to the following adjustments:
- (a) crediting to such Capital Account any amount which such Member is obligated to restore or is deemed to be obligated to restore pursuant to Treasury Regulations Sections 1.704-1(b)(2)(ii)(c), 1.704-2(g) (1) and 1.704-2(i); and
- (b) debiting to such Capital Account the items described in Treasury Regulation Section 1.704-1(b) (2)(ii)(d)(4), (5) and (6).
- "Affiliate" means, with respect to any Person, any other Person who, directly or indirectly (including through one or more intermediaries), Controls, is Controlled by, or is under common Control with, such Person. The term "Affiliated" shall have a correlative meaning. Notwithstanding the foregoing, for the purposes of this Agreement no Member shall be deemed to be an Affiliate of the Company or any other Member (unless such Members are members of the same Affiliated group of entities) merely by virtue of the fact that it is a Member of the Company.
  - "Affiliated Institution" has the meaning set forth in Section 14.3(g).

"Aggregate Liquidation Proceeds" has the meaning set forth in Section 13.3(e).

"Agreement" means this Eighth Amended and Restated Limited Liability Company Agreement of MEMX Holdings LLC, as executed and as it may be amended, modified, supplemented or restated from time to time, as provided herein.

"Alternate Director" has the meaning set forth in Section 8.12(a).

"Annual Budget" has the meaning set forth in Section 12.4(a).

"Antitrust Counsel" has the meaning set forth in Section 11.4(a).

"Applicable Law" means all applicable provisions of (a) constitutions, treaties, statutes, laws (including the common law), rules, regulations, decrees, ordinances, codes, proclamations, declarations or orders of any Governmental Authority; (b) any consents or approvals of any Governmental Authority; and (c) any orders, decisions, advisory or interpretative opinions or guidance, injunctions, writs, judgments, awards, decrees of, or agreements with, any Governmental Authority.

"Bank Director" means a Director nominated by a Bank Member.

"Bank Holding Company" shall have the meaning ascribed to the term "bank holding company" in the BHCA.

"Bank of America" means Banc of America Strategic Investments Corporation, together with its Affiliates that hold Units. For the sake of clarity, (a) Banc of America Strategic Investments Corporation and its Affiliates holding Units (if any) shall be deemed to be one (1) Member, one (1) Class A Member, one (1) Class C Member, one (1) Nominating Member, and one (1) holder of Units of each applicable type, class or series that Banc of America Strategic Investments Corporation and/or its Affiliates hold, (b) Banc of America Strategic Investments Corporation (for so long as it holds Units) shall be entitled to receive notices on behalf of itself and its Affiliates that hold Units, and shall be entitled to take all actions under this Agreement with respect to itself and its Affiliates that hold Units, and (c) at such time as Banc of America Strategic Investments Corporation no longer holds any Units, Banc of America Strategic Investments Corporation shall designate an Affiliate that then holds Units as the Member entitled to receive notices and take actions on behalf of itself and its Affiliates pursuant to written notice to the Company.

"Bank Member" means each of Bank of America, Morgan Stanley, UBS, JPMorgan, Goldman Sachs, Wells Fargo, Citi and any other Member that is specifically designated as a Bank Member, in each case, together with each of their respective Affiliates. For the avoidance of doubt, no Bank Member shall be deemed a Market Maker Member, a Buy Side Member or a Retail Broker Member, and no Market Maker Member, Buy Side Member or Retail Broker Member shall be deemed a Bank Member for the purposes of this Agreement.

"Bankruptcy" means, with respect to a Member, the occurrence of any of the following: (a) the filing of an application by such Member for, or a consent to, the appointment of a trustee of such Member's assets; (b) the filing by such Member of a voluntary petition in bankruptcy or the filing of a pleading in any court of record admitting in writing such Member's inability to pay its debts as they come due; (c) the making by such Member of a general assignment for the benefit of such Member's creditors; (d) the filing by such Member of an answer admitting the material allegations of, or such Member's consenting to, or defaulting in answering a bankruptcy or other insolvency petition filed against such Member in any bankruptcy or other insolvency proceeding; or (e) the expiration of sixty (60) days following the entry of an order, judgment or decree by any court of competent jurisdiction adjudicating such Member a bankrupt or insolvent or appointing a trustee of such Member's assets.

"BBA" has the meaning set forth in Section 12.3(a).

"Beneficially owned", "own beneficially" or any derivative thereof has the meaning set forth in Rule 13d-3 under the Exchange Act.

"BHCA" means the United States Bank Holding Company Act of 1956, as amended and in effect from time to time, and the rules and regulations promulgated thereunder.

"BlackRock" means BLK SMI, LLC, a Delaware limited liability company, together with its Affiliates that hold Units. For the sake of clarity, (a) BLK SMI, LLC and its Affiliates holding Units (if any) shall be deemed to be one (1) Member, one (1) Class A Member, one (1) Nominating Member, and one (1) holder of Units of each applicable type, class or series that BLK SMI, LLC and/or its Affiliates hold, (b) BLK SMI, LLC (for so long as it holds Units) shall be entitled to receive notices on behalf of itself and its Affiliates that hold Units, and shall be entitled to take all actions under this Agreement with respect to itself and its Affiliates that hold Units, and (c) at such time as BLK SMI, LLC no longer holds any Units, BLK SMI, LLC shall designate an Affiliate that then holds Units as the Member entitled to receive notices and take actions on behalf of itself and its Affiliates pursuant to written notice to the Company.

"Board" has the meaning set forth in Section 8.1.

"Board Observer" has the meaning set forth in Section 8.13(a).

"Book Depreciation" means, with respect to any Company asset for each Fiscal Year, the Company's depreciation, amortization, or other cost recovery deductions determined for federal income tax purposes, except that if the Book Value of an asset differs from its adjusted tax basis at the beginning of such Fiscal Year, Book Depreciation shall be an amount which bears the same ratio to such beginning Book Value as the federal income tax depreciation, amortization, or other cost recovery deduction for such Fiscal Year bears to such beginning adjusted tax basis; provided, that if the adjusted basis for federal income tax purposes of an asset at the beginning of such Fiscal Year is zero and the Book Value of the asset is positive, Book Depreciation shall be determined with reference to such beginning Book Value using any permitted method selected by the Board in accordance with Treasury Regulation Section 1.704-1(b)(2)(iv)(g)(3).

"Book Value" means, with respect to any Company asset, the adjusted basis of such asset for federal income tax purposes, except as follows:

- (a) the initial Book Value of any Company asset contributed by a Member to the Company shall be the gross Fair Market Value of such Company asset as of the date of such contribution;
- (b) immediately prior to the Distribution by the Company of any Company asset to a Member, the Book Value of such asset shall be adjusted to its gross Fair Market Value as of the date of such Distribution:
- (c) the Board may adjust the Book Value of all Company assets to equal their respective gross Fair Market Values, as determined by the Board, as of the following times:
  - (i) the acquisition of an additional Membership Interest in the Company by a new or existing Member in consideration of a Capital Contribution of more than a de minimis amount;
  - (ii) the Distribution by the Company to a Member of more than a de minimis amount of property (other than cash) as consideration for all or a part of such Member's Membership Interest in the Company;
  - (iii) the grant to an Officer, employee, consultant or other service provider of the Company of any Units as a profits interest; and
  - (iv) the liquidation of the Company within the meaning of Treasury Regulation Section 1.704-1(b)(2)(ii)(g); <u>provided</u>, that adjustments pursuant to clauses (i), (ii) and (iii) above need not be made if the Board reasonably determines that such adjustment is not necessary or appropriate to reflect the relative economic interests of the Members and that the absence of such adjustment does not adversely and disproportionately affect any Member; <u>provided</u>, <u>however</u>, that such adjustment shall be mandatory to the extent required by Treasury Regulation Section 1.704-1(b)(2)(iv);
- (d) the Book Value of each Company asset shall be increased or decreased, as the case may be, to reflect any adjustments to the adjusted tax basis of such Company asset pursuant to Code Section 734(b) or Code Section 743(b), but only to the extent that such adjustments are taken into account in determining Capital Account balances pursuant to Treasury Regulation Section 1.704-1(b)(2)(iv)(m); provided, that Book Values shall not be adjusted pursuant to this paragraph (d) to the extent that an adjustment pursuant to paragraph (c) above is made in conjunction with a transaction that would otherwise result in an adjustment pursuant to this paragraph (d); and
- (e) if the Book Value of a Company asset has been determined pursuant to paragraph (a) or adjusted pursuant to paragraphs (c) or (d) above, such Book Value shall thereafter

be adjusted to reflect the Book Depreciation taken into account with respect to such Company asset for purposes of computing Net Income and Net Losses.

"Business Day" means a day other than a Saturday, Sunday or other day on which commercial banks in the City of New York are authorized or required to close.

"Buy Side Director" means a Director nominated by a Buy Side Member.

"Buy Side Member" means each of BlackRock and any other Member that is specifically designated as a Buy Side Member, in each case, together with each of their respective Affiliates. For the avoidance of doubt, no Buy Side Member shall be deemed a Market Maker Member, a Bank Member or a Retail Broker Member, and no Market Maker Member, Bank Member or Retail Broker Member shall be deemed a Buy Side Member for the purposes of this Agreement.

"Capital Account" has the meaning set forth in Section 5.2.

"Capital Contribution" means, for any Member, the total amount of cash and cash equivalents and the Book Value of any property contributed to the Company by such Member.

"CEO" has the meaning set forth in Section 8.3(c).

"Certificate of Formation" has the meaning set forth in the Recitals.

"Chairman of the Board" means the chairman of the Board of the Company.

"Change of Control" means: (a) the sale or Transfer (in a single transaction or a series of related transactions) of all or substantially all of the consolidated assets of the Company and the Company Subsidiaries to a Third Party Purchaser; (b) a sale or Transfer (in a single transaction or a series of related transactions) resulting in no less than a majority of the Units on a Fully Diluted Basis being held by a Third Party Purchaser; or (c) a merger, consolidation, recapitalization or reorganization of the Company with or into a Third Party Purchaser that results in the inability of the Members as of immediately prior to such merger, consolidation, recapitalization or reorganization to nominate or elect a majority of the Directors (or the board of directors (or its equivalent) of the resulting entity or its parent company).

"Citadel" means Citadel Securities Principal Investments LLC, a Delaware limited liability company, together with its Affiliates that hold Units. For the sake of clarity, (a) Citadel Securities Principal Investments LLC and its Affiliates holding Units (if any) shall be deemed to be one (1) Member, one (1) Class A Member, one (1) Class C Member, one (1) Class D Member, one (1) Nominating Member, and one (1) holder of Units of each applicable type, class or series that Citadel Securities Principal Investments LLC and/or its Affiliates hold, (b) Citadel Securities Principal Investments LLC (for so long as it holds Units) shall be entitled to receive notices on behalf of itself and its Affiliates that hold Units, and shall be entitled to take all actions under this Agreement with respect to itself and its Affiliates that hold Units, and (c) at such time as Citadel Securities Principal Investments LLC no longer holds any Units, Citadel Securities Principal

Investments LLC shall designate an Affiliate that then holds Units as the Member entitled to receive notices and take actions on behalf of itself and its Affiliates pursuant to written notice to the Company.

"Citi" means Citicorp North America, Inc., a Delaware corporation, together with its Affiliates that hold Units. For the sake of clarity, (a) Citicorp North America, Inc. and its Affiliates holding Units (if any) shall be deemed to be one (1) Member, one (1) Class A Member, one (1) Class C Member, one (1) Nominating Member, and one (1) holder of Units of each applicable type, class or series that Citicorp North America, Inc. and/or its Affiliates hold, (b) Citicorp North America, Inc. (for so long as it holds Units) shall be entitled to receive notices on behalf of itself and its Affiliates that hold Units, and shall be entitled to take all actions under this Agreement with respect to itself and its Affiliates that hold Units, and (c) at such time as Citicorp North America, Inc. no longer holds any Units, Citicorp North America, Inc. shall designate an Affiliate that then holds Units as the Member entitled to receive notices and take actions on behalf of itself and its Affiliates pursuant to written notice to the Company.

"Class A Member" means a Member holding Class A-1 Units, Class A-2 Units, Nonvoting Class A-1 Units or Nonvoting Class A-2 Units, as applicable, in its capacity as such, together with its Affiliates that hold Class A-1 Units, Class A-2 Units, Nonvoting Class A-1 Units or Nonvoting Class A-2 Units, as applicable (for the sake of clarity, such Member and such Affiliates shall be considered to be one (1) Class A Member).

"Class A-1 Units" means the Units having the privileges, preference, duties, liabilities, obligations and rights specified with respect to "Class A-1 Units" in this Agreement.

"Class A-2 Units" means the Units having the privileges, preference, duties, liabilities, obligations and rights specified with respect to "Class A-2 Units" in this Agreement.

"Class A Units" means the Class A-1 Units, the Class A-2 Units, the Nonvoting Class A-1 Units and the Nonvoting Class A-2 Units.

"Class B Member" means a Member holding Class B Units, in its capacity as such.

"Class B Units" means the Units having the privileges, preference, duties, liabilities, obligations and rights specified with respect to "Class B Units" in this Agreement.

"Class C Member" means a Member holding Class C-1 Units or Class C-2 Units, as applicable, in its capacity as such, together with its Affiliates that hold Class C-1 Units or Class C-2 Units, as applicable (for the sake of clarity, such Member and such Affiliates shall be considered to be one (1) Class C Member).

"Class C-1 Units" means the Units having the privileges, preference, duties, liabilities, obligations and rights specified with respect to "Class C-1 Units" in this Agreement.

- "Class C-1/D-1 Voting Percentage" means, at any time of calculation, a fraction, expressed as a percentage, (i) the numerator of which is the number of then issued and outstanding Class C-1 Units and Class D-1 Units held by a Member and (ii) the denominator of which is the number of then issued and outstanding Class C-1 Units and Class D-1 Units held by all Members.
- "Class C-2 Units" means the Units having the privileges, preference, duties, liabilities, obligations and rights specified with respect to "Class C-2 Units" in this Agreement.
- "Class C Unit Original Purchase Price" means the purchase price per Class C Unit set forth in the Members Schedule as of the Effective Date.
  - "Class C Units" means the Class C-1 Units and the Class C-2 Units.
- "Class D Member" means a Member holding Class D-1 Units or Class D-2 Units, as applicable, in its capacity as such, together with its Affiliates that hold Class D-1 Units or Class D-2 Units, as applicable (for the sake of clarity, such Member and such Affiliates shall be considered to be one (1) Class D Member).
- "Class D-1 Units" means the Units having the privileges, preference, duties, liabilities, obligations and rights specified with respect to "Class D-1 Units" in this Agreement.
- "Class D-2 Units" means the Units having the privileges, preference, duties, liabilities, obligations and rights specified with respect to "Class D-2 Units" in this Agreement.
- "Class D Unit Original Purchase Price" means the purchase price per Class D Unit set forth in the Members Schedule as of the Effective Date.
  - "Class D Units" means the Class D-1 Units and the Class D-2 Units.
  - "Code" means the Internal Revenue Code of 1986, as amended.
  - "Combination" has the meaning set forth in Section 8.17(a).
- "Common Member" means a Member holding Common Units in its capacity as such, together with its Affiliates that hold Common Units (for the sake of clarity, such Member and such Affiliates shall be considered to be one (1) Common Member).
- "Common Units" means the Units having the privileges, preference, duties, liabilities, obligations and rights specified with respect to "Common Units" in this Agreement.
  - "Company" has the meaning set forth in the Preamble.
- "Company Minimum Gain" means "partnership minimum gain" as defined in Section 1.704-2(b)(2) of the Treasury Regulations, substituting the term "Company" for the term "partnership" as the context requires.

"Company Option Period" has the meaning set forth in Section 10.3(d)(ii).

"Company Related Party" means (a) any manager, officer, director, employee, independent contractor and/or consultant of the Company or any Company Subsidiary, (b) (i) any Member or holder of equity interests of the Company or any Company Subsidiary, (ii) any Affiliate or any manager, officer, director, employee, independent contractor and/or consultant of any Member or holder of equity interests of the Company or any Company Subsidiary, or (iii) any manager, officer, director, employee, independent contractor and/or consultant of any Affiliate of a Member or holder of equity interests of the Company or any Company Subsidiary, (c) any Immediate Family Member of any Person specified in clause (a), and (d) any Person Controlled by one or more Persons listed in the immediately preceding clauses (a), (b) and/or (c).

"Company Related Party Transaction" means any contract, arrangement or transaction of any nature whatsoever, between or among the Company or any Company Subsidiary, on the one hand, and a Company Related Party, on the other hand, or in which any Company Related Party otherwise has a direct or indirect material interest; <u>provided</u>, that none of the following shall be deemed to be a Company Related Party Transaction:

- (a) any agreement for use of the Company's or any Company Subsidiary's products or services which is substantially in the Company's or such Company Subsidiary's standard form and on standard terms and conditions;
  - (b) any Competitively Sensitive Agreement;
- (c) any award pursuant to any equity incentive plan (including the Incentive Plan) approved by Supermajority Board Vote;
- (d) any advance, reimbursement or similar expenditure made or incurred by any officer, manager, director, employee, independent contractor or agent of the Company or any Company Subsidiary in the ordinary course of the business, so long as such advance, reimbursement or expenditure is in compliance with the Company's expense reimbursement policies approved by the Board;
- (e) any repurchase of a Member's Units pursuant to this Agreement or pursuant to any equity incentive plan (including the Incentive Plan) approved by Supermajority Board Vote;
- (f) any other contract, arrangement or transaction exempted by the Board by Supermajority Board Vote from being a Company Related Party Transaction (with the applicable Supermajority Board Vote excluding for the avoidance of doubt, each applicable interested Director or Alternate Director);
- (g) any employment contract, consulting agreement, offer letter or other employment (or consulting)-related document which provides for annual aggregate compensation (salary plus target bonus) of less than seven hundred fifty thousand dollars (\$750,000) (provided that any of the foregoing which are entered into with a Person covered by clause (b) of the

definition of "Company Related Party" or any issuance of Units under the Incentive Plan to a Person covered by clause (b) of the definition of the "Company Related Party" shall constitute a Company Related Party Transaction); and

(h) any agreement with an employee, consultant or independent contractor (other than an employment contract, consulting agreement, offer letter or other employment (or consulting)-related document) which provides for payment or receipt of ten thousand dollars (\$10,000) or less by the Company or any Company Subsidiary.

"Company ROFO Exercise Notice" has the meaning set forth in Section 10.3(d)(ii).

"Company Subsidiary" means a Subsidiary of the Company.

"Competitively Sensitive Agreement" means any agreement for use of the Company's or any Company Subsidiary's products or services which, as the CEO determines in good faith and relying upon written opinion of counsel (a copy of which shall be provided to the Board pursuant to Section 8.16) (the "Opinion"), (A) was negotiated on an arm's-length basis, (B) may not be shared under applicable antitrust laws or regulations with representatives of actual or potential competitors of the counterparty who are either Directors, Alternate Directors or Board Observers and (C) includes material terms which cannot reasonably be shared with the Board (after giving effect to any proposed Director recusals) in a manner that does not reveal any competitively or commercially sensitive information, the disclosure of which would be prohibited under applicable antitrust laws or regulations; provided, that the Board shall have the right to appoint outside counsel, at the Company's sole expense, to review and confirm the conclusions of the Company's counsel with respect to the determination under (B) and (C).

"Confidential Information" has the meaning set forth in Section 11.1(a).

"Control" means, when used with respect to any specified Person, the power, direct or indirect, to direct or cause the direction of the management and policies of such Person, whether through ownership of voting securities or partnership or other ownership interests, by contract or otherwise; and the terms "Controlling" and "Controlled" shall have correlative meanings. Without limiting the generality of the foregoing, a Person shall be deemed to Control another Person in which it owns, directly or indirectly, a majority of the ownership interests or voting securities.

"Converted Common Member" means a Member holding Converted Common Units in its capacity as such, together with its Affiliates that hold Converted Common Units (for the sake of clarity, such Member and such Affiliates shall be considered to be one (1) Converted Common Member).

"Converted Common Units" means the Common Units which were issued in connection with the conversion of Class C Units or Class D Units pursuant to Section 3.11.

"Covered Person" has the meaning set forth in Section 14.1(a).

"<u>Delaware Act</u>" means the Delaware Limited Liability Company Act, 6 Del. C. Section 18-101 <u>et seq.</u>, as amended and in effect from time to time, and any successor statute.

"<u>Director</u>" has the meaning set forth in <u>Section 8.1</u>.

"Directors and Observers Schedule" has the meaning set forth in Section 8.3(e).

"Disclosing Party" has the meaning set forth in Section 11.1(a).

"<u>Distribution</u>" means a distribution made by the Company to a Member, whether in cash, property or securities of the Company and whether by liquidating distribution or otherwise; <u>provided</u>, that none of the following shall be a Distribution: (a) any redemption or repurchase by the Company or any Member of any Units or Unit Equivalents; (b) any recapitalization or exchange of securities of the Company; (c) any subdivision (by a split of Units or otherwise) or any combination (by a reverse split of Units or otherwise) of any outstanding Units; or (d) any fees or remuneration paid to any Member in such Member's capacity as a Director, Alternate Director, Officer, employee, consultant or other service provider of the Company for the Company or a Company Subsidiary, if and only if, in the case of the foregoing clauses (a)-(d), such actions are approved as set forth herein. "<u>Distribute</u>" when used as a verb and "<u>Distributive</u>" when used as an adjective shall have correlative meanings.

"<u>Distribution Amount</u>" means, for any distribution under <u>Section 7.2</u>, the amount that the Board by Supermajority Board Vote determines to distribute under Section 7.2.

"Drag-along Sale" has the meaning set forth in Section 10.4(a).

"Effective Date" has the meaning set forth in the Preamble.

"<u>Electronic Transmission</u>" means any form of communication not directly involving the physical transmission of paper that creates a record that may be retained, retrieved and reviewed by a recipient thereof and that may be directly reproduced in paper form by such a recipient through an automated process.

"Emergency Funding" has the meaning set forth in Section 9.1(i).

"Equity Interest Equivalents" means (a) any security or obligation that is by its terms, directly or indirectly, convertible into, exchangeable or exercisable for equity interests of a Company Subsidiary, (b) any option, warrant or other right to subscribe for, purchase or acquire equity interests of a Company Subsidiary, (c) any equity interest appreciation rights, phantom equity interests or phantom equity interest rights, contingent interest or other similar rights relating to a Company Subsidiary or (d) any equity or equity-linked securities of a Company Subsidiary or any of the securities referred to in clauses (a)-(c) above in connection with a combination of equity interests, recapitalization, exchange, merger, consolidation or other reorganization.

"Exchange Act" means the U.S. Securities Exchange Act of 1934, as amended and in effect from time to time, and any successor statute, and the applicable rules and regulations promulgated thereunder.

"Exchange Board" means the board of directors of a Regulated Securities Exchange Subsidiary.

"Exchange Board Observer" has the meaning set forth in Section 8.18(g).

"Exchange Board Observer Appointing Member" means each Member that is a Nominating Member, subject to Section 8.18(g)(iv).

"Exchange Director" means a member of an Exchange Board nominated by an Exchange Director Nominating Member.

"Exchange Director Nominating Member" means each of Morgan Stanley, Schwab and Virtu.

"Exchange Member" has the meaning set forth in the definition of "Related Persons."

"Exempted Securities" has the meaning set forth in Section 1.4(a)(v) of Exhibit G.

"Exercise Period" has the meaning set forth in Section 9.1(e).

"Exercising Member" has the meaning set forth in Section 9.1(f).

"Fair Market Value" means:

- (a) as applied to any asset constituting cash or cash equivalents, the amount of such cash or cash equivalents;
- (b) as applied to any asset constituting publicly traded securities that may be immediately sold in the public markets without any restrictions or limitations, the average, over a period of twenty-one (21) Business Days consisting of the date of valuation and the twenty (20) consecutive Business Days prior to that date, of the average of the closing prices of the sales of such securities on the primary securities exchange on which such securities may at that time be listed, or, if there have been no sales on such exchange on any Business Day, the average of the highest bid and lowest asked prices on such exchange at the end of such Business Day, or, if on any Business Day such securities are not so listed, the average of the representative bid and asked prices quoted in the NASDAQ System as of 4:00 p.m., New York time, or, if on any Business Day such securities are not quoted in the NASDAQ System, the average of the highest bid and lowest asked prices on such Business Day in the domestic over the counter market as reported by the National Quotation Bureau Incorporated, or any similar successor organization;
- (c) as applied to any assets other than cash, cash equivalents or publicly traded securities that may be immediately sold in the public markets without any restrictions or limitations,

the fair value of such assets, as determined by the Board in good faith based on such factors as the Board, in the exercise of its reasonable business judgment, considers relevant but without taking into account any discounts for lack of liquidity or minority interest or similar discounts; provided, that a Member may, within fifteen (15) Business Days following receipt by such Member of the Board's determination of Fair Market Value, direct the Board to obtain an independent third-party appraisal of the Board's determination, with the determination by the independent appraiser binding on the parties.

"FCPA" means the U.S. Foreign Corrupt Practices Act, as amended and in effect from time to time, and the rules and regulations promulgated thereunder.

"Fidelity" means FMR LLC, a Delaware limited liability company, together with its Affiliates that hold Units. For the sake of clarity, (a) FMR LLC and its Affiliates holding Units (if any) shall be deemed to be one (1) Member, one (1) Class A Member, one (1) Class C Member, one (1) Nominating Member, and one (1) holder of Units of each applicable type, class or series that FMR LLC and/or its Affiliates hold, (b) FMR LLC (for so long as it holds Units) shall be entitled to receive notices on behalf of itself and its Affiliates that hold Units, and shall be entitled to take all actions under this Agreement with respect to itself and its Affiliates that hold Units, and (c) at such time as FMR LLC no longer holds any Units, FMR LLC shall designate an Affiliate that then holds Units as the Member entitled to receive notices and take actions on behalf of itself and its Affiliates pursuant to written notice to the Company.

"Fifth Amended LLC Agreement" has the meaning set forth in the Recitals.

"FINRA" means the Financial Industry Regulatory Authority, Inc. and any successor thereto.

"First Amended LLC Agreement" has the meaning set forth in the Recitals.

"<u>Fiscal Year</u>" means the calendar year, unless the Company is required to have a taxable year other than the calendar year, in which case Fiscal Year shall be the period that conforms to its taxable year.

"Forfeiture Allocations" has the meaning set forth in Section 6.2(e).

"Fourth Amended LLC Agreement" has the meaning set forth in the Recitals.

"Fully Diluted Basis" means, as of any date of determination, (a) with respect to all the Units, all issued and outstanding Units of the Company and all Units issuable upon the exercise of any outstanding Unit Equivalents as of such date, whether or not such Unit Equivalent is at the time vested or exercisable, or (b) with respect to any specified type, class or series of Units, all issued and outstanding Units designated as such type, class or series and all such designated Units issuable upon the exercise of any outstanding Unit Equivalents as of such date, whether or not such Unit Equivalent is at the time vested or exercisable.

"Fully Participating Tag-along Member" has the meaning set forth in Section 10.5(e)(i).

"GAAP" means United States generally accepted accounting principles in effect from time to time and as applied on a consistent basis with past practices.

"Goldman Sachs" means Goldman Sachs PSI Global Holdings, LLC, a Delaware limited liability company, together with its Affiliates that hold Units. For the sake of clarity, (a) Goldman Sachs PSI Global Holdings, LLC and its Affiliates holding Units (if any) shall be deemed to be one (1) Member, one (1) Class A Member, one (1) Class C Member, one (1) Nominating Member, and one (1) holder of Units of each applicable type, class or series that Goldman Sachs PSI Global Holdings, LLC and/or its Affiliates hold, (b) Goldman Sachs PSI Global Holdings, LLC (for so long as it holds Units) shall be entitled to receive notices on behalf of itself and its Affiliates that hold Units, and shall be entitled to take all actions under this Agreement with respect to itself and its Affiliates that hold Units, and (c) at such time as Goldman Sachs PSI Global Holdings, LLC no longer holds any Units, Goldman Sachs PSI Global Holdings, LLC shall designate an Affiliate that then holds Units as the Member entitled to receive notices and take actions on behalf of itself and its Affiliates pursuant to written notice to the Company.

"Governmental Authority" means any federal, state, local or foreign government or political subdivision thereof, or any agency or instrumentality of such government or political subdivision, or any self-regulated organization or other non-governmental regulatory authority or quasi-governmental authority (to the extent that the rules, regulations or orders of such organization or authority have the force of law), or any arbitrator, court or tribunal of competent jurisdiction.

"Immediate Family Member" means with respect to a natural person, such individual's child, stepchild, grandchild, parent, stepparent, grandparent, spouse, former spouse, domestic partner, sibling, uncle, aunt, niece, nephew, cousin, mother-in-law, father-in-law, son-in-law, daughter-in-law, brother-in-law or sister-in-law, including adoptive relationships.

"Incentive Plan" has the meaning set forth in Section 3.3(b).

"Industry Advisory Board" has the meaning set forth in Section 8.19(a).

"Industry Advisory Board Member" has the meaning set forth in Section 8.19(a).

"Initial Public Offering" has the meaning set forth in Section 11.5(a).

"Initial Term" has the meaning set forth in Section 8.18(c)(iii).

"IPO Entity" has the meaning set forth in Section 11.5(a).

"Issuance Notice" has the meaning set forth in Section 9.1(c).

"Jane Street" means Jane Street Group, LLC, a Delaware limited liability company together with its Affiliates that hold Units. For the sake of clarity, (a) Jane Street Group, LLC and its Affiliates holding Units (if any) shall be deemed to be one (1) Member, one (1) Class A Member,

one (1) Class C Member, one (1) Class D Member, one (1) Nominating Member, and one (1) holder of Units of each applicable type, class or series that Jane Street Group, LLC and/or its Affiliates hold, (b) Jane Street Group, LLC (for so long as it holds Units) shall be entitled to receive notices on behalf of itself and its Affiliates that hold Units, and shall be entitled to take all actions under this Agreement with respect to itself and its Affiliates that hold Units, and (c) at such time as Jane Street Group, LLC no longer holds any Units, Jane Street Group, LLC shall designate an Affiliate that then holds Units as the Member entitled to receive notices and take actions on behalf of itself and its Affiliates pursuant to written notice to the Company.

"JPMorgan" means JPMC Strategic Investments I Corporation, a Delaware corporation, together with its Affiliates that hold Units. For the sake of clarity, (a) JPMC Strategic Investments I Corporation and its Affiliates holding Units (if any) shall be deemed to be one (1) Member, one (1) Class A Member, one (1) Class C Member, one (1) Nominating Member, and one (1) holder of Units of each applicable type, class or series that JPMC Strategic Investments I Corporation and/or its Affiliates hold, (b) JPMC Strategic Investments I Corporation (for so long as it holds Units) shall be entitled to receive notices on behalf of itself and its Affiliates that hold Units, and shall be entitled to take all actions under this Agreement with respect to itself and its Affiliates that hold Units, and (c) at such time as JPMC Strategic Investments I Corporation no longer holds any Units, JPMC Strategic Investments I Corporation shall designate an Affiliate that then holds Units as the Member entitled to receive notices and take actions on behalf of itself and its Affiliates pursuant to written notice to the Company.

"Liquidation Proceeds" has the meaning set forth in Section 13.3(c).

"Liquidator" has the meaning set forth in Section 13.3(a).

"Losses" has the meaning set forth in Section 14.3(a).

"Market Maker Director" means a Director nominated by a Market Maker Member.

"<u>Market Maker Member</u>" means each of Citadel, Virtu, Jane Street, Optiver and any other Member that is specifically designated as a Market Maker Member, in each case, together with each of their respective Affiliates. For the avoidance of doubt, no Market Maker Member shall be deemed a Bank Member, a Buy Side Member or a Retail Broker Member, and no Bank Member, Buy Side Member or Retail Broker Member shall be deemed a Market Maker Member for the purposes of this Agreement.

"Maximum Aggregate Economic Interest" has the meaning set forth in Section 3.9(a).

"Maximum Class C-1/D-1 Voting Percentage" has the meaning set forth in Section 3.10(a).

"Maximum Voting Class A Voting Percentage" has the meaning set forth in Section 3.10(a).

"Member" means (a) each Person that has been admitted as a Member of the Company as of the Effective Date, and (b) each Person that is admitted as a Member of the Company after the Effective Date in accordance with the terms of this Agreement and the Delaware Act, in each case so long as such Person is shown on the Company's books and records as the owner of one or more Units. The Members shall constitute the "members" (as that term is defined in the Delaware Act) of the Company.

"Member Nonrecourse Debt" means "partner nonrecourse debt" as defined in Treasury Regulation Section 1.704-2(b)(4), substituting the term "Company" for the term "partnership" and the term "Member" for the term "partner" as the context requires.

"Member Nonrecourse Debt Minimum Gain" means an amount, with respect to each Member Nonrecourse Debt, equal to the Company Minimum Gain that would result if the Member Nonrecourse Debt were treated as a Nonrecourse Liability, determined in accordance with Treasury Regulation Section 1.704-2(i)(3).

"Member Nonrecourse Deduction" means "partner nonrecourse deduction" as defined in Treasury Regulation Section 1.704-2(i), substituting the term "Member" for the term "partner" as the context requires.

"Member ROFO Exercise Notice" has the meaning set forth in Section 10.3(d)(iii).

"Members Schedule" has the meaning set forth in Section 3.1.

"Membership Interest" means an interest in the Company owned by a Member, including such Member's right (based on the type, class and series of Unit or Units held by such Member), as applicable, (a) to a Distributive share of Net Income, Net Losses and other items of income, gain, loss and deduction of the Company; (b) to a Distributive share of the assets of the Company; (c) to vote on, consent to or otherwise participate in any decision of the Members as provided in this Agreement; and (d) to any and all other benefits to which such Member may be entitled as provided in this Agreement or the Delaware Act.

"MEMX LLC Agreement" means the Second Amended and Restated Limited Liability Company Agreement of MEMX LLC, a Subsidiary of the Company, effective as of May 19, 2020, as may be amended or restated from time to time.

"MEMX LLC Exchange Director Nomination Rotation" has the meaning set forth in Section 8.18(c)(iii)(A).

"MX2 LLC Agreement" means the First Amended and Restated Limited Liability Company Agreement of MX2 LLC, a Subsidiary of the Company, effective as of September 17, 2024, as may be amended or restated from time to time.

"MX2 LLC Exchange Director Nomination Rotation" has the meaning set forth in <u>Section</u> 8.18(c)(iii)(B).

"MX2 LLC Stub Period" has the meaning set forth in Section 8.18(c)(ii).

"Morgan Stanley" means Strategic Investments I, Inc., together with its Affiliates that hold Units. For the sake of clarity, (a) Strategic Investments I, Inc. and its Affiliates holding Units (if any) shall be deemed to be one (1) Member, one (1) Class A Member, one (1) Class C Member, one (1) Nominating Member, one (1) Exchange Director Nominating Member, and one (1) holder of Units of each applicable type, class or series that Strategic Investments I, Inc. and/or its Affiliates hold, (b) Strategic Investments I, Inc. (for so long as it holds Units) shall be entitled to receive notices on behalf of itself and its Affiliates that hold Units, and shall be entitled to take all actions under this Agreement with respect to itself and its Affiliates that hold Units, and (c) at such time as Strategic Investments I, Inc. no longer holds any Units, Strategic Investments I, Inc. shall designate an Affiliate that then holds Units as the Member entitled to receive notices and take actions on behalf of itself and its Affiliates pursuant to written notice to the Company.

"Misallocated Item" has the meaning set forth in Section 6.5.

"Net Available Cash" shall mean, for a Fiscal Year or any other applicable period, the excess, if any, of (a) the total cash receipts of the Company and Company Subsidiaries for such Fiscal Year or other period, exclusive of cash proceeds of any obligations created, issued or incurred by the Company or any Company Subsidiary for borrowed money and capital contributions to the Company and Company Subsidiaries (unless and to the extent that the Board by Supermajority Board Vote determines to treat such cash proceeds or capital contributions as part of Net Available Cash), minus (b) the sum, without duplication, of (i) total cash disbursements made or to be made by the Company and Company Subsidiaries in such period, and (ii) the amount, if any, of such cash receipts or cash proceeds that the Board by Supermajority Board Vote determines in good faith is to be reserved for the anticipated needs of the Company's and Company Subsidiaries' business or activities (including based on the then-applicable Annual Budget and business plan of the Company), including in respect of the Company's and Company Subsidiaries' current liabilities (as determined in accordance with GAAP) or reserves set aside for the Company's and Company Subsidiaries' expenses, debt payments, capital improvements, liabilities, capital commitments, investments and contingencies.

"Net Income" and "Net Loss" mean, for each Fiscal Year or other period specified in this Agreement, an amount equal to the Company's net taxable income or net taxable loss, or particular items thereof, determined in accordance with Code Section 703(a) (where, for this purpose, all items of income, gain, loss or deduction required to be stated separately pursuant to Code Section 703(a)(1) shall be included in taxable income or taxable loss), but with the following adjustments:

- (a) any income realized by the Company that is exempt from federal income taxation, as described in Code Section 705(a)(1)(B), shall be added to such taxable income or taxable loss, notwithstanding that such income is not includable in gross income;
- (b) any expenditures of the Company described in Code Section 705(a)(2)(B), including any items treated under Treasury Regulation Section 1.704-1(b)(2)(iv)(i) as items described in Code Section 705(a)(2) (B), shall be subtracted from such taxable income or taxable loss, notwithstanding that such expenditures are not deductible for federal income tax purposes;

- (c) any gain or loss resulting from any disposition of Company property with respect to which gain or loss is recognized for federal income tax purposes shall be computed by reference to the Book Value of the property so disposed, notwithstanding that the adjusted tax basis of such property differs from its Book Value;
- (d) any items of depreciation, amortization and other cost recovery deductions with respect to Company property having a Book Value that differs from its adjusted tax basis shall be computed by reference to the property's Book Value (as adjusted for Book Depreciation) in accordance with Treasury Regulation Section 1.704-1(b)(2)(iv)(g);
- (e) if the Book Value of any Company property is adjusted as provided in the definition of Book Value, then the amount of such adjustment shall be treated as an item of gain or loss and included in the computation of such taxable income or taxable loss;
- (f) to the extent an adjustment to the adjusted tax basis of any Company property pursuant to Code Sections 732(d), 734(b) or 743(b) is required, pursuant to Treasury Regulation Section 1.704 1(b)(2)(iv)(m), to be taken into account in determining Capital Accounts, the amount of such adjustment to the Capital Accounts shall be treated as an item of gain (if the adjustment increases the basis of the asset) or loss (if the adjustment decreases such basis); and
- (g) Notwithstanding any other provision of this definition, any items that are specially allocated pursuant to <u>Section 6.2</u> or <u>Section 6.5</u> shall not be taken into account in computing Net Income and Net Loss.

The amounts of the items of Company income, gain, loss or deduction available to be specially allocated pursuant to <u>Section 6.2</u> or <u>Section 6.5</u> shall be determined by applying rules analogous to those set forth in clauses (a) through (g) above.

"New Interests" has the meaning set forth in Section 3.4.

"New Securities" has the meaning set forth in Section 9.1(b).

"Nominating Member" has the meaning set forth in Section 8.3(b).

"Nominating Transferee" has the meaning set forth in Section 8.10(b).

"Nomination Waiver" has the meaning set forth in Section 8.11(a).

"Non-Exercising Member" has the meaning set forth in Section 9.1(f).

"Nonrecourse Liability" has the meaning set forth in Treasury Regulations Section 1.704-2(b)(3).

"Nonvoting Class A Units" means the Nonvoting Class A-1 Units and the Nonvoting Class A-2 Units.

- "Nonvoting Class A-1 Units" has the meaning set forth in Section 3.2(c).
- "Nonvoting Class A-2 Units" has the meaning set forth in Section 3.2(d).
- "Nonvoting Common Units" has the meaning set forth in Section 3.2(f)(iii).
- "OFAC" means the Office of Foreign Assets Control of the United States Treasury Department or any successor thereto.
  - "Offered Units" has the meaning set forth in Section 10.3(a).
  - "Offering Member" has the meaning set forth in Section 10.3(a).
  - "Offering Member Notice" has the meaning set forth in Section 10.3(c)(i).
  - "Officers" has the meaning set forth in Section 8.14(a).
- "Opinion" has the meaning set forth in the definition of "Competitively Sensitive Agreement."
- "Optiver" means Optiver PSI B1 LLC, together with its Affiliates that hold Units. For the sake of clarity, (a) Optiver PSI B1 LLC and its Affiliates holding Units (if any) shall be deemed to be one (1) Member, one (1) Class D Member, one (1) Nominating Member, and one (1) holder of Units of each applicable type, class or series that Optiver PSI B1 LLC and/or its Affiliates hold, (b) Optiver PSI B1 LLC (for so long as it holds Units) shall be entitled to receive notices on behalf of itself and its Affiliates that hold Units, and shall be entitled to take all actions under this Agreement with respect to itself and its Affiliates that hold Units, and (c) at such time as Optiver PSI B1 LLC no longer holds any Units, Optiver PSI B1 LLC shall designate an Affiliate that then holds Units as the Member entitled to receive notices and take actions on behalf of itself and its Affiliates pursuant to written notice to the Company.
  - "Original LLC Agreement" has the meaning set forth in the Recitals.
  - "Over-allotment Exercise Period" has the meaning set forth in Section 9.1(f).
  - "Over-allotment Notice" has the meaning set forth in Section 9.1(f).
  - "Participation Threshold" has the meaning set forth in Section 3.3(e)(i).
  - "Partnership Audit Rules" has the meaning set forth in Section 12.3(b).
  - "Partnership Representative" has the meaning set forth in Section 12.3(b).
  - "Permitted Anti-Dilution Conversion" has the meaning set forth in Section 3.10(e)(ii).
  - "<u>Permitted Persons</u>" has the meaning set forth in <u>Section 11.2(a)</u>.

- "Permitted Regulatory Transfer" has the meaning set forth in Section 3.10(e)(i).
- "Permitted Transfer" means a Transfer of Units carried out pursuant to Section 10.2.
- "Permitted Transferee" means a recipient of a Permitted Transfer.
- "Person" means an individual, corporation, partnership, joint venture, limited liability company, Governmental Authority, unincorporated organization, trust, association or other entity.
  - "Post-Closing Preemptive Rights Notice" has the meaning set forth in Section 9.1(i).
  - "Pre-emptive Member" has the meaning set forth in Section 9.1(a).
  - "Prior Class C-1/D-1 Voting Percentage" has the meaning set forth in Section 3.10(e)(ii)(B).
- "Prior Voting Class A Voting Percentage" has the meaning set forth in Section 3.10(e)(ii)(A).

### "Pro Rata Portion" means:

- (a) for purposes of Section 9.1, with respect to any Pre-emptive Member holding Units, on any issuance date for New Securities, a fraction determined by dividing (i) the total number of outstanding Class A Units, Class C Units and Class D Units held by such Member on such date immediately prior to such issuance by (ii) the total number of Class A Units, Class C Units and Class D Units held by all Pre-emptive Members outstanding on such date immediately prior to such issuance; and
- (b) for purposes of <u>Section 10.3</u>, with respect to a ROFO Rightholder holding Units, on any date of a proposed Transfer by an Offering Member, a fraction determined by dividing (i) the number of outstanding vested Units held by such ROFO Rightholder immediately prior to such Transfer by (ii) the total number of outstanding vested Units held by all the ROFO Rightholders on such date immediately prior to such Transfer.
  - "Profits Unit Cap" has the meaning set forth in Section 13.3(e).
  - "Proposed Transferee" has the meaning set forth in Section 10.5(a).
- "<u>Proprietary Identifier</u>" means the name of a Member or its Affiliates or employees, or any trade name, trademark, trade device, logo, service mark, symbol or abbreviation, contraction or simulation thereof.
- "<u>Public Offering</u>" means any underwritten public offering pursuant to a registration statement filed in accordance with the Securities Act.
  - "Purchasing Rightholders" has the meaning set forth in Section 10.3(e)(ii).

"Qualified Member" has the meaning set forth in Section 12.1.

"Qualified Public Offering" means the sale, in a firm commitment underwritten public offering led by a nationally recognized underwriting firm pursuant to an effective registration statement under the Securities Act, of Units (or common stock of the Company or an IPO Entity) (a) having an aggregate offering value (net of underwriters' discounts and selling commissions) of at least one hundred million dollars (\$100,000,000), (b) which is at a per Unit offering price (or per share offering price for the shares into which each Class A Unit has converted) that is at least two times (2x) the purchase price paid for Class A-1 Units as of December 14, 2018 (subject to appropriate adjustment in the event of any Unit dividend, Unit split, combination or other similar recapitalization), and (c) following which at least 20% of the total Units (or common stock of the Company or an IPO Entity) on a Fully Diluted Basis shall have been sold to the public and shall be listed on any national securities exchange or quoted on the NASDAQ Stock Market System.

"Receiving Party" has the meaning set forth in Section 11.1(a).

"Registration Date" means the date on which MEMX LLC was registered with the SEC as a national securities exchange, which was May 4, 2020.

"Regulated Securities Exchange Subsidiary" shall mean any national securities exchange controlled, directly or indirectly, by the Company, including MEMX LLC and MX2 LLC.

"Regulated Securities Exchange Subsidiary LLC Agreement" shall mean the limited liability company agreement of each Registered Securities Exchange Subsidiary, including the MEMX LLC Agreement and MX2 LLC Agreement.

"Regulatory Allocations" has the meaning set forth in Section 6.2(d).

"Regulatory Hardship Determination" has the meaning set forth in Section 10.6(a).

"Related Persons" means with respect to any Person: (a) any "affiliate" of such Person (as such term is defined in Rule 12b-2 under the Exchange Act); (b) any other Person with which such first Person has any agreement, arrangement or understanding (whether or not in writing) to act together for the purpose of acquiring, voting, holding or disposing of Units; (c) in the case of a Person that is a company, corporation or similar entity, any executive officer (as defined under Rule 3b-7 under the Exchange Act) or director of such Person and, in the case of a Person that is a partnership or limited liability company, any general partner, managing member or manager of such Person, as applicable; (d) in the case of any Person that is a registered broker or dealer that has been admitted to membership in a Regulated Securities Exchange Subsidiary (an "Exchange Member"), any Person that is associated with the Exchange Member (as determined using the definition of "person associated with a member" as defined under Section 3(a)(21) of the Exchange Act); (e) in the case of a Person that is a natural person and Exchange Member, any broker or dealer that is also an Exchange Member with which such Person is associated; (f) in the case of a Person that is a natural person, any relative or spouse of such Person, or any relative of such spouse who has the same home as such Person or who is a director or officer of the Company, any Company

Subsidiary, or any of the Company's parent companies; (g) in the case of a Person that is an executive officer (as defined under Rule 3b-7 under the Exchange Act) or a director of a company, corporation or similar entity, such company, corporation or entity, as applicable; and (h) in the case of a Person that is a general partner, managing member or manager of a partnership or limited liability company, such partnership or limited liability company, as applicable.

"Remaining Portion" has the meaning set forth in Section 10.5(e)(i).

"Remaining Portion Notice" has the meaning set forth in Section 10.5(e)(i).

"Remaining Tag-along Notice" has the meaning set forth in Section 10.5(e)(ii).

"Representative" means, with respect to any Person, any and all directors, officers, employees, consultants, financial advisors, counsel, accountants and other agents of such Person.

"Restricted Economic Election" has the meaning set forth in Section 3.9(a).

"Restricted Economic Member" has the meaning set forth in Section 3.9(a).

"Restricted Entity" means any Person that is (i) a financial holding company, (ii) a bank holding company, (iii) a foreign bank that is subject to the BHCA, (iv) a savings and loan holding company under the Home Owners' Loan Act of 1933, as amended, (v) an investment advisor, an investment company, a broker-dealer, security-based swap dealer or security-based swap execution facility, (vi) an insurance company, (vii) a Person regulated by the Commodity Futures Trading Commission, (viii) a Controlled Affiliate of any of the foregoing, or subject to registration with any non-U.S. Governmental authority or any SRO, or (ix) any entity that is the functional equivalent of any of the Persons described in the foregoing subsections (i) - (viii).

"Restricted Voting Election" has the meaning set forth in Section 3.10(a).

"Restricted Voting Member" has the meaning set forth in Section 3.10(a).

"Retail Broker Director" means a Director nominated by a Retail Broker Member.

"Retail Broker Member" means each of Fidelity, Schwab, and any other Member that is specifically designated as a Retail Broker Member and which, or an Affiliate of which, is a broker-dealer registered with the Financial Industry Regulatory Authority, Inc. which provides services to retail customers, in each case, together with each of their respective Affiliates. For the avoidance of doubt, no Retail Broker Member shall be deemed a Bank Member, a Buy Side Member or a Market Maker Member, and no Bank Member, Buy Side Member or Market Maker Member shall be deemed a Retail Broker Member for the purposes of this Agreement.

"ROFO Rightholders" has the meaning set forth in Section 10.3(a).

"ROFO Rightholder Option Period" has the meaning set forth in Section 10.3(d)(iii).

"Sale Notice" has the meaning set forth in Section 10.5(c).

"Schwab" means The Charles Schwab Corporation, a Delaware corporation, together with its Affiliates that hold Units. For the sake of clarity, (a) The Charles Schwab Corporation and its Affiliates holding Units (if any) shall be deemed to be one (1) Member, one (1) Class A Member, one (1) Nominating Member, one (1) Exchange Director Nominating Member, and one (1) holder of Units of each applicable type, class or series that The Charles Schwab Corporation and/or its Affiliates hold, (b) The Charles Schwab Corporation (for so long as it holds Units) shall be entitled to receive notices on behalf of itself and its Affiliates that hold Units, and shall be entitled to take all actions under this Agreement with respect to itself and its Affiliates that hold Units, and (c) at such time as The Charles Schwab Corporation no longer holds any Units, The Charles Schwab Corporation shall designate an Affiliate that then holds Units as the Member entitled to receive notices and take actions on behalf of itself and its Affiliates pursuant to written notice to the Company.

"SEC" means the U.S. Securities and Exchange Commission and any successor thereto.

"Second Amended LLC Agreement" has the meaning set forth in the Recitals.

"Securities Act" means the Securities Act of 1933, as amended, or any successor federal statute, and the rules and regulations thereunder, which shall be in effect at the time.

"Selling Member" has the meaning set forth in Section 10.5(a).

"Seventh Amended LLC Agreement" has the meaning set forth in the Recitals.

"Shortfall Amount" has the meaning set forth in Section 7.1.

"Sixth Amended LLC Agreement" has the meaning set forth in the Recitals.

"SRO" means a "self-regulatory organization" as defined in Section 3(a)(26) of the Exchange Act or any foreign equivalent.

"Subsidiary" means, with respect to any specified Person, any other Person of which such specified Person, directly or indirectly through one or more Subsidiaries, (a) owns a majority of the outstanding shares or other equity interests having the power to vote for directors or comparable managers of such other Person, or (b) has the power to generally direct the business and policies of that other Person, whether by contract or as a general partner, managing member, manager, joint venturer, agent or otherwise.

"Supermajority Board Matter" means any action or transaction set forth in Exhibit C.

"Supermajority Board Vote" means the affirmative vote of at least seventy-three percent (73%) of the votes of all Directors then entitled to vote on the matter under consideration and who have not recused themselves, whether or not present at the applicable meeting of the Board; provided that if such affirmative vote threshold results in the necessity of the affirmative vote of

eight (8) such Directors or fewer, an affirmative vote of all but two (2) of such Directors shall be required instead with respect to such matter.

"<u>Tag-along Member</u>" has the meaning set forth in <u>Section 10.5(a)</u>.

"Tag-along Notice" has the meaning set forth in Section 10.5(d)(ii).

"Tag-along Period" has the meaning set forth in Section 10.5(d)(ii).

"Tag-along Portion" has the meaning set forth in Section 10.5(d)(i).

"Tag-along Sale" has the meaning set forth in Section 10.5(a).

"Tax Advances" has the meaning set forth in Section 7.1.

"Tax Amount Base" of a Member for a Fiscal Year or other period shall mean (a) the total cumulative taxable income and gain (for U.S. federal income tax purposes) over (b) the total cumulative amount of losses and deductions (for U.S. federal income tax purposes, taking into account such losses and deductions solely to the extent of a character that can be utilized against such subsequent income) allocated to the Member by the Company for such Fiscal Year or other period; provided, that such taxable income shall be computed without regard to any taxable income or loss recognized by a Member (other than through its distributive share of income or gain of the Company) in connection with the dissolution, Initial Public Offering or any other sale of equity securities of the Company, sale of substantially all equity or assets of the Company or any similar event.

"<u>Tax Amounts</u>" of a Member for a Fiscal Year or other period shall mean the product of (a) the Tax Rate for such Fiscal Year or other period, and (b) the Member's Tax Amount Base for such Fiscal Year or other period, and shall be reduced by (c) any United States federal, state or local income tax credits allocated to the Member by the Company for such Fiscal Year or other period, all as estimated in good faith by the Company.

"Tax Rate" shall mean, for any Fiscal Year the greater of (a) the highest single combined rate (expressed as a percentage) of U.S. federal, state and local income taxation that would be applicable to any Member that is subject to tax as an individual for U.S. federal tax purposes, and (b) the highest single combined rate (expressed as a percentage) of U.S. federal, state and local income taxation that would be applicable to any Member that is subject to tax as a corporation for U.S. federal tax purposes, in each case on allocations of income to such Member and not taking into account any additional tax that could be imposed on distributions, deemed distributions, or dividend equivalent amounts, and in each case assuming that such Member is resident in New York City and that in case of an individual state taxes are not deductible for federal income tax purposes; provided that the Board by Supermajority Board Vote may establish any alternative rate. Each Member shall be subject to the same Tax Rate; provided that for any Fiscal Year the Board by Supermajority Board Vote may permit the use of different Tax Rates for the Members (e.g., to take into account whether a Member (or its direct and indirect owners) is subject to tax as an individual or

corporation for U.S. federal tax purposes). The Tax Rate shall take into consideration the nature of the income being allocated to the Members, such as whether such income is capital gain, qualified dividends or ordinary income.

"Taxing Authority" has the meaning set forth in Section 7.4(b).

"Third Amended LLC Agreement" has the meaning set forth in the Recitals.

"Third Party Purchaser" means any Person who, immediately prior to the contemplated transaction, (a) does not directly or indirectly own or have the right to acquire any outstanding Units (or applicable Unit Equivalents) or (b) is not a Permitted Transferee of any Person who directly or indirectly owns or has the right to acquire any Units (or applicable Unit Equivalents).

"Trading and Investment Business" has the meaning set forth in Section 11.1(d).

"Transfer" means to, directly or indirectly, sell, transfer, assign, pledge, encumber, hypothecate or similarly dispose of, either voluntarily or involuntarily, by operation of law or otherwise, or to enter into any contract, option or other arrangement or understanding with respect to the sale, transfer, assignment, pledge, encumbrance, hypothecation or similar disposition of, any Units owned by a Person or any interest (including a beneficial interest) in any Units or Unit Equivalents owned by a Person. A "Transfer" shall also include a direct change of Control of a Member, where the Member is a special purpose vehicle that owns no other material assets other than the Units and the change of Control of such Member is for the principal purpose of Transferring the Units owned by such Member to a non-Affiliate third party. "Transfer" when used as a noun shall have a correlative meaning. "Transferor" and "Transferee" mean a Person who makes or receives a Transfer, respectively.

"<u>Treasury Regulations</u>" means the final or temporary regulations issued by the United States Department of Treasury pursuant to its authority under the Code, and any successor regulations.

"UBS" means UBS Americas Inc., together with its Affiliates that hold Units. For the sake of clarity, (a) UBS Americas Inc. and its Affiliates holding Units (if any) shall be deemed to be one (1) Member, one (1) Class A Member, one (1) Class C Member, one (1) Nominating Member, and one (1) holder of Units of each applicable type, class or series that UBS Americas Inc. and/or its Affiliates hold, (b) UBS Americas Inc. (for so long as it holds Units) shall be entitled to receive notices on behalf of itself and its Affiliates that hold Units, and shall be entitled to take all actions under this Agreement with respect to itself and its Affiliates that hold Units, and (c) at such time as UBS Americas Inc. no longer holds any Units, UBS Americas Inc. shall designate an Affiliate that then holds Units as the Member entitled to receive notices and take actions on behalf of itself and its Affiliates pursuant to written notice to the Company.

"Unallocated Item" has the meaning set forth in Section 6.5.

"<u>Unit</u>" means a unit representing a fractional part of the Membership Interests of the Members and shall include all types, classes and series of Units, including the Class A Units, the

Class B Units, the Class C Units, the Class D Units and the Common Units; <u>provided</u>, that any type, class or series of Units shall have the privileges, preference, duties, liabilities, obligations and rights set forth in this Agreement and the Membership Interests represented by such type, class or series of Unit shall be determined in accordance with such privileges, preference, duties, liabilities, obligations and rights.

"<u>Unit Equivalents</u>" means (a) any security or obligation that is by its terms, directly or indirectly, convertible into, exchangeable or exercisable for Units, (b) any option, warrant or other right to subscribe for, purchase or acquire Units, (c) any equity interest appreciation rights, phantom equity interests or phantom equity interest rights, contingent interest or other similar rights relating to the Company or (d) any equity or equity-linked securities of the Company or issuable with respect to the equity interests of the Company or any securities referred to in clauses (a)-(c) above in connection with a combination of equity interests, recapitalization, exchange, merger, consolidation or other reorganization.

"<u>Using Member</u>" has the meaning set forth in <u>Section 11.6(c)</u>

"Virtu" means Virtu Investments, LLC, a Delaware limited liability company, together with its Affiliates that hold Units. For the sake of clarity, (a) Virtu Investments, LLC and its Affiliates holding Units (if any) shall be deemed to be one (1) Member, one (1) Class A Member, one (1) Class C Member, one (1) Class D Member, one (1) Nominating Member, one (1) Exchange Director Nominating Member, and one (1) holder of Units of each applicable type, class or series that Virtu Investments, LLC and/or its Affiliates hold, (b) Virtu Investments, LLC (for so long as it holds Units) shall be entitled to receive notices on behalf of itself and its Affiliates that hold Units, and shall be entitled to take all actions under this Agreement with respect to itself and its Affiliates that hold Units, and (c) at such time as Virtu Investments, LLC no longer holds any Units, Virtu Investments, LLC shall designate an Affiliate that then holds Units as the Member entitled to receive notices and take actions on behalf of itself and its Affiliates pursuant to written notice to the Company.

"Voting Class A Units" means the Class A-1 Units and Class A-2 Units.

"Voting Class A Voting Percentage" means, at any time of calculation, a fraction, expressed as a percentage (a) the numerator of which is be the number of then issued and outstanding Voting Class A Units held by a Class A Member and (b) the denominator of which is the number of then issued and outstanding Voting Class A Units held by all Class A Members.

"Voting Common Units" has the meaning set forth in Section 3.2(f)(ii).

"Voting Conversion Notice" has the meaning set forth in Section 3.10(e)(ii)(A).

"Wells Fargo" means Wells Fargo Central Pacific Holdings, Inc., a California corporation, together with its Affiliates that hold Units. For the sake of clarity, (a) Wells Fargo Central Pacific Holdings, Inc. and its Affiliates holding Units (if any) shall be deemed to be one (1) Member, one (1) Class A Member, one (1) Class C Member, one (1) Nominating Member, and one (1) holder of

Units of each applicable type, class or series that Wells Fargo Central Pacific Holdings, Inc. and/or its Affiliates hold, (b) Wells Fargo Central Pacific Holdings, Inc. (for so long as it holds Units) shall be entitled to receive notices on behalf of itself and its Affiliates that hold Units, and shall be entitled to take all actions under this Agreement with respect to itself and its Affiliates that hold Units, and (c) at such time as Wells Fargo Central Pacific Holdings, Inc. no longer holds any Units, Wells Fargo Central Pacific Holdings, Inc. shall designate an Affiliate that then holds Units as the Member entitled to receive notices and take actions on behalf of itself and its Affiliates pursuant to written notice to the Company.

#### "Withholding Advances" has the meaning set forth in Section 7.4(b).

Interpretation. For purposes of this Agreement, (a) the words "include," "includes" and "including" shall be deemed to be followed by the words "without limitation"; (b) the word "or" is not exclusive; and (c) the words "herein," "hereof," "hereby," "hereto" and "hereunder" refer to this Agreement as a whole. The definitions given for any defined terms in this Agreement shall apply equally to both the singular and plural forms of the terms defined. Whenever the context may require, any pronoun shall include the corresponding masculine, feminine and neuter forms. Unless the context otherwise requires, references herein: (x) to Articles, Sections, and Exhibits mean the Articles and Sections of, and Exhibits attached to, this Agreement; (y) to an agreement, instrument or other document means such agreement, instrument or other document as amended, supplemented and modified from time to time to the extent permitted by the provisions thereof and (z) to a statute means such statute as amended from time to time and includes any successor legislation thereto and any regulations promulgated thereunder. This Agreement shall be construed without regard to any presumption or rule requiring construction or interpretation against the party drafting an instrument or causing any instrument to be drafted. The Exhibits referred to herein shall be construed with, and as an integral part of, this Agreement to the same extent as if they were set forth verbatim herein.

# ARTICLE II. ORGANIZATION

### 2.1 <u>Formation; Agreement.</u>

- (a) The Company was formed on September 6, 2018, pursuant to the provisions of the Delaware Act, upon the filing of the Certificate of Formation with the Secretary of State of the State of Delaware.
- (b) This Agreement amends and restates the Seventh Amended LLC Agreement in its entirety. From and after the Effective Date, this Agreement constitutes the "limited liability company agreement" (as that term is used in the Delaware Act) of the Company. The rights, powers, duties, obligations and liabilities of the Members shall be determined pursuant to the Delaware Act and this Agreement. To the extent that the rights, powers, duties, obligations and liabilities of any Member are different by reason of any provision of this Agreement than they would be under the Delaware Act in the absence of such provision, this Agreement shall, to the extent permitted by the Delaware Act, control.

- 2.2 Name. The name of the Company is "MEMX Holdings LLC" or such other name or names as the Board may from time to time designate; <u>provided</u> that the name shall always contain the words "Limited Liability Company" or the abbreviation "L.L.C." or the designation "LLC." The Board shall give prompt notice to each of the Members of any change to the name of the Company.
- 2.3 <u>Principal Office</u>. The principal office of the Company shall be located at 382 NE 191<sup>st</sup> Street, Suite 92178, Miami, FL 33179, or such other place as may from time to time be determined by the Board. The Board shall give prompt notice of any such change to each of the Members.

# 2.4 Registered Office; Registered Agent.

- (a) The registered office of the Company shall be the office of the initial registered agent named in the Certificate of Formation or such other office (which need not be a place of business of the Company) as the Board may designate from time to time in the manner provided by the Delaware Act and Applicable Law.
- (b) The registered agent for service of process on the Company in the State of Delaware shall be the initial registered agent named in the Certificate of Formation or such other Person or Persons as the Board may designate from time to time in the manner provided by the Delaware Act and Applicable Law.

# 2.5 <u>Purpose; Powers.</u>

Subject to the provisions of this Agreement, the nature of the business to be (a) conducted or promoted by the Company is to (i) operate a national securities exchange for the trading of equity securities, directly or through one or more of its Subsidiaries, and (ii) engage in any other lawful act or activity for which limited liability companies may be organized under the Delaware Act and which is approved by the Board in accordance with Exhibit C, provided, however that the Company may not be or become a Restricted Entity; and provided, further, that any proposed or actual expansion of the business of the Company or its Subsidiaries into an options exchange and/or operating a global equities exchange (and any regulatory filings or application in connection therewith) shall, notwithstanding Items 33, 34 and 37 of Exhibit C, only require a majority Board approval (it being understood that if a Class A Member, a Class C Member or a Class D Member notifies the Board that it has concerns about legal and/or tax structuring implications of such expansion into (A) operating an options exchange or (B) a particular jurisdiction, the Company and the Members shall work in good faith to address such legal and/or tax structuring issues). For the avoidance of doubt, this Section 2.5(a) shall not in any way supersede the provisions set forth in Section 15.18. The Company may, subject to Section 11.3 and the other provisions of this Agreement, engage in any and all activities necessary, desirable or incidental to the accomplishment of the foregoing. Notwithstanding anything herein to the contrary, nothing set forth herein shall be construed as authorizing the Company to possess any purpose or power, or to do any act or thing, forbidden by Applicable Law to a limited liability company organized under the laws of the State of Delaware.

- (b) The Company shall have all the powers necessary or convenient to carry out the purposes for which it is formed, including the powers granted by the Delaware Act.
- 2.6 <u>Term.</u> The term of the Company commenced on the date the Certificate of Formation was filed with the Secretary of State of the State of Delaware and shall continue in existence perpetually until the Company is dissolved in accordance with the provisions of this Agreement.
- 2.7 <u>No State-Law Partnership</u>. The Members intend that the Company shall be treated as a partnership for federal and, if applicable, state and local income tax purposes, and, the Company shall not take any action inconsistent with such treatment. The Company and each Member shall file all tax returns and shall otherwise take all tax and financial reporting positions in a manner consistent with such treatment and no Member shall take any action inconsistent with such treatment. The Members intend that the Company shall not be a partnership (including, without limitation, a limited partnership) or joint venture, and that no Member, Director, Alternate Director or Officer of the Company shall be a partner or joint venturer of any other Member, Director, Alternate Director or Officer of the Company, for any purposes other than as set forth in the first sentence of this Section 2.7.

# ARTICLE III. UNITS; OWNERSHIP LIMITATIONS

- 3.1 <u>Units Generally.</u> The Membership Interests of the Members shall be represented by issued and outstanding Units, which may be divided into one or more types, classes or series. Each type, class or series of Units shall have the privileges, preference, duties, liabilities, obligations and rights, including voting rights, if any, set forth in this Agreement with respect to such type, class or series. The Board shall maintain a schedule of all Members, their respective mailing addresses and the amount and series of Units held by them (the "Members Schedule"), and shall update the Members Schedule upon the issuance or Transfer of any Units to any new or existing Member.
- 3.2 <u>Authorization and Issuance of Class A Units, Class C Units, Class D Units and Common Units.</u> Subject to compliance with <u>Section 8.6(d)</u>, <u>Section 9.1</u> and <u>Section 10.1(b)</u>, the Company is hereby authorized to issue (i) a series of Units designated as Class A-1 Units, (ii) a series of Units designated as Class A-2 Units, (iii) a series of Units designated as Nonvoting Class A-1 Units, (iv) a series of Units designated as Class C-1 Units, (vi) a series of Units designated as Class C-2 Units, (vii) a series of Units designated as Class D-1 Units, (viii) a series of Units designated as Class D-2 Units, (ix) a series of Units designated as Voting Common Units and (x) a series of Units designated as Nonvoting Common Units.
- (a) The Class A-1 Units are issued and outstanding and held by certain Class A Members as set forth on the Members Schedule as of the Effective Date and, as of the Effective Date, no additional Class A-1 Units are available for issuance. Class A-1 Units shall not be entitled to vote on any matter except as set forth in Sections 4.7(a), 4.7(b), 4.7(h) and 4.7(i) or as required by Applicable Law.

- (b) The Class A-2 Units are issued and outstanding and held by certain Class A Members as set forth on the Members Schedule as of the Effective Date and, as of the Effective Date, no additional Class A-2 Units are available for issuance. Class A-2 Units shall not be entitled to vote on any matter except as set forth in Sections 4.7(a), 4.7(b), 4.7(h) and 4.7(i) or as required by Applicable Law.
- (c) The Nonvoting Class A-1 Units issued and outstanding as of the Effective Date are set forth on the Members Schedule as of the Effective Date. Nonvoting Class A-1 Units ("Nonvoting Class A-1 Units") shall have all of the same rights, privileges and obligations as Class A-1 Units, except that Nonvoting Class A-1 Units shall not be entitled to vote on any matter except as set forth in Sections 4.7(h) and 4.7(i) or as required by Applicable Law. On all matters on which Nonvoting Class A-1 Units are entitled to vote, each Nonvoting Class A-1 Unit shall have one (1) vote per Nonvoting Class A Unit.
- (d) The Nonvoting Class A-2 Units issued and outstanding as of the Effective Date are set forth on the Members Schedule as of the Effective Date. Nonvoting Class A-2 Units ("Nonvoting Class A-2 Units") shall have all of the same rights, privileges and obligations as Class A-2 Units, except that Nonvoting Class A-2 Units shall not be entitled to vote on any matter except as set forth in Sections 4.7(h) and 4.7(i) or as required by Applicable Law. On all matters on which Nonvoting Class A-2 Units are entitled to vote, each Nonvoting Class A-2 Unit shall have one (1) vote per Nonvoting Class A-2 Unit.
- (e) (i) The Class C-1 Units issued and outstanding as of the Effective Date are set forth on the Members Schedule as of the Effective Date. Class C-1 Units shall have the voting rights set forth in Sections 4.7(c), 4.7(d), 4.7(e), 4.7(g) and 4.7(i) and as required by Applicable Law. On all matters on which Class C-1 Units are entitled to vote, each Class C-1 Unit shall have one (1) vote per Class C-1 Unit.
  - (ii) The Class C-2 Units issued and outstanding as of the Effective Date are set forth on the Members Schedule as of the Effective Date. Class C-2 Units shall have all of the same rights, privileges and obligations as Class C-1 Units, except that Class C-2 Units shall not be entitled to vote on any matter except as set forth in Sections 4.7(e), 4.7(g) and 4.7(i) or as required by Applicable Law. On all matters on which Class C-2 Units are entitled to vote, each Class C-2 Unit shall have one (1) vote per Class C-2 Unit.
  - (f) (i) The Class D-1 Units issued and outstanding as of the Effective Date are set forth on the Members Schedule as of the Effective Date. Class D-1 Units shall have the voting rights set forth in Sections 4.7(c), 4.7(d), 4.7(f), 4.7(g) and 4.7(i) and as required by Applicable Law. On all matters on which Class D-1 Units are entitled to vote, each Class D-1 Unit shall have one (1) vote per Class D-1 Unit.
  - (ii) The Class D-2 Units issued and outstanding as of the Effective Date are set forth on the Members Schedule as of the Effective Date. Class D-2 Units shall have all of the same rights, privileges and obligations as Class D-1 Units, except that Class D-2 Units shall not be entitled to vote on any matter except as set forth in Sections 4.7(f),

4.7(g) and 4.7(i) or as required by Applicable Law. On all matters on which Class D-2 Units are entitled to vote, each Class D-2 Unit shall have one (1) vote per Class D-2 Unit.

- (g) No Common Units are issued and outstanding as of the Effective Date. Subject to Sections 3.5, 3.6, 8.6(d) and 9.1, the number of Common Units which the Company is authorized to issue is not limited.
  - (i) Common Units shall only be issuable:
    - (A) in connection with an investment in the Company; or
    - (B) upon conversion of Class C Units or Class D Units as set forth in Section 3.11.
  - (ii) Voting Common Units (the "<u>Voting Common Units</u>") shall have the voting rights set forth in <u>Sections 4.7(c)</u> and <u>4.7(i)</u> and as required by Applicable Law. On all matters on which Voting Common Units are entitled to vote, each Voting Common Unit shall have one (1) vote per Voting Common Unit.
  - (iii) Nonvoting Common Units (the "Nonvoting Common Units") shall have all of the same rights, privileges and obligations as Voting Common Units, except that Nonvoting Common Units shall not be entitled to vote on any matter except as set forth in Section 4.7(i) or as required by Applicable Law. On all matters on which Nonvoting Common Units are entitled to vote, each Nonvoting Common Unit shall have one (1) vote per Nonvoting Common Unit.

## 3.3 <u>Service Provider Equity Pool; Class B Units.</u>

- (a) The Company is hereby authorized, subject to compliance with this <u>Section 3.3</u> and <u>Section 8.6(d)</u>, to issue a class of Units designated as Class B Units to (i) any employee, officer or director of the Company or any Company Subsidiary and (ii) any Person, including a consultant or a contractor, whose duties include matters that support the business of the Company and/or any Company Subsidiary, pursuant to the Incentive Plan.
- (b) As of October 19, 2019, the Members have approved the Amended and Restated MEMX Holdings LLC 2018 Profits Interests Plan pursuant to which all Class B Units shall be granted in compliance with Rule 701 of the Securities Act or another applicable exemption (such plan as in effect from time to time, the "Incentive Plan"). The number of Class B Units available for issuance pursuant to such Incentive Plan shall not exceed 21,754,087 Class B Units (subject to adjustment in the event of any Unit split, Unit combination, reorganization, reclassification, recapitalization or the like) minus any Class B Units issued by the Company pursuant to the Members Holdings LLC 2018 Profits Interests Plan approved by the Members of the Company on December 14, 2018 or the Incentive Plan, as applicable, prior to the Effective Date which have not been cancelled, forfeited, repurchased or redeemed and subsequently re-issued. The Company is hereby authorized to negotiate agreements for the issuance of awards pursuant to the

Incentive Plan, which shall include such terms, conditions, rights and obligations as may be determined by the Board, in its sole discretion, and shall be approved by the Board by Supermajority Board Vote, consistent with the terms herein.

- (c) Class B Units shall not have any voting rights other than as required by Applicable Law.
- (d) Class B Units are intended to qualify as "profits interests" under Internal Revenue Service Revenue Procedures 93-27 and 2001-43, and this Agreement shall be interpreted and applied consistently therewith. The Board may restrict distributions with respect to Class B Units to the extent necessary (in the sole discretion of the Board) for such Class B Units to qualify as profits interests.
- (e) (i) Each Class B Unit shall have a "<u>Participation Threshold</u>" determined in accordance with this <u>Section 3.3(e)</u>.
  - (ii) The initial Participation Threshold with respect to a Class B Unit shall reflect the amount that would, but for the limitations set forth in Section 7.2(b), Section 13.3(c)(v) and (vi), Section 13.3(d), or Section 13.3(e) as applied to all Class B Units granted on the date of grant, be distributed to the holder of such Class B Unit (ignoring vesting requirements applicable to such Class B Unit) if, as of the date of grant, all of the Company's assets (including its goodwill and going concern value) had been sold for Fair Market Value and the proceeds distributed in accordance with the terms of this Agreement. The Board shall have the discretion to set any Class B Unit's Participation Threshold to an amount that is greater than the amount determined in the prior sentence.
- 3.4 Other Issuances. In addition to the Units, the Company may, subject to compliance with Section 3.6, Section 8.6(d), Section 9.1 and Section 10.1(b), authorize and issue or sell to any Person any of the following (collectively, "New Interests"): (a) any new type, class or series of Units not otherwise described in this Agreement, which Units may be designated as types, classes or series of the Units but having different rights; and (b) Unit Equivalents. The Board by Supermajority Board Vote is hereby authorized, subject to Section 15.9, to amend this Agreement to reflect such issuance and to fix the relative privileges, preference, duties, liabilities, obligations and rights of any such New Interests, including the number of such New Interests to be issued, the preference (with respect to Distributions, in liquidation or otherwise) over any other Units and any contributions required in connection therewith.

#### 3.5 Limitations on Ownership.

(a) Commencing on the Registration Date and for so long as the Company Controls, directly or indirectly, any Regulated Securities Exchange Subsidiary, except as provided in Sections 3.5(b)(i) and 3.5(b)(ii):

- (i) No Person, either alone or together with its Related Persons, may own, directly or indirectly, of record or beneficially, Units constituting more than forty percent (40%) of any class of Units;
- (ii) No Exchange Member, either alone or together with its Related Persons, may own, directly or indirectly, of record or beneficially, Units constituting more than twenty percent (20%) of any class of Units; and
- (iii) No Person, either alone or together with its Related Persons, at any time may, directly, indirectly or pursuant to any voting trust, agreement, plan or other arrangement, vote or cause the voting of Units or give any consent or proxy with respect to Units representing more than twenty percent (20%) of the voting power of the then issued and outstanding Units, nor may any Person, either alone or together with its Related Persons, enter into any agreement, plan or other arrangement with any other Person, either alone or together with its Related Persons, under circumstances that would result in the Units that are subject to such agreement, plan or other arrangement not being voted on any matter or matters or any proxy relating thereto being withheld, where the effect of such agreement, plan or other arrangement would be to enable any Person, either alone or together with its Related Persons, to vote, possess the right to vote or cause the voting of Units which would represent more than twenty percent (20%) of such voting power.

The parties hereto acknowledge that none of the Members shall be deemed to be in breach of this <u>Section 3.5(a)</u> if such Member is in violation of the limitations set forth above as a result of an action by any other Person (other than such Member or such Member's Affiliates) (including, for the avoidance of doubt, any transfer or surrender of Units by another Member or a redemption of Units by the Company).

- (b) Commencing on the Registration Date and for so long as the Company shall Control, directly or indirectly, any Regulated Securities Exchange Subsidiary, subject to <u>Sections</u> 3.5(c) and 3.5(d):
  - (i) The limitations in <u>Sections 3.5(a)(i)</u> and <u>3.5(a)(iii)</u> above shall not apply in the case of any class of Units that does not have the right by its terms to nominate any Directors or vote on other matters that may require the approval of the holders of voting Units of the Company, if any (other than matters affecting the rights, preferences or privileges of said class of Units); and
  - (ii) The limitations in Sections 3.5(a)(i) and 3.5(a)(iii) above (except with respect to Exchange Members and their Related Persons) may be waived by the Board pursuant to a resolution duly adopted by the Board by Supermajority Board Vote, if, in connection with the taking of such action, the Board adopts a resolution stating that it is the determination of such Board that such action shall not impair the ability of each Regulated Securities Exchange Subsidiary to carry out its functions and responsibilities as an "exchange" under the Exchange Act, and the rules and regulations promulgated thereunder; that it is otherwise in the best interests of the Company, its members and each Regulated

Securities Exchange Subsidiary, and that it shall not impair the ability of the SEC to enforce the Exchange Act and the rules and regulations promulgated thereunder, and such resolution shall not be effective until it is filed with and approved by the SEC. In making the determinations referred to in the immediately preceding sentence, the Board may impose on the Person in question and its Related Persons such conditions and restrictions as it may in its sole discretion deem necessary, appropriate or desirable in furtherance of the objectives of the Exchange Act and the rules and regulations promulgated thereunder, and the governance of any Regulated Securities Exchange Subsidiary.

- (c) Notwithstanding Sections 3.5(b)(i) and 3.5(b)(ii) above, in any case where a Person, either alone or together with its Related Persons, would own or vote more than any of the above percentage limitations upon consummation of any proposed sale, assignment or transfer of the Units, such sale, assignment or transfer shall not become effective until the Board shall have determined, by resolution, that such Person and its Related Persons are not subject to any applicable "statutory disqualification" (within the meaning of Section 3(a)(39) of the Exchange Act).
- (d) Notwithstanding Sections 3.5(b)(i) and 3.5(b)(ii) above, and without giving effect to the same, any Person that either alone or together with its Related Persons proposes to own, directly or indirectly, of record or beneficially, Units constituting more than forty percent (40%) of the outstanding Units of any class, or to exercise voting rights, or grant any proxies or consents with respect to Units constituting more than twenty percent (20%) of the voting power of the then issued and outstanding Units, shall have delivered to the Board of Directors a notice in writing, not less than forty-five (45) days (or any shorter period to which said Board shall expressly consent) before the proposed ownership of such Units, or the proposed exercise of said voting rights or the granting of said proxies or consents, of its intention to do so.
- (e) Notwithstanding anything in this Agreement to the contrary, the provisions of this Agreement shall be construed in a manner such that the Class C Units, the Class D Units and the Common Units together shall be treated as a single class of securities for purposes of this Section 3.5 and Section 3.8.
- 3.6 <u>Notices of Ownership</u>. Commencing on the Registration Date and for so long as the Company shall Control, directly or indirectly, any Registered Securities Exchange Subsidiary:
- (a) Any Person that, either alone or together with its Related Persons, owns, directly or indirectly (whether by acquisition or by a change in the number of Units outstanding), of record or beneficially five percent (5%) or more of the then outstanding Units shall, immediately upon acquiring knowledge of its ownership of five percent (5%) or more of the then outstanding Units, give the Board written notice of such ownership, which notice shall state: (i) such Person's full legal name; (ii) such Person's title or status and the date on which such title or status was acquired; (iii) such Person's (and its Related Person's) approximate ownership interest of the Company; and (iv) whether such Person has the power, directly or indirectly, to direct the management or policies of the Company, whether through ownership of securities, by contract or otherwise, provided that no Member shall be required to provide notice to the Company pursuant to this Section 3.6(a) in connection with the execution of this Agreement.

- shall update such notice promptly after any change in the contents of that notice; <u>provided</u> that no updated notice pursuant to this <u>Section 3.6(b)</u> shall be required to be provided to the Board (i) in the event of an increase or decrease in the ownership percentage so reported of less than one percent (1%) of the then outstanding Units (such increase or decrease to be measured cumulatively from the amount shown on the last such notice), unless any increase or decrease of less than one percent (1%) results in such Person owning more than twenty percent (20%) or more than forty percent (40%) of the Units then outstanding (at a time when such Person previously owned less than such percentages) or such Person owning less than twenty percent (20%) or less than forty percent (40%) of the Units then outstanding (at a time when such Person previously owned more than such percentages); or (ii) in the event the Company issues additional Units or New Interests or takes any other action that dilutes the ownership of such Person, or acquires or redeems Units or takes any other action that increases the ownership of such Person, in each case without any change in the number of Units held by such Person.
- (c) The Board shall have the right to require any Person reasonably believed to be subject to and in violation of this <u>Section 3.6</u> to provide the Company complete information as to all Units owned, directly or indirectly, of record or beneficially, by such Person and its Related Persons and as to any other factual matter relating to the applicability or effect of this <u>Section 3.6</u> as may reasonably be requested of such Person.

### 3.7 Certain Prohibited Actions.

- (a) As set forth in <u>Section 10.1(c)</u>, any Transfer or attempted Transfer of any Units or Unit Equivalents in violation of any provisions of this Agreement shall be null and void, no such Transfer shall be recorded on the Company's books and the purported Transferee in any such Transfer shall not be treated (and the purported Transferor shall continue be treated) as the owner of such Units or Unit Equivalents for all purposes of this Agreement, including without limitation, voting, payment of dividends and distributions with respect to such Units whether upon liquidation or otherwise.
- (b) As set forth in Section 4.7(o), if any Member purports to vote, or to grant any proxy or enter into any agreement, plan or other arrangement relating to the voting of, Units that would violate the provisions of this Agreement, then the Company shall not honor such vote, proxy, agreement, plan or other arrangement to the extent that such provisions would be violated, and any Units subject to that arrangement shall not be entitled to be voted to the extent of such violation.
- (c) Subject to <u>Sections 3.7(a)</u> and <u>3.7(b)</u>, commencing on the Registration Date and for so long as the Company Controls, directly or indirectly, any Regulated Securities Exchange Subsidiary, if any Member purports to Transfer any Units or Unit Equivalents and such Transfer results in a violation of <u>Section 3.5</u>, then the Company shall have the right to, and shall promptly after confirming such violation and to the extent funds are legally available, redeem all of the Units or Unit Equivalents the holding of which by the holder thereof results in a violation of <u>Section 3.5</u> for a price per Unit or Unit Equivalent, as applicable, equal to the Fair Market Value of such Units or Unit Equivalents; <u>provided</u>, that if either such Member or such holder has received written notice

from the Company prior to such Transfer, or a director or officer of such Member (if an entity) or such Member (if an individual) is otherwise actually aware, that such Transfer will result in a violation of Section 3.5, such applicable Units or Unit Equivalents shall be redeemed for a price per Unit or Unit Equivalent, as applicable, equal to the lesser of (a) book value or (b) Fair Market Value of such Units or Unit Equivalents. The number of Units or Unit Equivalents to be redeemed by the Company pursuant to the preceding sentence shall be calculated by the Company after taking into account the fact that immediately upon their redemption such redeemed Units or Unit Equivalents shall become treasury shares and shall no longer be deemed to be outstanding. Written notice shall be given by the Company to the holders of the redeemable Units or Unit Equivalents at the address of such holders appearing on the books of the Company, which notice shall specify a date for redemption of such Units or Unit Equivalents that shall be not less than ten (10) days nor more than thirty (30) days from the date of such notice. Any Units or Unit Equivalents which have been so called for redemption shall not be deemed outstanding Units or Unit Equivalents after the date on which written notice of redemption has been given to the holders of those Units or Unit Equivalents if a sum sufficient to redeem such Units or Unit Equivalents shall have been irrevocably deposited or set aside to pay the redemption price to the holders of the Units or Unit Equivalents. From and after the applicable redemption date (unless the Company shall default in providing funds for the payment of the redemption price) the Units or Unit Equivalents which have been redeemed by the Company as aforesaid shall become treasury shares, and all rights of the holder of such redeemed Units or Unit Equivalents as a Member of the Company associated with such Units or Unit Equivalents (except the right to receive from the Company the applicable redemption price against delivery to the Company of evidence of ownership of such Units or Unit Equivalents) shall cease. Written notice shall be given by the Company to all holders of Units of any redemption by the Company (including, without limitation, a redemption pursuant to this Section 3.7(c)) not more than ten (10) days after consummation of the applicable redemption, which notice shall specify the number of Units of each class outstanding after such redemption. In the event that any redemption or other action by the Company has resulted in any Member owning such number of Units or Unit Equivalents that is in violation of the provisions of Section 3.5, the Company shall have the right to and shall promptly after confirming such violation, redeem such Units or Unit Equivalents for a price per Unit or Unit Equivalent, as applicable, equal to Fair Market Value, and otherwise pursuant to the provisions of this Section 3.7(c) and subject to Section 3.8.

3.8 <u>Certain Special Transfer Rights</u>. Commencing on the Registration Date and for so long as the Company shall Control, directly or indirectly, any Regulated Securities Exchange Subsidiary, if a Member which is, or any Affiliate of which is, an Exchange Member, either alone or together with its Related Persons, owns, directly or indirectly, of record or beneficially, Units constituting more than twenty percent (20%) of any class of Units, such Member may Transfer the number of Units which account for the excess over such twenty percent (20%) ownership limitation to one or more Members, in which case Sections 10.1(a)(i) and 10.3 shall not apply.

## 3.9 Restricted Economic Elections.

(a) A Member may give an irrevocable notice of its election (the "Restricted Economic Election") to be treated for purposes of this Agreement as a "Restricted Economic Member," such that the maximum percentage of the aggregate economic interests attributable to the

Units that such Member may own is such percentage as is designated in such Member's Restricted Economic Election (such Restricted Economic Member's "<u>Maximum Aggregate Economic</u> Interest"), by delivering written notice to the Company in the form attached as Exhibit E hereto.

- (b) Upon receiving any Restricted Economic Election, the Company shall promptly (but no later than five (5) Business Days following receipt of such notice) notify the other Members in writing upon any such election having been made by a Member, including the Maximum Aggregate Economic Interest included in such election.
- (c) In the event a Member makes a Restricted Economic Election and there is any distribution on, or redemption of Units or other equity securities of the Company that would cause such electing Member to exceed its Maximum Aggregate Economic Interest, then the amount (and only such amount) of such distribution or redemption that would cause such Member's economic interest to exceed the Maximum Aggregate Economic Interest shall instead be redistributed to other Members of the Company pursuant to Section 7.3 (but excluding from such Section 7.3 distribution any such Member who has exceeded its Maximum Aggregate Economic Interest).

### 3.10 Specification of Maximum Voting Percentage.

- Election. On or at any time following the Effective Date, (i) any Class A (a) Member may give notice to the Company of its election to specify its respective maximum Voting Class A Voting Percentage (the "Maximum Voting Class A Voting Percentage"), and (ii) any Class C Member and/or Class D Member may give notice to the Company of its election to specify its respective maximum Class C-1/D-1 Voting Percentage (the "Maximum Class C-1/D-1 Voting Percentage"). Any such election shall be referred to in this Agreement as a "Restricted Voting Election" and any Member which has given any such Restricted Voting Election shall be deemed a "Restricted Voting Member" under this Agreement. Any such Restricted Voting Election made by a Restricted Voting Member shall be irrevocable (except as set forth in Section 3.10(f)); provided, that a Restricted Voting Member may subsequently elect to specify (A) a Maximum Voting Class A Voting Percentage that is greater than its then-applicable Maximum Voting Class A Voting Percentage, or (B) a Maximum Class C-1/D-1 Voting Percentage that is greater than its thenapplicable Maximum Class C-1/D-1 Voting Percentage, as applicable, in each case solely in order to give effect to the voting power associated with a purchase of additional Units by delivering an amended notice of its Restricted Voting Election to the Company, but no such election is permissible in connection with the receipt of additional Units by means of any subdivision, distribution in kind or other circumstance not constituting such a purchase of additional Units; provided further, that in no event shall any change by a Restricted Voting Member to its Maximum Voting Class A Voting Percentage or Maximum Class C-1/D-1 Voting Percentage result in or allow Units that were or are held or controlled by such Member to be or become voting Units, including through conversion or transfer. Elections pursuant to this Section 3.10(a) shall be in the form attached as Exhibit F hereto.
- (b) <u>Amendment to Members Schedule</u>. Upon receiving an election from the applicable Restricted Voting Member pursuant to <u>Section 3.10(a)</u>, the Company shall promptly (i)

amend the Members Schedule to reflect such electing Restricted Voting Member's specified Maximum Voting Class A Voting Percentage or Maximum Class C-1/D-1 Voting Percentage, as applicable, and any resulting changes in Unit holdings resulting from any automatic conversion of Units arising pursuant to Sections 3.10(d)(i) and 3.10(d)(ii), and notify all Class A Members, all Class C Members and all Class D Members of such election, which notice shall include a copy of the amended Members Schedule.

(c) <u>Maximum Voting Percentages as of the Effective Date</u>. Notwithstanding the foregoing, the Members understand and agree that, as of the Effective Date, (i) the Maximum Voting Class A Voting Percentage applicable to any particular Class A Member or (ii) the Maximum Class C-1/D-1 Voting Percentage applicable to any particular Class C Member and/or Class D Member shall be as specified in the Members Schedule as of the Effective Date, and (A) any Class A Member with a Maximum Voting Class A Voting Percentage, or (B) any Class C Member and/or Class D Member with a Maximum Class C-1/D-1 Voting Percentage, as applicable, specified in the Members Schedule as of the Effective Date shall be deemed a Restricted Voting Member under this Agreement.

## (d) <u>Effect of Maximum Voting Percentage</u>.

- (i) <u>Automatic Conversion</u>. Notwithstanding anything herein to the contrary, if a Restricted Voting Member would otherwise be deemed to own, control, or have the power to vote (for any reason):
  - Voting Member's Voting Class A Voting Percentage to exceed its Maximum Voting Class A Voting Percentage, then the minimum number of Voting Class A Units held by such Restricted Voting Member necessary to cause such Restricted Voting Member's Voting Class A Voting Percentage to equal its Maximum Voting Class A Voting Percentage shall automatically, immediately, and without any action on the part of such Restricted Voting Member, be converted, on a one-for-one basis, into Nonvoting Class A-1 Units (in the case of Class A-1 Units) or Nonvoting Class A-2 Units (in the case of Class A-2 Units); provided that in case of such conversion a Class A Member may, by written notice to the Company concurrently with or promptly after such conversion, elect the number of Class A-1 Units that convert into Nonvoting Class A-1 Units; or
  - (B) an aggregate number of Class C-1 Units and Class D-1 Units that causes such Restricted Voting Member's Class C-1/D-1 Voting Percentage to exceed its Maximum Class C-1/D-1 Voting Percentage, then (1) the number of Class C-1 Units, (2) if such Restricted Voting Member does not hold any Class C-1 Units, the number of Class D-1 Units, or (3) if the conversion of all of the Class C-1 Units held by such Restricted Voting Member does not reduce the Restricted Voting Member's Class C-1/D-1 Voting Percentage below its Maximum Class C-1/D-1 Voting Percentage, the number of Class D-1 Units, held by such Restricted Voting

Member necessary to cause such Restricted Voting Member's Class C-1/D-1 Voting Percentage to equal its Maximum Class C-1/D-1 Voting Percentage shall automatically, immediately, and without any action on the part of such Restricted Voting Member, be converted, on a one-for-one basis, into Class C-2 Units (in case of Class C-1 Units) or Class D-2 Units (in case of Class D-1 Units), as and where applicable.

The automatic conversions pursuant to clauses (A) or (B) immediately above, as applicable, shall occur if the Maximum Voting Class A Voting Percentage or Maximum Class C-1/D-1 Voting Percentage, as applicable, of a Restricted Voting Member would be exceeded for any reason, including due to (1) a Transfer, (2) any change in the overall number of thenoutstanding Voting Class A Units, Class C-1 Units or Class D-1 Units, as applicable, for any reason, including by conversion (automatic or otherwise), redemption, or repurchase of Units held by other Members, (3) the acquisition by such Restricted Voting Member of any additional Voting Class A Units, Class C-1 Units or Class D-1 Units, as applicable, or (4) an election by any Class A Member, Class C Member or Class D Member to specify or amend a Maximum Voting Class A Voting Percentage or a Maximum Class C-1/D-1 Voting Percentage, as applicable. Except as expressly provided in this Section 3.10, any conversion of Units pursuant to this Section 3.10(d)(i) shall be irrevocable.

- (ii) The Company shall, promptly upon the occurrence of any automatic conversion pursuant to Section 3.10(d)(i), amend the Members Schedule to reflect changes in ownership of Units resulting therefrom; provided, that the failure to make such amendments shall not affect such automatic conversion, which shall be deemed to be effective notwithstanding any such failure.
- (iii) When calculating a Restricted Voting Member's number of Class A-1 Units, Class A-2 Units, Class C-1 Units or Class D-1 Units solely for purposes of Section 3.10(d)(i), a Restricted Voting Member shall be deemed to continue to control any Class A-1 Units, Class A-2 Units, Class C-1 Units and Class D-1 Units that such Restricted Voting Member has Transferred other than any Class A-1 Units, Class A-2 Units, Class C-1 Units and Class D-1 Units that were Transferred by such Restricted Voting Member in connection with a Permitted Regulatory Transfer.
- (iv) If any conversion pursuant to this <u>Section 3.10</u> would reasonably be expected to result in adverse legal, compliance or regulatory consequences to any Class A Member, Class C Member, Class D Member or an Affiliate of any such Class A Member, Class C Member or Class D Member as determined in good faith and based on the advice of counsel (which may be in-house counsel) to such impacted Member, the Company shall use commercially reasonable efforts to work with the impacted Member to structure such conversion to eliminate or minimize such adverse legal, compliance or regulatory consequences.
- (e) <u>Limited Conversion of Nonvoting Class A Units, Class C-2 Units and Class</u>

  <u>D-2 Units.</u>

- Upon any Transfer of Nonvoting Class A Units, Class C-2 Units or Class D-2 Units (other than a Transfer to an Affiliate of the Transferor or to another Restricted Voting Member): (A) pursuant to a widespread public distribution; (B) to the Company; (C) pursuant to a private placement in which no one Person acquires the right to purchase 2% or more of any class of voting securities of the Company; (D) pursuant to an assignment to a single Person (e.g., a broker or investment banker) for the purpose of conducting a widespread public distribution on behalf of a Restricted Voting Member; or (E) pursuant to a transfer to a Person who would control more than fifty percent (50%) of the voting securities of the Company without giving effect to the Nonvoting Class A Units, Class C-2 Units or Class D-2 Units transferred by a Restricted Voting Member (each of clauses (A), (B), (C), (D) or (E), a "Permitted Regulatory Transfer"), such transferred Nonvoting Class A Units, Class C-2 Units or Class D-2 Units shall automatically, and without any action by the Company or any Person, be converted, on a one-for-one basis, into Class A-1 Units (in the case of Nonvoting Class A-1 Units), Class A-2 Units (in the case of Nonvoting Class A-2 Units), Class C-1 Units (in the case of Class C-2 Units) or Class D-1 Units (in case of Class D-2 Units) and the transferred Nonvoting Class A Units, Class C-2 Units, or Class D-2 Units shall thereby cease to exist; provided, that, if the Transferee is a Restricted Voting Member, such conversion shall occur only to the extent that the Transferee's resulting (1) Voting Class A Voting Percentage does not exceed its specified Maximum Voting Class A Voting Percentage, or (2) Class C-1/D-1 Voting Percentage does not exceed its specified Maximum Class C-1/D-1 Voting Percentage, as applicable.
- (ii) If the Company issues any new Units or Unit Equivalents and, due to such issuance, the Voting Class A Units, Class C-1 Units or Class D-1 Units held by a Restricted Voting Member reflect:
  - (A) a Voting Class A Voting Percentage that is less than such Restricted Voting Member's Voting Class A Voting Percentage (the "Prior Voting Class A Voting Percentage") immediately prior to such issuance, then such Restricted Voting Member may, by delivery of a notice to the Company (each such notice, a "Voting Conversion Notice"), require the conversion of a number of its Nonvoting Class A-1 Units into Class A-1 Units, or its Nonvoting Class A-2 Units into Class A-2 Units, as the case may be, such that the number of Voting Class A Units held by such Restricted Voting Member after such conversion represents (but does not exceed) the Prior Voting Class A Voting Percentage of such Restricted Voting Member or, if applicable, a lesser Voting Class A Voting Percentage; or
  - (B) a Class C-1/D-1 Voting Percentage that is less than such Restricted Voting Member's Class C-1/D-1 Voting Percentage (the "Prior Class C-1/D-1 Voting Percentage") immediately prior to such issuance, then such Restricted Voting Member may, by delivery of a Voting Conversion Notice, require the conversion of a number of its Class C-2 Units into Class C-1 Units and/or Class D-2 Units into Class D-1 Units, as applicable, such that the aggregate number of Class C-1 Units and/or Class D-1 Units held by such Restricted Voting Member after such

conversion represents (but does not exceed) the Prior Class C-1/D-1 Voting Percentage of such Restricted Voting Member or, if applicable, a lesser Class C-1/D-1 Voting Percentage.

Any conversion pursuant to clause (A) or (B) immediately above is referred to herein as a "Permitted Anti-Dilution Conversion." The (1) Nonvoting Class A Units specified in a Voting Conversion Notice shall automatically convert into Class A-1 Units, or Class A-2 Units, as the case may be, and (2) Class C-2 Units or Class D-2 Units, as applicable, specified in a Voting Conversion Notice shall automatically convert into Class C-1 Units (with respect to Class C-2 Units) or Class D-1 Units (with respect to Class D-2 Units), as applicable, on the date of receipt of the applicable Voting Conversion Notice unless such Voting Conversion Notice states that the applicable Permitted Anti-Dilution Conversion is to become effective at a later date, or when any conditions specified in such Voting Conversion Notice have been fulfilled, in which case such Permitted Anti-Dilution Conversion shall become effective at such a later date, or when such conditions have been fulfilled, as applicable.

- (iii) Upon a Permitted Regulatory Transfer or Permitted Anti-Dilution Conversion, as applicable, the Company shall promptly amend the Members Schedule to reflect any changes resulting from the applicable conversion of Nonvoting Class A Units into Class A-1 Units or Class A-2 Units, as the case may be, of Class C-2 Units into Class C-1 Units, or of Class D-2 Units into Class D-1 Units, as applicable; <u>provided</u>, that the failure to make such amendments shall not affect such conversion, which shall be deemed to be effective notwithstanding any such failure.
- Exception to Irrevocable Election. Notwithstanding that a Maximum Voting Class A Voting Percentage election or a Maximum Class C-1/D-1 Voting Percentage election, as applicable, made by a Restricted Voting Member is otherwise irrevocable, such Restricted Voting Member may, with a written determination by the Federal Reserve or its staff that it may increase its Maximum Voting Class A Voting Percentage or Maximum Class C-1/D-1 Voting Percentage, or reverse its Restricted Voting Election, without being deemed to control or presumed to control the Company for purposes of the BHCA (provided, that no Restricted Voting Member shall ever be obligated to seek such approval), upon written notice to the Company, increase its Maximum Voting Class A Voting Percentage or Maximum Class C-1/D-1 Voting Percentage, as applicable, or reverse its election to be treated as a Restricted Voting Member in its entirety (but only to the extent consistent with such written determination by the Federal Reserve or its staff) and, following such notice, exchange some or all of its Nonvoting Class A-1 Units for newly-issued Class A-1 Units, or of its Nonvoting Class A-2 Units for newly issued Class A-2 Units, or of its Class C-2 Units for newly-issued Class C-1 Units, or of its Class D-2 Units for newly-issued Class D-1 Units, as applicable. The Company shall promptly amend the Members Schedule to reflect such exchange; provided, that the failure to make such amendments shall not affect such exchange, which shall be deemed to be effective notwithstanding any such failure.
- (g) <u>BHCA Impact on Transferees</u>. Notwithstanding anything to the contrary in this Agreement, (i) in no event shall a Restricted Voting Member together with its Transferees

(other than Transferees receiving Units pursuant to a Permitted Regulatory Transfer) be entitled to vote Voting Class A Units representing more than such Restricted Voting Member's Maximum Voting Class A Voting Percentage of the voting power of the Class A Units, except as set forth in Sections 4.7(a), 4.7(b), 4.7(h) and 4.7(i), and (ii) in no event shall a Restricted Voting Member together with its Transferees (other than Transferees receiving Units pursuant to a Permitted Regulatory Transfer) be entitled to vote Class C-1 Units and/or Class D-1 Units representing more than such Restricted Voting Member's Maximum Class C-1/D-1 Voting Percentage of the voting power of the Class C-1 Units and/or Class D-1 Units, except as set forth in Sections 4.7(c), 4.7(d), 4.7(e), 4.7(f), 4.7(g) and 4.7(i).

- (h) <u>Construction</u>. For the purposes of this Agreement, (i) references to Class A Units owned by a Restricted Voting Member shall include all Class A Units Transferred to another Person or Persons by such Restricted Voting Member after making its Maximum Voting Class A Voting Percentage election (excluding any Permitted Regulatory Transfers), and (ii) references to Class C Units and/or Class D Units owned by a Restricted Voting Member shall include all Class C Units and/or Class D Units Transferred to another Person or Persons by such Restricted Voting Member after making its Maximum Class C-1/D-1 Voting Percentage election (excluding any Permitted Regulatory Transfers).
- (i) <u>Construction Consistent with BHCA</u>. Notwithstanding anything herein to the contrary, the provisions of this Agreement shall be construed in a manner such that (i) the Voting Class A Units together shall be treated as a single class of voting securities for purposes of the BHCA; (ii) the Class C-1 Units, Class D-1 Units and Voting Common Units together shall be treated as a single class of voting securities for purposes of the BHCA; (iii) no Class C-2 Units, Class D-2 Units, Nonvoting Class A Units or Nonvoting Common Units shall be treated as voting securities for purposes of the BHCA; (iv) no Restricted Voting Member is at any time deemed to own, control, or have the power to vote any class of voting securities of the Company exceeding its Maximum Voting Class A Voting Percentage or Maximum Class C-1/D-1 Voting Percentage, as applicable; and (iv) the terms "control," "class," "voting securities," and provisions related to Permitted Regulatory Transfers in this <u>Section 3.10</u> shall be interpreted and applied in a manner consistent with the BHCA.

### 3.11 <u>Class C Unit and Class D Unit Conversion</u>.

- (a) <u>Optional Conversion of Class C Units and Class D Units</u>. The Class C Units and the Class D Units shall have the conversion rights set forth in <u>Exhibit G</u>.
- (b) <u>Mandatory Conversion of Class C Units and Class D Units</u>. Subject to <u>Sections 3.10(i)</u> and <u>3.11(c)</u>, immediately upon the consummation of a Qualified Public Offering, all then-outstanding Class C Units and all then-outstanding Class D Units shall, without any further action by any holder of the Class C Units or the Class D Units, be converted automatically into such number of fully paid and nonassessable Common Units as is determined:
  - (i) with respect to the Class C Units, by dividing (A) the sum of the applicable Class C Unit Original Purchase Price, plus, as applicable, an amount equal to the

sum of any declared but unpaid Distributions therefor, by (B) the then-applicable Class C Unit Conversion Price; and

- (ii) with respect to the Class D Units, by dividing (A) the sum of the applicable Class D Unit Original Purchase Price, <u>plus</u>, as applicable, an amount equal to the sum of any declared but unpaid Distributions therefor, by (B) the then-applicable Class D Unit Conversion Price.
- (c) <u>Conversion of Class C-1 Units and Class C-2 Units</u>. In the event of any conversion to Common Units of any Class C Units, Class C-1 Units shall be converted into Voting Common Units, and Class C-2 Units shall be converted into Nonvoting Common Units.
- (d) <u>Conversion of Class D-1 Units and Class D-2 Units</u>. In the event of any conversion to Common Units of any Class D Units, Class D-1 Units shall be converted into Voting Common Units, and Class D-2 Units shall be converted into Nonvoting Common Units.

## 3.12 Certification of Units.

- (a) The Board in its sole discretion may, but shall not be required to, issue certificates to the Members representing the Units held by such Member.
- (b) In the event that the Board shall issue certificates representing Units in accordance with Section 3.12(a), then in addition to any other legend required by Applicable Law, all certificates representing issued and outstanding Units shall bear a legend substantially in the following form:

THE UNITS REPRESENTED BY THIS CERTIFICATE ARE SUBJECT TO THE EIGHTH AMENDED AND RESTATED LIMITED LIABILITY COMPANY AGREEMENT AMONG MEMX HOLDINGS LLC AND ITS MEMBERS, A COPY OF WHICH IS ON FILE AT THE PRINCIPAL EXECUTIVE OFFICE OF MEMX HOLDINGS LLC. NO TRANSFER, SALE, ASSIGNMENT, PLEDGE, HYPOTHECATION OR OTHER DISPOSITION OF THE UNITS REPRESENTED BY THIS CERTIFICATE MAY BE MADE EXCEPT IN ACCORDANCE WITH THE PROVISIONS OF SUCH LIMITED LIABILITY COMPANY AGREEMENT.

THE UNITS REPRESENTED BY THIS CERTIFICATE HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933, AS AMENDED, OR UNDER ANY OTHER APPLICABLE SECURITIES LAWS AND MAY NOT BE TRANSFERRED, SOLD, ASSIGNED, PLEDGED, HYPOTHECATED OR OTHERWISE DISPOSED EXCEPT (A) PURSUANT TO A REGISTRATION STATEMENT EFFECTIVE UNDER SUCH ACT AND LAWS, OR (B) PURSUANT TO AN EXEMPTION FROM REGISTRATION THEREUNDER.

## ARTICLE IV. MEMBERS

### 4.1 Admission of New Members.

- (a) New Members may be admitted from time to time (i) in connection with an issuance of Units by the Company, subject to compliance with the provisions of <u>Section 8.6(d)</u>, <u>Section 9.1</u> and <u>Section 10.1(b)</u>, as applicable, and (ii) in connection with a Transfer of Units, subject to compliance with the provisions of <u>Article X</u>.
- (b) In order for any Person not already a Member of the Company to be admitted as a Member, whether pursuant to an issuance or Transfer of Units, such Person shall have executed and delivered to the Company a written undertaking substantially in the form of the Adherence Agreement. Upon the amendment of the Members Schedule by the Board (which amendment may not be unreasonably conditioned or delayed), and the satisfaction of any other applicable conditions set forth in this Agreement, including, if a condition, the receipt by the Company of payment for the issuance of the applicable Units, such Person shall be admitted as a Member and deemed listed as such on the books and records of the Company and thereupon shall be issued his, her or its Units. The Board shall also adjust the Capital Accounts of the Members as necessary in accordance with Section 5.2.
- 4.2 <u>Representations and Warranties of Members</u>. By execution and delivery of this Agreement or an Adherence Agreement, as applicable, each of the Members, as of the date such Member acquired Units, represents and warrants to the Company and acknowledges that:
- (a) Such Member understands that the Units have not been registered under the Securities Act or the securities laws of any other jurisdiction, are issued in reliance upon federal and state exemptions for transactions not involving a public offering and cannot be disposed of unless (i) they are subsequently registered or exempted from registration under the Securities Act and (ii) the provisions of this Agreement have been complied with;
- (b) If such Member did not receive its Units pursuant to an equity incentive plan (including the Incentive Plan), such Member is an "accredited investor" within the meaning of Rule 501 promulgated under the Securities Act, and agrees that it shall not take any action that could have an adverse effect on the availability of the exemption from registration provided by Rule 506 promulgated under the Securities Act with respect to the offer and sale of the Units;
- (c) Such Member's Units are being acquired for its own account solely for investment and not with a view to resale or distribution thereof;
- (d) Such Member has had the opportunity to conduct its own independent review and analysis of the business, operations, assets, liabilities, results of operations, financial condition and prospects of the Company and the Company Subsidiaries and such Member acknowledges that it has been provided adequate access to the personnel, properties, premises and records of the Company and the Company Subsidiaries for such purpose;

- (e) The determination of such Member to acquire Units has been made by such Member independent of any other Member and independent of any statements or opinions as to the advisability of such purchase or as to the business, operations, assets, liabilities, results of operations, financial condition and prospects of the Company and the Company Subsidiaries that may have been made or given by any other Member or by any agent or employee of any other Member;
- (f) Such Member has such knowledge and experience in financial and business matters and is capable of evaluating the merits and risks of an investment in the Company and making an informed decision with respect thereto;
- (g) Such Member is able to bear the economic and financial risk of an investment in the Company for an indefinite period of time;
- (h) The execution, delivery and performance of this Agreement have been duly authorized by such Member and do not require such Member to obtain any consent or approval that has not been obtained and do not contravene or result in a default in any material respect under any provision of any law or regulation applicable to such Member or other governing documents or any agreement or instrument to which such Member is a party or by which such Member is bound;
- (i) This Agreement is valid, binding and enforceable against such Member in accordance with its terms, except as may be limited by bankruptcy, insolvency, reorganization, moratorium, and other similar laws of general applicability relating to or affecting creditors' rights or general equity principles (regardless of whether considered at law or in equity); and
- (j) Neither the issuance of any Units to such Member nor any provision contained herein shall entitle such Member to remain in the employment of the Company or any Company Subsidiary or affect the right of the Company or any Company Subsidiary to terminate such Member's employment at any time for any reason, other than as otherwise provided in such Member's employment agreement or other similar agreement with the Company or Company Subsidiary, if applicable.
- 4.3 <u>No Personal Liability</u>. Except as otherwise provided in the Delaware Act, by Applicable Law, no Member shall be obligated personally for any debt, obligation or liability of the Company or of any Company Subsidiaries or other Members, whether arising in contract, tort or otherwise, solely by reason of being a Member. No Member or its Affiliates shall have any responsibility to restore any negative balance in its Capital Account or to contribute to or in respect of the liabilities or obligations of the Company or of any Company Subsidiary or return Distributions made by the Company not in error.
- 4.4 <u>No Withdrawal</u>. A Member shall not cease to be a Member as a result of the Bankruptcy of such Member or as a result of any other events specified in § 18-304 of the Delaware Act, provided that the Class B Units may be redeemable or forfeitable pursuant to the terms of the Incentive Plan. So long as a Member continues to hold any Units, such Member shall not have the ability to withdraw or resign as a Member prior to the dissolution and winding up of the Company

and any such withdrawal or resignation or attempted withdrawal or resignation by a Member prior to the dissolution or winding up of the Company shall be null and void. As soon as any Person who is a Member ceases to hold any Units, such Person shall no longer be a Member.

4.5 <u>Death</u>. The death of any Member shall not cause the dissolution of the Company. In such event the Company and its business shall be continued by the remaining Member or Members and the Units owned by the deceased Member shall automatically be Transferred to such Member's heirs; <u>provided</u>, that within a reasonable time after such Transfer, the applicable heirs shall sign a written undertaking substantially in the form of the Adherence Agreement.

## 4.6 No Voting Rights or Power to Bind the Company.

- (a) Except as required by Sections 4.7(a), 4.7(b), 4.7(c), 4.7(d), 4.7(e), 4.7(f), 4.7(g), 4.7(h), 4.7(i), 15.9 or Applicable Law, (i) Members shall not have voting or management rights and the management of the Company shall be vested in the Board and the Officers of the Company, (ii) Members shall not have regular meetings, and (iii) no Member shall have the power to act for or on behalf of, or to bind, the Company.
- (b) If Applicable Law requires that the Members vote on a particular matter, Members shall vote together as a single class (other than the Class B Members, the holders of Nonvoting Class A Units, the holders of Class C-2 Units, the holders of Class D-2 Units, and the holders of Nonvoting Common Units (if any) which shall nevertheless not vote unless Applicable Law, as applicable, requires that they also vote).
- (c) Neither the Class A Members (including the holders of Class A-1 Units, the holders of Class A-2 Units and the holders of the Nonvoting Class A Units), the Class B Members, the Class C Members (including the holders of Class C-1 Units and the holders of Class C-2 Units), the Class D Members (including the holders of Class D-1 Units and the holders of Class D-2 Units), nor the Common Members (including the holders of Voting Common Units and the holders of Nonvoting Common Units), shall be entitled to vote as a separate class on a particular matter, unless specifically provided for in this Agreement.

#### 4.7 Voting Rights of Members; Meetings of Members.

- (a) From and after the Effective Date, the following actions shall not be effected without the approval of a majority of the then-outstanding Voting Class A Units, voting together as a single class (it being understood that each such action shall also require approval of the Board by Supermajority Board Vote):
  - (i) subject to <u>Sections 7.2(b)</u>, <u>7.3</u> and <u>13.3</u>, approval of any Distributions of profits or capital of the Company to the Members (other than Tax Advances);
  - (ii) approval of a transaction to which the Company is a party and which results in a Change of Control;

- (iii) any liquidation, dissolution or winding up of any Subsidiary of the Company (other than a Regulated Securities Exchange Subsidiary) and, if applicable, the related appointment of a liquidating trustee; and
- (iv) commencement, filing or initiation of any proceeding relating to voluntary or involuntary bankruptcy or insolvency with respect to the Company.
- (b) From and after the Effective Date, any waiver or amendment of any provision of this Agreement which would significantly and adversely affect the rights, preferences, powers or privileges of the Class A-1 Units and Class A-2 Units shall not be effected without the approval of the majority of the then-outstanding Voting Class A Units, voting together as a single class (it being understood that each such action shall also require approval of the Board by Supermajority Board Vote).
- (c) From and after the Effective Date, the following actions shall not be effected without the approval of a majority of the then-outstanding Class C-1 Units, Class D-1 Units and Voting Common Units, voting together as a single class (it being understood that each such action shall also require approval of the Board by Supermajority Board Vote):
  - (i) subject to <u>Sections 7.2(b)</u>, <u>7.3</u> and <u>13.3</u>, approval of any Distributions of profits or capital of the Company to the Members (other than Tax Advances);
  - (ii) approval of a transaction to which the Company is a party and which results in a Change of Control;
  - (iii) any liquidation, dissolution or winding up of any Subsidiary of the Company (other than a Regulated Securities Exchange Subsidiary) and, if applicable, the related appointment of a liquidating trustee; and
  - (iv) commencement, filing or initiation of any proceeding relating to voluntary or involuntary bankruptcy or insolvency with respect to the Company.
- (d) From and after the Effective Date, (i) any waiver or amendment of any provision of this Agreement which would significantly and adversely affect the rights, preferences, powers or privileges of the Class C-1 Units shall not be effected without the approval of a majority of the then-outstanding Class C-1 Units (it being understood that each such action shall also require approval of the Board by Supermajority Board Vote) and (ii) any waiver or amendment of any provision of this Agreement which would significantly and adversely affect the rights, preferences, powers or privileges of the Class D-1 Units shall not be effected without the approval of a majority of the then-outstanding Class D-1 Units (it being understood that each such action shall also require approval of the Board by Supermajority Board Vote).
- (e) From and after the Effective Date, the following actions (which shall be construed in a manner consistent with 12 C.F.R. §225.2(q)(2)(i)) shall not be effected without the approval of the majority of the then-outstanding Class C-1 Units and Class C-2 Units, voting

together as a single class (it being understood that each such action shall also require approval of the Board by Supermajority Board Vote): any exchange, reclassification or cancellation (whether by merger, consolidation or otherwise) or modification of the terms of all or part of the Class C Units which exchange, reclassification, cancellation or modification, as applicable, significantly and adversely affects the rights or preferences of the Class C Units.

- (f) From and after the Effective Date, the following actions (which shall be construed in a manner consistent with 12 C.F.R. §225.2(q)(2)(i)) shall not be effected without the approval of the majority of the then-outstanding Class D-1 Units and Class D-2 Units, voting together as a single class (it being understood that each such action shall also require approval of the Board by Supermajority Board Vote): any exchange, reclassification or cancellation (whether by merger, consolidation or otherwise) or modification of the terms of all or part of the Class D Units which exchange, reclassification, cancellation or modification, as applicable, significantly and adversely affects the rights or preferences of the Class D Units.
- (g) From and after the Effective Date, the following actions (which shall be construed in a manner consistent with 12 C.F.R. §225.2(q)(2)(i)) shall not be effected without the approval of the majority of the then-outstanding Class C-1 Units, Class C-2, Class D-1 Units, and Class D-2 Units, voting together as a single class (it being understood that each such action shall also require approval of the Board by Supermajority Board Vote): any issuance of Units or Unit Equivalents of the Company that have (i) a preference in respect of Distributions or return of capital that is senior to the holders of the Class C Units and the holders of the Class D Units or (ii) no right to convert into Common Units.
- (h) From and after the Effective Date, the following actions (which shall be construed in a manner consistent with 12 C.F.R. §225.2(q)(2)(i)) shall not be effected without the approval of the majority of the then-outstanding Class A-1 Units, Class A-2 Units, Nonvoting Class A-1 Units and Nonvoting Class A-2 Units, voting together as a single class (it being understood that each such action shall also require approval of the Board by Supermajority Board Vote):
  - (i) any issuance of Units or Unit Equivalents of the Company that have a preference in respect of Distributions or return of capital that is senior to the holders of the Class A Units; and
  - (ii) any exchange, reclassification or cancellation (whether by merger, consolidation or otherwise) or modification of the terms of all or part of the Class A Units which exchange, reclassification, cancellation or modification, as applicable, significantly and adversely affects the rights or preferences of the Class A Units.
- (i) From and after the Effective Date, any liquidation, dissolution or winding up of the Company (which shall be construed in a manner consistent with 12 C.F.R. §225.2(q)(2)(i)) shall not be effected without the approval of the majority of the then-outstanding Class A-1 Units, Class A-2 Units, Nonvoting Class A-1 Units, Nonvoting Class A-2 Units, Class C-1 Units, Class C-2 Units, Class D-1 Units, Class D-2 Units, Voting Common Units and Nonvoting Common Units,

voting together as a single class (it being understood that each such action shall also require approval of the Board by Supermajority Board Vote).

- (j) Meetings of Members may be called by (i) by a simple majority of the thenserving Directors, (ii) by the Class A Members holding, in the aggregate, at least ten percent (10%) of the aggregate then-outstanding Class A Units, or (iii) by the Class C Members and/or Class D Members holding, in the aggregate, at least twenty percent (20%) of the aggregate then-outstanding Class C Units and Class D Units.
- (k) Written notice stating the place, day and hour of the meeting and describing the purpose for which the meeting is called shall be delivered to each Member entitled to such notice not fewer than fifteen (15) Business Days nor more than sixty (60) calendar days before the date of the meeting. The notice of any meeting of Members shall include an agenda specifying in reasonable detail the matters to be discussed at such meeting. Except as required by Applicable Law, Class B Members, in their capacity as such, shall not be entitled to notice of, and shall not have a right to attend, any meetings of Members.
- (l) Any Member that is entitled to notice of a meeting may waive such notice in writing, whether before or after the time of such meeting. Attendance by a Member at a meeting of Members shall constitute a waiver of notice of such meeting by such Member, except when such Member attends such meeting for the express purpose of objecting, at the beginning of such meeting, to the transaction of any business at such meeting because such meeting is called or convened in violation of this Agreement or any Applicable Law.
- (m) Members may participate in and hold a meeting by means of telephone conference or similar communications equipment by means of which all persons participating in the meeting can hear and communicate with each other. Participation in a meeting by such means shall constitute presence in person at the meeting, except where a Member participates in the meeting for the express purpose of objecting to the transaction of any business on the ground that such meeting was called or convened in violation of this Agreement or any Applicable Law.
- (n) If any action is required by Applicable Law to be taken by the Members, such action to be taken at any meeting of Members may be taken without a meeting if the action is taken in writing (which may be via email communication) by consent of such number of Members as would otherwise be required to approve such action, and the writing or writings are filed with the minutes of the meeting of Members (or, where required by Applicable Law, a class or series thereof, as applicable). For purposes of the foregoing, an action shall be deemed to have been taken in writing via email communication if (i) an email communication is sent by an Officer to all Members entitled to vote on the matter at issue clearly specifying the action to be taken, (ii) the number of Members required to approve the matter at issue respond to the Officer's email with an unambiguous approval of such matter, and (iii) the Officer's email and all such responses are filed with the minutes of the meetings of Members.
- (o) The presence in person or by proxy of Members holding at least fifty percent (50%) of the then-outstanding Class A Units, Class C Units and Class D Units (considered in the

aggregate) shall constitute a quorum for the transaction of business for such matter. Any meeting regarding such a matter may be adjourned from time to time by the holders of a majority of the votes properly cast upon such matter, whether or not a quorum is present, and the meeting may be adjourned without further notice. Upon resumption of such adjourned meeting at which a quorum is present, any business may be transacted which might have been transacted at the original meeting.

- (p) No Member shall enter into any voting trust or other similar legally binding agreements or arrangements of any kind with any Person with respect to voting of any Units or other voting securities of the Company or any Company Subsidiaries.
- (q) If any Member purports to vote, or to grant any proxy or enter into any agreement, plan or other arrangement relating to the voting of, Units that would violate the provisions of this Agreement, then the Company shall not honor such vote, proxy, agreement, plan or other arrangement to the extent that such provisions would be violated, and any Units subject to that arrangement shall not be entitled to be voted to the extent of such violation.
- 4.8 No Interest in Company Property. No real or personal property of the Company shall be deemed to be owned by any Member individually, but shall be owned by, and title shall be vested solely in, the Company. Without limiting the foregoing, each Member hereby irrevocably waives during the term of the Company any right that such Member may have to maintain any action for partition with respect to the property of the Company.

## ARTICLE V. CAPITAL CONTRIBUTIONS; CAPITAL ACCOUNTS

- 5.1 <u>Capital Contributions and Unit Issuances of the Class A Members, Class C Members</u> and Class D Members.
- (a) The Capital Contributions and the date such Capital Contributions were made to the Company by each Class A Member, each Class C Member or each Class D Member, as applicable, in connection with such Class A Member's, Class C Member's or Class D Member's purchase of Units, are set forth opposite such Class A Member's, Class C Member's or Class D Member's respective name on the Members Schedule, as amended from time to time in accordance with this Agreement.
- (b) No Member shall be required to make any additional Capital Contributions to the Company. Any future Capital Contributions made by any Member may only be made with the consent of the Board by Supermajority Board Vote and in accordance with the terms of this Agreement.
- (c) No Member shall be required to lend any funds to the Company and no Member shall have any personal liability for the payment or repayment of any Capital Contribution by or to any other Member.

- 5.2 <u>Maintenance of Capital Accounts</u>. The Company shall establish and maintain for each Member a separate capital account (a "<u>Capital Account</u>") on its books and records in accordance with this <u>Section 5.2</u>. Each Capital Account shall be established and maintained in accordance with the following provisions:
  - (a) Each Member's Capital Account shall be increased by the amount of:
    - (i) such Member's Capital Contributions;
  - (ii) any Net Income or other item of income or gain allocated to such Member pursuant to Article VI; and
  - (iii) any liabilities of the Company that are assumed by such Member or secured by any property Distributed to such Member.
    - (b) Each Member's Capital Account shall be decreased by:
  - (i) the cash amount or Book Value of any property Distributed to such Member pursuant to <u>Article VII</u> and <u>Section 13.3(c)</u>;
  - (ii) the amount of any Net Loss or other item of loss or deduction allocated to such Member pursuant to <u>Article VI</u>; and
  - (iii) the amount of any liabilities of such Member assumed by the Company or which are secured by any property contributed by such Member to the Company.
- 5.3 <u>Succession Upon Transfer</u>. In the event that any Units are Transferred in accordance with the terms of this Agreement, the Transferee shall succeed to the Capital Account of the Transferor to the extent it relates to the Transferred Units and, subject to <u>Section 6.4</u>, shall receive allocations and Distributions pursuant to <u>Article VI</u>, <u>Article VII</u> and <u>Article XIII</u> in respect of such Units.
- 5.4 <u>Negative Capital Accounts</u>. In the event that any Member shall have a deficit balance in his, her or its Capital Account, such Member shall have no obligation, during the term of the Company or upon dissolution or liquidation thereof, to restore such negative balance or make any Capital Contributions to the Company by reason thereof, except as may be required by Applicable Law or in respect of any negative balance resulting from a withdrawal of capital or dissolution in contravention of this Agreement.
- 5.5 No Withdrawal. No Member shall be entitled to withdraw any part of his, her or its Capital Account or to receive any Distribution from the Company, except as provided in this Agreement. No Member shall receive any interest, salary or drawing with respect to its Capital Contributions or its Capital Account, except as otherwise provided in this Agreement. The Capital Accounts are maintained for the sole purpose of allocating items of income, gain, loss and

deduction among the Members and shall have no effect on the amount of any Distributions to any Members, in liquidation or otherwise.

- 5.6 <u>Treatment of Loans From Members</u>. Loans by any Member to the Company shall not be considered Capital Contributions and shall not affect the maintenance of such Member's Capital Account.
- 5.7 <u>Modifications</u>. The foregoing provisions and the other provisions of this Agreement relating to the maintenance of Capital Accounts are intended to comply with Section 1.704-1(b) of the Treasury Regulations and shall be interpreted and applied in a manner consistent with such Treasury Regulations. If the Board determines that it is prudent to modify the manner in which the Capital Accounts, or any increases or decreases to the Capital Accounts, are computed in order to comply with such Treasury Regulations, the Board may authorize such modifications.

# ARTICLE VI. ALLOCATIONS

- Allocation of Net Income and Net Loss. For each Fiscal Year (or portion thereof), except as otherwise provided in this Agreement, Net Income or Net Loss (and, to the extent necessary, individual items of income, gain, loss or deduction) of the Company shall be allocated among the Members in a manner such that, after giving effect to the special allocations set forth in Section 6.2 and Section 6.5, the Capital Account balance of each Member, immediately after making such allocations, is, as nearly as possible, equal to (a) the Distributions that would be made to such Member pursuant to Section 13.3(c) if the Company were dissolved, its affairs wound up and its assets sold for cash equal to their Book Value, all Company liabilities were satisfied (limited with respect to each Nonrecourse Liability to the Book Value of the assets securing such liability), and the net assets of the Company were Distributed, in accordance with Section 13.3(c), to the Members immediately after making such allocations, minus (b) such Member's share of Company Minimum Gain and Member Nonrecourse Debt Minimum Gain, computed immediately prior to the hypothetical sale of assets.
  - 6.2 Regulatory and Special Allocations. Notwithstanding the provisions of Section 6.1:
- (a) If there is a net decrease in Company Minimum Gain (determined according to Treasury Regulations Section 1.704-2(d)(1)) during any Fiscal Year, each Member shall be specially allocated income and gain for such Fiscal Year (and, if necessary, subsequent Fiscal Years) in an amount equal to such Member's share of the net decrease in Company Minimum Gain, determined in accordance with Treasury Regulations Section 1.704-2(g). The items to be so allocated shall be determined in accordance with Treasury Regulations Sections 1.704-2(f)(6) and 1.704-2(j)(2). This Section 6.2(a) is intended to comply with the "minimum gain chargeback" requirement in Treasury Regulation Section 1.704-2(f) and shall be interpreted consistently therewith.
- (b) Member Nonrecourse Deductions shall be allocated in the manner required by Treasury Regulations Section 1.704-2(i). Except as otherwise provided in Treasury Regulations

Section 1.704-2(i)(4), if there is a net decrease in Member Nonrecourse Debt Minimum Gain during any Fiscal Year, each Member that has a share of such Member Nonrecourse Debt Minimum Gain shall be specially allocated income and gain for such Fiscal Year (and, if necessary, subsequent Fiscal Years) in an amount equal to that Member's share of the net decrease in Member Nonrecourse Debt Minimum Gain. Items to be allocated pursuant to this paragraph shall be determined in accordance with Treasury Regulations Sections 1.704-2(i)(4) and 1.704-2(j)(2). This Section 6.2(b) is intended to comply with the "minimum gain chargeback" requirements in Treasury Regulations Section 1.704-2(i)(4) and shall be interpreted consistently therewith.

- (c) In the event any Member unexpectedly receives any adjustments, allocations or Distributions described in Treasury Regulations Section 1.704-1(b)(2)(ii)(d)(4), (5) or (6), income and gain shall be specially allocated to such Member in an amount and manner sufficient to eliminate the Adjusted Capital Account Deficit created by such adjustments, allocations or Distributions as quickly as possible. This Section 6.2(c) is intended to comply with the qualified income offset requirement in Treasury Regulations Section 1.704-1(b)(2)(ii)(d) and shall be interpreted consistently therewith.
- (d) The allocations set forth in paragraphs (a), (b) and (c) above (the "Regulatory Allocations") are intended to comply with certain requirements of the Treasury Regulations under Code Section 704. Notwithstanding any other provisions of this Article VI (other than the Regulatory Allocations), the Regulatory Allocations shall be taken into account in allocating Net Income and Net Losses among Members so that, to the extent possible, the net amount of such allocations of Net Income and Net Losses and other items and the Regulatory Allocations to each Member shall be equal to the net amount that would have been allocated to such Member if the Regulatory Allocations had not occurred.
- (e) The Company and the Members acknowledge that allocations like those described in Proposed Treasury Regulation Section 1.704-1(b)(4)(xii)(c) ("Forfeiture Allocations") result from the allocations of Net Income and Net Loss provided for in this Agreement. For the avoidance of doubt, the Company is entitled to make Forfeiture Allocations and, once required by applicable final or temporary guidance, allocations of Net Income and Net Loss shall be made in accordance with Proposed Treasury Regulation Section 1.704-1(b)(4)(xii)(c) or any successor provision or guidance.

#### 6.3 Tax Allocations.

(a) Subject to Section 6.3(b) through Section 6.3(e), all income, gains, losses and deductions of the Company shall be allocated, for federal, state and local income tax purposes, among the Members in accordance with the allocation of such income, gains, losses and deductions among the Members for computing their Capital Accounts, except that if any such allocation for tax purposes is not permitted by the Code or other Applicable Law, the Company's subsequent income, gains, losses and deductions shall be allocated among the Members for tax purposes, to the extent permitted by the Code and other Applicable Law, so as to reflect as nearly as possible the allocation set forth herein in computing their Capital Accounts.

- (b) Items of Company taxable income, gain, loss and deduction with respect to any property contributed to the capital of the Company shall be allocated among the Members in accordance with Code Section 704(c) and the traditional method of Treasury Regulations Section 1.704-3(b), so as to take account of any variation between the adjusted basis of such property to the Company for federal income tax purposes and its Book Value.
- (c) If the Book Value of any Company asset is adjusted pursuant to Treasury Regulation Section 1.704-1(b)(2)(iv)(f) as provided in clause (c) of the definition of Book Value, subsequent allocations of items of taxable income, gain, loss and deduction with respect to such asset shall take account of any variation between the adjusted basis of such asset for federal income tax purposes and its Book Value in the same manner as under Code Section 704(c).
- (d) Allocations of tax credit, tax credit recapture and any items related thereto shall be allocated to the Members according to their interests in such items as determined by the Board taking into account the principles of Treasury Regulations Section 1.704-1(b)(4)(ii).
- (e) Allocations pursuant to this <u>Section 6.3</u> are solely for purposes of federal, state and local taxes and shall not affect, or in any way be taken into account in computing, any Member's Capital Account or share of Net Income, Net Losses, Distributions or other items pursuant to any provisions of this Agreement.
- 6.4 <u>Allocations in Respect of Transferred Units</u>. In the event of a Transfer of Units during any Fiscal Year made in compliance with the provisions of <u>Article X</u>, Net Income, Net Losses and other items of income, gain, loss and deduction of the Company attributable to such Units for such Fiscal Year shall be determined using the interim closing of the books method.
- 6.5 <u>Curative Allocations</u>. In the event that the Board determines, after consultation with counsel experienced in income tax matters, that the allocation of any item of Company income, gain, loss or deduction is not specified in this <u>Article VI</u> (an "<u>Unallocated Item</u>"), or that the allocation of any item of Company income, gain, loss or deduction hereunder is clearly inconsistent with the Members' economic interests in the Company (determined by reference to the general principles of Treasury Regulations Section 1.704-1(b) and the factors set forth in Treasury Regulations Section 1.704-1(b)(3)(ii)) (a "<u>Misallocated Item</u>"), then the Board may allocate such Unallocated Items, or reallocate such Misallocated Items, to reflect such economic interests; <u>provided</u>, that no such allocation shall be made without the prior consent of each Member that would be adversely and disproportionately affected thereby; and <u>provided</u>, <u>further</u>, that no such allocation shall have any material effect on the amounts distributable to any Member, including the amounts to be distributed upon the complete liquidation of the Company.

## ARTICLE VII. DISTRIBUTIONS

7.1 <u>Tax Advances</u>. At least fifteen (15) calendar days prior to the end of each fiscal quarter, the Company shall make advances ("<u>Tax Advances</u>") to the Members so that each Member shall have received Tax Advances equal to such Member's Tax Amount for such fiscal quarter (or,

in the event that Net Available Cash is less than the total required Tax Advances for such fiscal quarter, the Company shall distribute the Net Available Cash pro rata among the Members in proportion to such required Tax Advances for such fiscal quarter). If after the end of any Fiscal Year it is determined that a Member's Tax Amount for the Fiscal Year exceeds the sum of the cumulative Tax Advances made to the Member hereunder and the Distributions made to such Member under Section 7.2, if any, for such Fiscal Year (any such excess, a "Shortfall Amount"), then the Company shall, on or before March 15 of the following Fiscal Year, make an additional Tax Advance to the Members of their respective Shortfall Amounts. If the aggregate Tax Advances to any Member pursuant to this Section 7.1 for a Fiscal Year exceed the Member's Tax Amount for such Fiscal Year, such excess shall be deducted when calculating the Tax Advances to be made to such Member for each subsequent Fiscal Year until the excess has been fully accounted for. All Tax Advances to a Member shall be treated as advances against any subsequent Distributions to be made to such Member under Sections 7.2 or 13.3(c) and shall be repaid by reducing the amount of the next succeeding Distribution to such Member under Sections 7.2 or 13.3(c). If any Distributions to a Member provided for in Sections 7.1, 7.2 or 13.3(c) are reduced pursuant to the preceding sentences of this Section 7.1, for all other purposes of this Agreement, such Member shall be treated as having received all Distributions without taking into account such reduction, and the Member shall be treated as having repaid to the Company the Tax Advance.

## 7.2 General.

- (a) Subject to Sections 4.7(a)(i), 4.7(c)(i), 7.1, 7.3, 8.6(d) and 13.3 the Board shall have the discretion regarding the amounts and timing of Distributions to Members, including to decide to forego payment of Distributions in order to provide for the retention and establishment of reserves of, or payment to third parties of, such funds as it deems necessary with respect to the reasonable business needs of the Company (which needs may include the payment or the making of provision for the payment when due of the Company's obligations, including, but not limited to, present and anticipated debts and obligations, capital needs and expenses, the payment of any management or administrative fees and expenses, and reasonable reserves for contingencies).
- (b) Notwithstanding any provision to the contrary contained in this Agreement, the Company shall not make any Distribution to Members if such Distribution would violate § 18-607 of the Delaware Act or other Applicable Law.
- 7.3 <u>Priority of Distributions</u>. Subject to the priority of Distributions pursuant to <u>Section 13.3(c)</u>, if applicable, all Distributions determined to be made by the Board pursuant to this <u>Section 7.3</u> shall be made to the Members as follows:

#### (a) First:

(i) with respect to each outstanding Class C Unit an amount equal to the unreturned Capital Contributions with respect to such Class C Unit; <u>provided</u> that the amount of the unreturned Capital Contributions with respect to each outstanding Class C Unit shall be reduced by the amount of any previous Distributions made with respect to such

Class C Unit; <u>provided</u>, <u>further</u> that such amounts will be distributed to the Class C Members *pro rata* in accordance with their unreturned Capital Contributions; and

(ii) with respect to each outstanding Class D Unit an amount equal to the unreturned Capital Contributions with respect to such Class D Unit; <u>provided</u> that the amount of the unreturned Capital Contributions with respect to each outstanding Class D Unit shall be reduced by the amount of any previous Distributions made with respect to such Class D Unit; <u>provided</u>, <u>further</u> that such amounts will be distributed to the Class D Members *pro rata* in accordance with their unreturned Capital Contributions;

<u>provided</u>, that if the amount available for distribution at any applicable time is insufficient to distribute to the Class C Members and Class D Members the full amounts to which the Class C Members and the Class D Members would be entitled pursuant to clauses (i) and (ii) immediately above, as applicable, then the applicable amount available for distribution shall be shared *pro rata* among such Class C Members and Class D Members based on the amounts to which such Class C Members and Class D Members would have been entitled pursuant to clauses (i) and (ii) immediately above, as applicable.

- (b) <u>Second</u>, with respect to each outstanding Class A Unit an amount equal to the unreturned Capital Contributions with respect to such Class A Unit; <u>provided</u> that the amount of the unreturned Capital Contributions with respect to each outstanding Class A Unit shall be reduced by the amount of any previous Distributions made with respect to such Class A Unit; <u>provided further</u> that such amounts will be distributed to the Class A Members *pro rata* in accordance with their unreturned Capital Contributions;
- <u>Thereafter</u>, with respect to each outstanding Class A Unit, vested Class B Unit, outstanding Class C Unit (on an as-converted to Common Unit basis), outstanding Class D Unit (on an as-converted to Common Unit basis) and each outstanding Common Unit, an amount equal to the remaining Distribution Amount divided by the aggregate number of all outstanding Class A Units, vested Class B Units, outstanding Class C Units (on an as-converted to Common Unit basis), outstanding Class D Units (on an as-converted to Common Unit basis), and Common Units outstanding at the time of the Distribution which are entitled to participate in the Distribution (taking into account the computational rules of Section 13.3(d)); provided that no Distributions shall be made pursuant to this Section 7.3(c) with respect to any Class C Unit (on an as-converted to Common Unit basis) or any Class D Unit (on an as-converted to Common Unit basis), and any Converted Common Unit until the amount of aggregate per-Unit Distributions pursuant to this Section 7.3(c) and any additional Distributions made as a result of operation of Section 7.3(d) with respect to each Class A Unit and each Common Unit which is not a Converted Common Unit is equal to the amount previously distributed (including pursuant to Section 7.3(a)) with respect to each Class C Unit (on an as-converted to Common Unit basis) and each Class D Unit (on an asconverted to Common Unit basis) or Converted Common Unit, and any amount not distributed as a consequence of this proviso shall be re-allocated and distributed pursuant to this Section 7.3(c), with this proviso applied on an iterative basis.

(d) Notwithstanding Section 7.3(c), to the extent that any portion of the Distribution Amount represents an amount earned before the grant of any vested Class B Unit (including amounts representing directly or indirectly the proceeds of the sale of any of the Company's assets at the time of such grant to the extent allocable to the value of such asset at the time of such grant), that portion of the Distribution Amount shall be distributed to the holder of such vested Class B Unit only to the extent such portion exceeds the Participation Threshold with respect to such vested Class B Unit. Distributions limited by the foregoing sentence shall reduce the applicable vested Class B Unit's Participation Threshold and shall, in lieu of distribution to the holder of such vested Class B Unit, be distributed *pro rata* to the Class A Units, the vested Class B Units not subject to such a limitation and the Common Units.

### 7.4 Tax Withholding; Withholding Advances; Tax Election.

- (a) <u>Tax Withholding</u>. If requested by the Board, each Member shall, if able to do so, deliver to the Board such information, documentation or certification as may be reasonably requested by the Board in connection with any tax return or information filing requirements or to obtain a reduced rate of, or exemption from, any applicable tax, whether pursuant to the laws of such jurisdiction or an applicable tax treaty.
- (b) <u>Withholding Advances</u>. The Company is hereby authorized at all times to make payments ("<u>Withholding Advances</u>") with respect to each Member in amounts required to discharge any obligation of the Company (as determined by the Board based on the advice of legal or tax counsel to the Company) to withhold or make payments to any federal, state, local or foreign taxing authority (a "<u>Taxing Authority</u>") with respect to any Distribution or allocation by the Company of income or gain to such Member (including payments made pursuant to Code Section 6225 as amended by the BBA and apportioned to a Member as determined by the Board in its reasonable discretion by Supermajority Board Vote based on the advice of legal or tax counsel to the Company) and to withhold the same from Distributions to such Member. Any funds withheld from a Distribution by reason of this <u>Section 7.4(b)</u> shall nonetheless be deemed Distributed to the Member in question for all purposes under this Agreement and, at the option of the Board, shall be charged against the Member's Capital Account. The Board shall promptly notify a Member of any proposed Withholding Advances.
- (c) Repayment of Withholding Advances. Any Withholding Advance made by the Company to a Taxing Authority on behalf of a Member and not simultaneously withheld from a Distribution to that Member shall, with interest thereon accruing from the later of (1) the date of payment or (2) the date that the Company notified such Member of such Withholding Advance at a rate equal to the prime rate published in *The Wall Street Journal* on the date of payment plus two percent (2.0%) per annum:
  - (i) be promptly repaid to the Company by the Member on whose behalf the Withholding Advance was made (which repayment by the Member shall not constitute a Capital Contribution, but shall credit the Member's Capital Account if the Board shall have initially charged the amount of the Withholding Advance to the Capital Account); or

(ii) with the consent of the Board, be repaid by reducing the amount of the next succeeding Distribution or Distributions to be made to such Member (which reduction amount shall be deemed to have been Distributed to the Member, but which shall not further reduce the Member's Capital Account if the Board shall have initially charged the amount of the Withholding Advance to the Capital Account).

Interest shall cease to accrue from the time the Member on whose behalf the Withholding Advance was made repays such Withholding Advance (and all accrued interest) by either method of repayment described above.

- (d) <u>Indemnification</u>. Each Member hereby agrees to indemnify and hold harmless the Company and the other Members from and against (i) any liability with respect to taxes, interest or penalties which may be asserted by reason of the Company's failure to deduct and withhold tax on amounts distributable or allocable to such Member (excluding any liability attributable to fraud, willful misconduct, gross negligence or material breach of this Agreement by the Company and/or another Member, as and where applicable) and (ii) such Member's (or former Member's) share of any imputed underpayment within the meaning of Code Section 6225, in each case as determined reasonably and in good faith by the Board based on the advice of legal or tax counsel to the Company. The provisions of this <u>Section 7.4(d)</u> and the obligations of a Member pursuant to <u>Section 7.4(c)</u> shall survive the termination, dissolution, liquidation and winding up of the Company and the withdrawal of such Member from the Company or Transfer of its Units. The Company may pursue and enforce all rights and remedies it may have against each Member under this <u>Section 7.4</u>, including bringing a lawsuit to collect repayment with interest of any Withholding Advances.
- (e) <u>Overwithholding</u>. Neither the Company nor the Board shall be liable for any excess taxes withheld in respect of any Distribution or allocation of income or gain to a Member. In the event of any such overwithholding, a Member's sole recourse shall be to apply for a refund from the appropriate Taxing Authority.
- (f) <u>Tax Election</u>. The Company shall make an election under Section 754 of the Code (to the extent permissible under the Code) upon its filing of its 2020 U.S. federal income tax return.
- 7.5 <u>Distributions in Kind</u>. Except as expressly provided to the contrary herein or required by Applicable Law: (a) the Company is not authorized to make Distributions to the Members in the form of securities or other property held by the Company; and (b) no Member may be required to accept consideration with respect to a merger, business combination or other transaction to which the Company or a Subsidiary of the Company is a party in the form of securities or other property if such member notifies the Company that receipt of such consideration by such Member would violate the BHCA or other Applicable Law or cause such Member to control or be presumed to control the issuer of such asset under the BHCA. In either such case, the affected Member may elect, in the alternative, to receive the Fair Market Value of such consideration in cash.

# ARTICLE VIII. MANAGEMENT

- 8.1 <u>The Board</u>. The Company shall have a board of directors of the Company (the "<u>Board</u>") that shall be comprised of natural Persons (each such Person, a "<u>Director</u>") who shall be nominated and deemed elected to the Board in accordance with the provisions of <u>Section 8.3</u>.
- 8.2 <u>Authority of Board</u>. The business and affairs of the Company shall be managed, operated and controlled by or under the direction of the Board, and the Board shall have, and is hereby granted, the full and complete power, authority and discretion for, on behalf of and in the name of the Company, to take such actions as it may in its sole discretion deem necessary or advisable to carry out any and all of the objectives and purposes of the Company, subject only to the terms of this Agreement.

## 8.3 <u>Board Composition; Vacancies.</u>

- (a) Subject to <u>Section 8.11</u> and <u>Section 8.17</u>, the size of the Board shall be set at fifteen (15) Directors unless otherwise determined by the Board by Supermajority Board Vote, but in no event shall the Board be less than five (5) Directors. If the number of Directors is reduced to five (5), the Members will work in good faith to identify and elect an additional director(s).
- (b) The Company and the Members shall take such actions as may be required to ensure that the number of Directors constituting the Board is at all times such number as determined by the Board by Supermajority Board Vote. Each Class A Member which, at the time of its initial investment in the Company, purchases at least five million (5,000,000) Class A Units and each of Wells Fargo, Citi, and Optiver shall have the right to nominate one (1) individual as a Director (the Members which have the rights to nominate Directors hereunder collectively referred to herein as a "Nominating Members" and individually as a "Nominating Member"). All of the individuals so nominated shall be deemed elected to the Board upon such nomination. The right of a Nominating Member to nominate a Director may be eliminated or waived, as applicable, as set forth in Section 8.10, Section 8.11 and Section 8.17. For the avoidance of doubt, a Member shall not be a Nominating Member for so long as such Member's right to nominate a Director is eliminated or waived pursuant to the immediately preceding sentence.
- (c) The individual serving as the chief executive officer of the Company (the "<u>CEO</u>") shall be deemed elected to the Board as a Director at the time of his or her appointment as the CEO by the Board.
- (d) In the event that a vacancy is created on the Board at any time due to the death, disability, retirement, resignation or removal of a Director, then the Nominating Member that nominated such Director shall have the right to nominate an individual to fill such vacancy and the applicable individual shall be deemed elected to the Board upon such nomination.
- (e) The Board shall maintain a schedule of all Directors, Alternate Directors and Board Observers (the "<u>Directors and Observers Schedule</u>"), and shall update the Directors and

Observers Schedule upon the removal or replacement of any Director, Alternate Director or Board Observer in accordance with this <u>Section 8.3</u>, <u>Section 8.4</u>, <u>Section 8.12</u>, <u>Section 8.13</u>, or <u>Section 8.17</u>, as applicable. A copy of the Directors and Observers Schedule is attached hereto as <u>Exhibit B</u>. Each Member that nominates a Director or Alternate Director, or designates a Board Observer, shall, concurrently with such nomination, provide the Company with the contact information for such person or persons.

## 8.4 Removal; Resignation.

- (a) A Director or Board Observer may be removed from his or her position as such, or replaced at any time, with or without cause, upon, and only upon, the written request of the Nominating Member that nominated such Director or the Member that appointed such Board Observer, as applicable; <u>provided</u>, that notwithstanding the foregoing, a Director or Board Observer may be subject to a statutory disqualification (within the meaning of Section 3(a)(39) of the Exchange Act). A Director or Board Observer who becomes subject to a statutory disqualification shall automatically and immediately be removed from the Board.
- (b) A Nominating Member may irrevocably waive its right in Section 8.4(a) to remove or replace a Director nominated by such Nominating Member. If a Nominating Member makes such an election and a Director nominated by such Nominating Member dies, resigns in accordance with Section 8.4(c), is removed as a result of a statutory disqualification in accordance with Section 8.4(a) or is a Director for whom the Nominating Member has provided written notice to the Company of the Director's termination or resignation as an employee of his or her Nominating Member or any of its Affiliates in accordance with Section 8.4(f), the Nominating Member that nominated such Director may nominate a new Director in accordance with Section 8.3.
- (c) A Director or Board Observer may resign at any time from his or her position as such by delivering his or her written resignation to the Board. Any such resignation shall be effective upon receipt thereof unless it is specified to be effective at some other time or upon the occurrence of some other event. The Board's acceptance of a resignation shall not be necessary to make it effective.
- (d) If a Director dies, resigns or is removed from the Board, such Director's Alternate Director shall serve as a Director until the Nominating Member that nominated such Director nominates a new Director.
- (e) If the CEO resigns or is removed from his or her position as such, the applicable individual shall automatically and immediately be removed from his or her position as a Director.
- (f) If a Director is an employee of his or her Nominating Member or any of its Affiliates, and such Director's Nominating Member has provided written notice to the Company of the Director's termination or resignation as an employee of the Nominating Member or any of its

Affiliates, the applicable individual shall automatically and immediately be removed from his or her position as a Director upon the Company's receipt of such notice.

## 8.5 <u>Meetings</u>.

- (a) Generally. The Chairman of the Board shall be responsible for setting the time and place at which regularly scheduled meetings of the Board shall take place; provided that meetings of the Board shall take place at least quarterly. Meetings of the Board may be held either in person or by means of telephone or video conference or other communications device that permits all participants to hear each other, at the offices of the Company or such other place (either within or outside the State of Delaware) as may be determined from time to time by the Board. Written notice of each meeting of the Board shall be given to each Director, each Alternate Director and each Board Observer at least five (5) Business Days prior to each such meeting; provided that if the sole purpose of a meeting of the Board is to address matters pertaining to the employment of the CEO, the CEO, in his or her capacity as a Director, shall not be provided notice of such meeting.
- (b) <u>Special Meetings</u>. Special meetings of the Board shall be held on the call of any Director upon at least ten (10) days' written notice (unless a shorter notice period is required in an event of an emergency as determined in good faith by the Chairman of the Board) to the Directors, or upon such shorter notice as may be approved by all the Directors. Any Director may waive such notice as to himself or herself.
- (c) Attendance and Waiver of Notice. Attendance of a Director or, when serving as a Director, an Alternate Director, at any meeting shall constitute a waiver of notice of such meeting, except where a Director or an Alternate Director, as applicable, attends a meeting for the express purpose of objecting to the transaction of any business on the ground that the meeting is not lawfully called or convened. Neither the business to be transacted at, nor the purpose of, any regular or special meeting of the Board need be specified in the notice or waiver of notice of such meeting.

#### 8.6 Quorum; Manner of Acting.

### (a) Quorum.

(i) A quorum for the transaction of business of the Board shall constitute a number of Directors which both (A) represents the majority of the votes of the Directors serving on the Board, and (B) includes (x) at least one (1) Market Maker Director (or his or her Alternate Director), (y) at least one (1) Retail Broker Director (or his or her Alternate Director) or at least one (1) Buy Side Director (or his or her Alternate Director), and (z) at least one (1) Bank Director (or his or her Alternate Director), provided, however, in each case, that if no such Director is then serving on the Board, such Director shall not be required for purposes of establishing a quorum; and provided, further, that if the sole purpose of a meeting of the Board is to address matters pertaining to the employment of the CEO, the CEO, in his or her capacity as a Director, shall not be counted for the purpose of establishing whether a quorum is present for the purposes of such meeting. At all times when the Board is conducting business at a meeting of the Board, a quorum of the Board

must be present at such meeting. If a quorum shall not be present at any meeting of the Board, then the Directors present at the meeting may adjourn the meeting from time to time, without notice other than announcement at the meeting, until a quorum shall be present. If a quorum shall not be present at any meeting of the Board, the Directors (and, where applicable, Alternate Directors) present at such meeting may adjourn the meeting; provided, that such adjourned meeting shall be rescheduled with at least three (3) Business Days' prior written notice of such rescheduled meeting.

- (ii) If a Director and his or her Alternate Director (where applicable) fail to attend two (2) consecutively scheduled meetings (whether regular or special meetings) of the Board then until such Director or his or her Alternate Director attends a meeting of the Board:
  - (A) at all subsequent meetings of the Board a quorum shall not be found to be lacking for the sole reason that such Director and Alternate Director are not in attendance. In addition, if (I) such Director is a Market Maker Director, the presence of at least one (1) Market Maker Director shall not be required for a quorum to be present if such Director is then the sole Market Maker Director serving on the Board, (II) such Director is a Retail Broker Director or a Buy Side Director the presence of at least one (1) Retail Broker Director or Buy Side Director shall not be required for a quorum to be present if such Director is then the sole Director serving on the Board that was nominated by either a Retail Broker Member or a Buy Side Member, and (III) such Director is a Bank Director the presence of at least one (1) Bank Director shall not be required for a quorum to be present if such Director is then the sole Bank Director serving on the Board.
  - (B) during any such meeting at which such Director and Alternate Director are not in attendance any approval thresholds shall be calculated as if the Class A Member which nominated such Director was not then entitled to nominate a Director.
- (b) <u>Participation</u>. Any Director may participate in a meeting of the Board by means of telephone or video conference or other communications device that permits all participants in the meeting to hear each other, and participation in a meeting by such means shall constitute presence in person at such meeting. A Director may vote or be present at a meeting either in person or by proxy, and such proxy may be granted in writing, by means of Electronic Transmission or as otherwise permitted by Applicable Law.
- (c) <u>Director Votes</u>. Each Director shall have one (1) vote on all matters submitted to the Board.
- (d) <u>Supermajority Board Matters</u>. Notwithstanding anything to the contrary contained in this Agreement, the Company shall not, and shall not permit any Company Subsidiary to, engage in or cause any of the transactions or take any of the Supermajority Board Matters, and the Board shall not permit or cause the Company or any Company Subsidiary to engage in, take or

cause any such transaction or action except, in each case, with the prior approval of such transaction or action by Supermajority Board Vote; provided, that notwithstanding the foregoing, the Company shall not, and shall not permit any Company Subsidiary to, engage in or cause any of the transactions or take any of the Supermajority Board Matters, and the Board shall not permit or cause the Company or any Company Subsidiary to engage in, take or cause any such transaction or action, to the extent such transaction or action adversely and disproportionately affects any Member or group of Members relative to the other Members without the prior written consent of such disproportionately affected Member or Members (as applicable); provided, further that, notwithstanding the foregoing, with respect to any Regulated Securities Exchange Subsidiary, such approval by Supermajority Board Vote shall not be required if such Regulated Securities Exchange Subsidiary engaging in or causing any applicable transaction or taking any applicable action (which transaction or action is a Supermajority Board Matter and would otherwise require approval by Supermajority Board Vote) is required by such Regulated Securities Exchange Subsidiary to fulfill its regulatory functions or responsibilities or to oversee the structure of the market that such Regulated Securities Exchange Subsidiary regulates or not to violate or potentially violate any Applicable Law, in each case as determined by the Exchange Board of such Regulated Securities Exchange Subsidiary upon receipt of an opinion of counsel (which may be in-house counsel and need not be in writing) to such effect.

- (e) <u>Other Board Matters</u>. With respect to any matter, action or transaction before the Board, other than a matter, action, or transaction subject to Supermajority Board Vote as set forth herein, the act of a number of Directors representing the majority of the votes of the Directors shall be the act of the Board.
- 8.7 Action By Written Consent. Notwithstanding anything herein to the contrary, any action of the Board may be taken without a meeting if a written consent (including via email communication) of all of the Directors then constituting the Board approves such action. With respect to any Director, such written consent may be provided by such Director's Alternate Director. Such consent shall have the same force and effect as a vote at a meeting where a quorum was present and may be stated as such in any document or instrument filed with the Secretary of State of Delaware. For purposes of the foregoing, an action shall be deemed to have been taken in writing via email communication if (i) an email communication is sent by an Officer or the Chairman of the Board to all Directors entitled to vote on the matter at issue clearly specifying the action to be taken, (ii) the number of Directors required to approve the matter at issue respond to the Officer's or the Chairman of the Board's email with an unambiguous approval of such matter, and (iii) the Officer's or Chairman of the Board's email and all such responses are filed with the minutes of the meetings of Directors.

## 8.8 <u>Compensation; No Employment.</u>

(a) Each Director and Alternate Director shall be reimbursed for his or her reasonable out-of-pocket expenses incurred in the performance of his or her duties as a Director or Alternate Director, as applicable, pursuant to such policies as from time to time established by the Board. Nothing contained in this <u>Section 8.8</u> shall be construed to preclude any Director or

Alternate Director from serving the Company in any other capacity and receiving reasonable compensation for such services.

(b) This Agreement does not, and is not intended to, confer upon any Director or Alternate Director any rights with respect to continued employment by the Company, and nothing herein should be construed to have created any employment agreement with any Director or Alternate Director.

#### 8.9 Committees.

- (a) <u>Board Advisory Committees</u>. The Board may, by Supermajority Board Vote, designate from among the Directors and Alternate Directors one or more committees, each of which shall be comprised of one or more Directors and Alternate Directors. Committees shall have the authority to make recommendations to the Board. Any committees which are so designated shall have an advisory role only and shall not have the authority to act for or on behalf of, or to bind, the Company or any Company Subsidiaries. No action taken by a committee shall be binding on the Company unless approved by the Board. The Board may dissolve any committee or remove any member of a committee at any time.
- (b) <u>Market Structure Committees</u>. The Board shall maintain market structure committees (each, a "<u>Market Structure Committee</u>") as set forth in this <u>Section 8.9(b)</u>. A Market Structure Committee shall have the authority to make recommendations to the Board. A Market Structure Committee shall have an advisory role only and shall not have the authority to act for or on behalf of, or to bind, the Company or any Company Subsidiaries. No action taken by a Market Structure Committee shall be binding on the Company unless approved by the Board.
  - (i) The Board shall maintain an equities Market Structure Committee (the "Equities Market Structure Committee"). The Equities Market Structure Committee may be comprised of Directors, Alternate Directors, Board Observers and/or other representatives of Nominating Members. So long as BlackRock remains a Nominating Member, (A) BlackRock shall have the right, but not the obligation, to designate one of its representatives to serve on the Equities Market Structure Committee at all times, and (B) if BlackRock so requests, a representative of BlackRock shall be the chairperson of the Equities Market Structure Committee.
  - (ii) The Board shall maintain an options Market Structure Committee (the "Options Market Structure Committee"). The Options Market Structure Committee may be comprised of Directors, Alternate Directors, Board Observers, or representatives of Members. So long as Optiver remains a Nominating Member, (A) Optiver shall have the right, but not the obligation, to designate one of its representatives to serve on the Options Market Structure Committee at all times, and (B) if Optiver so requests, a representative of Optiver shall be the chairperson of the Options Market Structure Committee.
- 8.10 <u>Loss or Transfer of Right to Nominate a Director; Grant of Right to Nominate a Director.</u>

- (a) Subject to Section 8.10(b), if a Nominating Member ceases to own an aggregate number of Class A Units, Class C Units and/or Class D Units equal to at least 2,500,000 (subject to adjustment in the event of any Unit split, Unit combination, reorganization, reclassification, recapitalization or the like), then:
  - (i) such Nominating Member shall no longer have the right to nominate a Director;
  - (ii) the Director (and Alternate Director) nominated by such Nominating Member shall automatically and immediately be removed from the Board; and
  - (iii) the number of Directors shall be reduced by one (1) (subject, where applicable, to Section 8.10(b)).
- (b) If a Nominating Member Transfers an aggregate number of Class A Units, Class C Units and/or Class D Units equal to at least 2,500,000 (subject to adjustment in the event of any Unit split, Unit combination, reorganization, reclassification, recapitalization or the like) to a third party, the Board may determine to grant such third party (a "Nominating Transferee") the right to nominate one (1) Director to the Board; provided, however, that such Nominating Transferee's nominee for Director shall not be appointed to serve on the Board unless such nominee's appointment is approved by the Board.
- (c) Without limiting <u>Section 8.10(b)</u>, the Board may, by Supermajority Board Vote, grant a Member that is not a Nominating Member the right to nominate a Director. Upon the grant of such right, such Member shall be deemed a Nominating Member for purposes of this Agreement.

## 8.11 Waiver of Right to Nominate a Director.

- (a) A Nominating Member may, at its election, give a revocable or an irrevocable notice of waiver (the "Nomination Waiver") of its right to nominate a Director to the Board pursuant to this <u>Article VIII</u> by delivering written notice to the Company in the form attached as Exhibit H hereto.
- (b) Upon receiving any Nomination Waiver, the Company shall promptly (but no later than five (5) Business Days following receipt of such notice) notify the other Members in writing upon any such election having been made by a Nominating Member.
- (c) In connection with such Nomination Waiver, such Nominating Member shall cause a Director nominated by such Nominating Member to resign from the Board.
- (d) During the time in which a Member's Nomination Waiver is in effect, no Person shall be entitled to fill the vacancy on the Board with respect to the nomination right previously held by such Member.

(e) A Member may, if it has given a revocable Nomination Waiver (but not an irrevocable Nomination Waiver), in its sole discretion at any time, reverse the waiver of its right to nominate a Director to the Board upon written notice to the Company.

#### 8.12 Alternate Directors.

- Subject to the remainder of this Section 8.12, each Nominating Member (a) entitled to nominate one or more Directors shall be entitled to nominate an alternate for each such Director that such Nominating Member is entitled to nominate (each such alternate, an "Alternate Director"), who shall be deemed elected as such and shall have the right to serve, act and vote as a Director in the absence of the principal Director from time to time. Such Alternate Director shall be permitted to attend all meetings of the Board even if the principal Director is present at such meetings. If a principal Director is present at a meeting, the Alternate Director for such Director shall attend as an observer, shall not be counted towards the quorum at such meeting, and shall not have the right to act or vote as a Director at such meeting. If a principal Director is not present at a meeting, the Alternate Director for such Director shall attend as a full Director and be entitled to vote and act as a Director at such meeting and shall be counted toward the quorum at such meeting. If a principal Director attends a meeting, is counted toward the quorum, and, thereafter (i) recuses himself or herself from a vote on any particular matter, the Alternate Director for such Director, if present at such meeting, shall be allowed to vote as a Director on such matter, or (ii) leaves such meeting, the Alternate Director for such Director, if present at such meeting, shall sit as a Director for the remainder of such meeting, subject, in each case, to the provisions of Section 8.16 which would prohibit such Alternate Director from voting on a particular matter.
- (b) If a Member ceases to have the right to nominate a Director as provided in this Agreement, any Alternate Director nominated by such Member shall immediately cease to be an Alternate Director.
- (c) The Company shall provide to each Alternate Director copies of all notices, board materials, reports, minutes and consents at the time and in the manner as they are provided to the Directors; <u>provided</u> that each Alternate Director shall be subject to the confidentiality obligations set forth in <u>Section 11.1</u>.
- (d) The provisions of <u>Section 8.4(c)</u>, <u>Section 8.4(d)</u> and <u>Section 8.8</u> shall apply to Alternate Directors, *mutatis mutandis*, as if the references to Directors therein were references to Alternate Directors.
- (e) The initial Alternate Directors for the initial principal Directors shall be as set forth on Exhibit B.

#### 8.13 Board Observers.

(a) If a Member no longer has the right to nominate a Director as a result of operation of Section 8.10, but continues to hold an aggregate number of Class A Units, Class C Units and/or Class D Units equal to at least 1,250,000 (subject to adjustment in the event of any

Unit split, Unit combination, reorganization, reclassification, recapitalization or the like), such Member shall have the right, but not the obligation, to appoint one (1) observer to the Board (a "Board Observer"). If such Member has appointed such Board Observer and, thereafter, ceases to hold an aggregate number of Class A Units, Class C Units and/or Class D Units equal to at least 1,250,000 (subject to adjustment in the event of any Unit split, Unit combination, reorganization, reclassification, recapitalization or the like), such Member shall no longer have the right to appoint a Board Observer and the Board Observer appointed by such Member shall automatically and immediately be removed from his or her position as such.

- (b) If a Member Transfers an aggregate number of Class A Units, Class C Units and/or Class D Units equal to at least 1,250,000 (calculated in aggregate and subject to adjustment in the event of any Unit split, Unit combination, reorganization, reclassification, recapitalization or the like) to a third party, the Board may determine to grant such transferee the right to appoint one (1) Board Observer; provided, however, that such transferee's designee for Board Observer shall not be appointed as a Board Observer unless such designee's appointment is approved by the Board.
- (c) Without limiting <u>Section 8.13(a)</u> and <u>Section 8.13(b)</u>, the Board may, by Supermajority Board Vote, permit a Member to appoint a Board Observer.
- (d) The Board Observers shall have the right to participate in any discussions taking place at a meeting of the Board, but shall not have any rights to participate in the vote of the Board.
- (e) The Company shall provide to each Board Observer copies of all notices, board materials, reports, minutes and consents at the time and in the manner as they are provided to the Directors; provided that each Board Observer shall be subject to the confidentiality obligations set forth in Section 11.1. Each Board Observer shall be entitled to be present in person as an observer at any meeting of the Board or, if a meeting is held by video or telephone conference, to participate therein for the purpose of listening thereto; provided that no Board Observer other than the CEO shall be under an obligation to the Company to attend or participate in any such meeting. Notwithstanding the foregoing, no Board Observer shall be entitled to receive any information or materials or be present at a meeting of the Board (or, if applicable, a portion of the meeting of the Board) where such information or materials are discussed if, in the reasonable judgment of the Board or its designee (i) the Board Observers having access to such information or materials would result in a waiver of any applicable legal privilege, or that (ii) disclosure of such information or materials to the Board Observers is not permitted under Applicable Law. Notwithstanding the foregoing, the Board may, at its discretion, exclude the CEO from discussions regarding executive compensation.
- (f) The provisions of <u>Section 8.4(c)</u>, <u>Section 8.4(d)</u> and <u>Section 8.8</u> shall apply to Board Observers, *mutatis mutandis*, as if the references to Directors therein were references to Board Observers.
  - 8.14 Officers.

- (a) The Board may appoint individuals as officers of the Company (the "Officers") as it deems necessary or desirable to carry on the business of the Company and the Board may delegate to such Officers such power and authority as the Board deems advisable. No Officer need be a Member or Director. Any individual may hold two or more offices of the Company. Each Officer shall hold office until his successor is designated by the Board or until his earlier death, resignation or removal. Any Officer may resign at any time upon written notice to the Board. Any Officer may be removed by the Board with or without cause at any time. A vacancy in any office occurring because of death, resignation, removal or otherwise, may, but need not, be filled by the Board.
- (b) The Board shall select the Chairman of the Board from time to time. The Chairman of the Board shall preside at all meetings of the Members and at all meetings of the Board, and shall perform such other duties as the Board may from time to time determine.
- 8.15 <u>No Personal Liability</u>. Except as otherwise provided in the Delaware Act, by Applicable Law or expressly in this Agreement, no Director or Alternate Director shall be obligated personally for any debt, obligation or liability of the Company or of any Company Subsidiaries, whether arising in contract, tort or otherwise, solely by reason of being a Director or Alternate Director.
- Company Related Party Transactions. Notwithstanding anything in this Agreement 8.16 to the contrary, (a) a Director or Alternate Director shall be entitled (subject to the following provisos) to participate in deliberations with respect to any Company Related Party Transaction in which such Director or Alternate Director, the Member that nominated such Director or Alternate Director, or any other Company Related Party with respect to such Member, has an interest; provided that such Director or Alternate Director has disclosed such interest to the Board; and provided, further that the Board (excluding such Director or Alternate Director consistent with clause (b) of this Section 8.16), may elect to exclude such Director or Alternate Director from all or any portion of such deliberations, but (b) no such Director or Alternate Director shall be entitled to vote on or approve any such Company Related Party Transaction, and all determinations of the requirements for quorum and voting pursuant to Section 8.6 shall be recalculated by disregarding (where applicable for the purposes of both the applicable numerator and the applicable denominator), for purposes of any matter, any such Director or Alternate Director who is not entitled to vote in respect of such matter pursuant to this Section 8.16. Without limiting the generality of clause (b) of this Section 8.16, if the Board has previously approved a contract, arrangement or transaction as a Company Related Party Transaction and, thereafter, the Company desires to exercise a right or take action with respect to such Company Related Party Transaction, a Director or an Alternate Director representing a Company Related Party that is party to such Company Related Party transaction shall not be entitled to vote on or approve the exercise of any such right or the taking of any such action. The Company may not enter into any Competitively Sensitive Agreement, unless the CEO notifies the Board in writing, which notice shall be delivered to the Board together with the Opinion, prior to entering into a Competitively Sensitive Agreement (and any amendment thereof) identifying the parties to such agreement and the general subject matter of such agreement.

# 8.17 Combination of Nominating Members.

Notwithstanding any other provision of this Agreement:

- (a) If a Nominating Member merges, consolidates or otherwise combines with, obtains Control over, or becomes Affiliated with, another Nominating Member (a "<u>Combination</u>"), the surviving Affiliated group shall (i) if both such Nominating Members had nominated a Director that is serving on the Board at the time of the Combination, remove or cause the removal of one of such Directors effective upon the consummation of such Combination, and (ii) thereafter have the right to nominate only one Director and the number of Directors shall be reduced accordingly.
- (b) If the surviving Affiliated group in a Combination is required to remove a Director pursuant to Section 8.17(a), but does not do so, effective upon the consummation of such Combination, the Directors, other than the Directors nominated by the Nominating Members party to such Combination shall remove one (1) of the Directors nominated by the Nominating Members party to such Combination.
- (c) If a Board vacancy is created as a result of the operation of this <u>Section 8.17</u>, the number of Directors constituting the full Board shall automatically be reduced by one (1) Director.
- 8.18 <u>Governance of Company Subsidiaries; Certain Agreements Related to the Exchange</u> Boards.
- (a) Unless otherwise set forth in the limited liability company agreement of a Company Subsidiary or the remainder of this Section 8.18, each Company Subsidiary that is a limited liability company shall, to the extent permitted by Applicable Law, be managed by the Company as the sole member and managing member (or, if applicable, the sole voting member and managing member), subject to the provisions of this Agreement requiring approvals of certain matters by the Board. Notwithstanding the foregoing, the Company and the Members agree that each Regulated Securities Exchange Subsidiary shall be managed by its respective Exchange Board.
- (b) Commencing on the Registration Date, notwithstanding Section 8.18(a) or any other provision of this Agreement to the contrary, for so long as the Company Controls any Regulated Securities Exchange Subsidiary, the Company shall ensure that the Directors, the Alternate Directors, the Officers, the employees and the agents of the Company give due regard to the preservation of the independence of the self-regulatory function of each Regulated Securities Exchange Subsidiary and to its obligations to investors and the general public and shall not take any actions which would interfere with the effectuation of any decisions by the Exchange Board of such Regulated Securities Exchange Subsidiary relating to its regulatory functions (including disciplinary matters) or which would interfere with the ability of such Regulated Securities Exchange Subsidiary to carry out its responsibilities under the Exchange Act. No present or past Member, employee, beneficiary, agent, customer, creditor, regulatory authority (or member thereof) or other Person shall have any rights against the Company or any Director, Officer, employee or

agent of the Company under this <u>Section 8.18(b)</u>. The Company shall take reasonable steps necessary to cause its Directors, Officers, employees and agents, prior to accepting a position as a Director, Officer, employee or agent, as applicable, of the Company to consent in writing to the applicability to them of this <u>Section 8.18(b)</u> and <u>Sections 11.3(h)</u>, <u>12.2</u> and <u>15.12(b)</u>.

- (c) Each Exchange Director Nominating Member shall have the right, but not the obligation, to nominate an Exchange Director in the manner set forth in this <u>Section 8.18(c)</u>.
  - (i) There shall be two (2) Exchange Directors on each of MEMX and MX2 at any given time.
  - (ii) The Exchange Directors shall serve on an Exchange Board for one (1)-year terms; provided, however, that the Exchange Director Nominating Members acknowledge and agree that the initial term of the first two (2) Exchange Directors of MX2 LLC shall also include the period between the date on which MX2 LLC was registered with the SEC as a national securities exchange and the date of the first annual meeting of the members of MX2 LLC at which meeting the Company shall elect the members of the Exchange Board of MX2 LLC for one (1)-year terms, as set forth in the Regulated Securities Exchange Subsidiary LLC Agreement of MX2 LLC (such period, the "MX2 LLC Stub Period").
  - (iii) Subject to the other provisions of this Section 8.18, the Exchange Director Nominating Members shall have the right to nominate Exchange Directors (A) of MEMX LLC in the order set forth in Exhibit I-1 (such order, the "MEMX LLC Exchange Director Nomination Rotation"), it being understood that the MEMX LLC Exchange Director Nomination Rotation commenced August 7, 2020, and the initial term of the Exchange Directors of MEMX LLC (the "Initial Term") also included the period between the Registration Date and the first (1st) meeting of the Exchange Board of MEMX LLC at which such Exchange Directors were elected, and (B) MX2 LLC in the order set forth in Exhibit I-2 (such order, the "MX2 LLC Exchange Director Nomination Rotation").
  - (iv) If at any time an Exchange Director Nominating Member reaches its turn in (A) the MEMX LLC Exchange Director Nomination Rotation but already has a director, an officer or an employee serving on the Exchange Board of MEMX LLC (in a capacity other than as the designee of such Exchange Director Nominating Member pursuant to the terms of this Section 8.18), such Exchange Director Nominating Member shall have the right to nominate an Exchange Director of MEMX LLC upon the resignation or removal of such director, officer, or employee from the Exchange Board of MEMX LLC, and (B) the MX2 LLC Exchange Director Nomination Rotation but already has a director, an officer or an employee serving on the Exchange Board of MX2 LLC (in a capacity other than as the designee of such Exchange Director Nominating Member pursuant to the terms of this Section 8.18), such Exchange Director Nominating Member shall have the right to nominate an Exchange Director of MX2 LLC upon the resignation or removal of such director, officer, or employee from the Exchange Board of MX2 LLC.

- (v) An Exchange Director Nominating Member may waive its right to nominate an Exchange Director to one or more Exchange Boards by giving a written notice to the Company. Such notice shall state whether such waiver is revocable or irrevocable and to which Exchange Boards such waiver applies. Upon receiving any nomination waiver, the Company shall promptly (but no later than five (5) Business Days following receipt of such notice) notify the other Exchange Director Nominating Members in writing. If such notice provides that such waiver is revocable, such Exchange Director Nominating Member may revoke such waiver by written notice to the Company, in which case such Exchange Director Nominating Member will once again be eligible to nominate an Exchange Director to one or more Exchange Boards pursuant to this Section 8.18. If at any time an Exchange Director Nominating Member waives its right to nominate one or more Exchange Directors, such Exchange Director Nominating Member shall cause its nominee then serving on each applicable Exchange Board (if applicable) to resign from each such Exchange Board no later than the next Business Day following the date on which such Exchange Director Nominating Member provides its waiver notice pursuant to this Section 8.18(c)(v).
- (vi) In the event of a Combination of Exchange Director Nominating Members:
  - (A) following such Combination, the surviving Affiliated group of Exchange Director Nominating Members shall have the right to nominate one (1) Exchange Director to an Exchange;
  - (B) if both applicable Exchange Director Nominating Members then have their respective nominees serving on each applicable Exchange Board at the time of the Combination, the surviving Affiliated group shall cause one (1) of such nominees to resign from each such Exchange Board no later than the next Business Day following the date on which the Combination becomes effective;
  - (C) if such surviving Affiliated group does not remove an Exchange Director pursuant to Section 8.18(c)(vi)(B), the Company shall remove one (1) Exchange Director at its option on the following Business Day in accordance with each applicable Regulated Securities Exchange Subsidiary LLC Agreement;
  - (D) if only one (1) of the applicable Exchange Director Nominating Members has its respective nominee serving on each applicable Exchange Board at the time of the Combination, such nominee shall remain on each such Exchange Board (and, for the sake of clarity, the surviving Affiliated group shall take the place of the Exchange Director Nominating Member which nominated such nominee in the Exchange Director Nomination Rotation) and the other applicable Exchange Director Nominating Member shall be removed from the MEMX LLC Exchange Director Nomination Rotation and the MX2 LLC Exchange Director Nomination Rotation; and

(E) if neither of the applicable Exchange Director Nominating Members has its respective nominee serving on each applicable Exchange Board at the time of the Combination, then the Exchange Director Nominating Member which then has a higher position in (1) the MEMX LLC Exchange Director Nomination Rotation shall remain in the MEMX LLC Exchange Director Nomination Rotation (and, for the sake of clarity, the surviving Affiliated group shall take the place of such Exchange Director Nominating Member in the MEMX LLC Exchange Director Nomination Rotation), and (2) the MX2 LLC Exchange Director Nomination Rotation (and, for the sake of clarity, the surviving Affiliated group shall take the place of such Exchange Director Nominating Member in the MX2 LLC Exchange Director Nomination Rotation) and the other Exchange Director Nominating Member shall be removed from the Exchange Director Nomination Rotation.

## (vii) If:

- (A) as a result of operation of Section 8.18(c)(v), Section 8.18(c)(vi), Section 8.18(f), or a Combination, as applicable, the Company has only two (2) Exchange Director Nominating Members that are then eligible to nominate Exchange Directors, such Exchange Director Nominating Members shall nominate Exchange Directors on an annual basis, unless and until the third (3<sup>rd</sup>) Exchange Director Nominating Member is once again eligible to nominate an Exchange Director, in which case (I) the nominees of such two (2) Exchange Director Nominating Members shall serve as Exchange Directors until the next annual meeting of the members of the applicable Regulated Securities Exchange Subsidiary, (II) prior to such next annual meeting of members of the applicable Regulated Securities Exchange Subsidiary, all of the Exchange Director Nominating Members which are then able to nominate an Exchange Director shall agree among themselves on the order in which they shall nominate the two (2) Exchange Directors and shall notify the Company to update Exhibit I (and if such Exchange Director Nominating Members fail to so agree at least five (5) Business Days prior to such next annual meeting of the members of the applicable Regulated Securities Exchange Subsidiary the Company shall conduct a lottery determining such order), (III) the Exchange Director Nominating Members which have the right to do so shall nominate the Exchange Directors which shall be elected by the Company at such annual meeting of members of the applicable Regulated Securities Exchange Subsidiary in accordance with the applicable Regulated Securities Exchange Subsidiary LLC Agreement, and (IV) the order determined pursuant to clause (II) shall thereafter serve as the rotation for the nomination of Exchange Directors by Exchange Director Nominating Members;
- (B) as a result of operation of <u>Section 8.18(c)(v)</u>, <u>Section 8.18(c)(vi)</u>, <u>Section 8.18(f)</u>, or a Combination, as applicable, the Company has only one (1) Exchange Director Nominating Member that is eligible to nominate an Exchange Director, such Exchange Director Nominating Member shall nominate one

- (1) Exchange Director on an annual basis and the manner in which the second (2<sup>nd</sup>) Exchange Director is nominated shall be determined by the Board by Supermajority Board Vote. If thereafter another Exchange Director Nominating Member is once again eligible to nominate an Exchange Director, the provisions of Section 8.18(c)(vii)(A) shall apply; and
- (C) as a result of operation of Section 8.18(c)(v), Section 8.18(c)(vi), Section 8.18(f), or a Combination, as applicable, the Company does not have any Exchange Director Nominating Members that are eligible to nominate an Exchange Director, the manner in which the Exchange Directors are nominated shall be determined by the Board by Supermajority Board Vote; provided, that if the Board wishes to designate any Class A Member as an Exchange Director Nominating Member, the Board must first obtain written consent of such Class A Member. If thereafter one or more Exchange Director Nominating Members are once again eligible to nominate an Exchange Director, the provisions of Section 8.18(c)(vii)(A) or Section 8.18(c)(vii)(B), as applicable, shall apply.
- (viii) Election of an Exchange Director nominee nominated by an Exchange Director Nominating Member to an Exchange Board shall be subject to Applicable Law and reasonable qualifications for members of such Exchange Board established by the applicable Regulated Securities Exchange Subsidiary.
- (d) The Company shall, in each case, elect the nominees of the Exchange Director Nominating Members to the applicable Exchange Board in accordance with the applicable Regulated Securities Exchange Subsidiary LLC Agreement. The Company shall continue to elect each such nominee to such Exchange Board subject to Section 8.18(c) and the Regulated Securities Exchange Subsidiary LLC Agreement (as applicable at the time of such election).
- (e) Notwithstanding the provisions of <u>Section 8.18(c)</u> and <u>8.18(d)</u>, in its capacity as a member of each Regulated Securities Exchange Subsidiary:
  - (i) the Company shall vote in favor of only those directors, members of the Nominating Committee of such Regulated Securities Exchange Subsidiary, and members of the Member Nominating Committee of such Regulated Securities Exchange Subsidiary who are nominated in the manner set forth in the Regulated Securities Exchange Subsidiary LLC Agreement of such Regulated Securities Exchange Subsidiary;
  - (ii) with respect to any action taken by written consent, the Company shall cause to be validly executed only the written consents electing only the directors, members of the Nominating Committee, and members of the Member Nominating Committee of such Regulated Securities Exchange Subsidiary referred to in the preceding clause (i); and
  - (iii) the Company shall not remove (A) any Exchange Director of such Regulated Securities Exchange Subsidiary nominated by an Exchange Director Nominating

Member, or (B) any Exchange Board Observer of such Regulated Securities Exchange Subsidiary appointed by an Exchange Board Observer Appointing Member, except as (I) required by Applicable Law or the Exchange Board of such Regulated Securities Exchange Subsidiary (in accordance with the Regulated Securities Exchange Subsidiary LLC Agreement of such Regulated Securities Exchange Subsidiary) or (II) requested by (x) the Exchange Director Nominating Member which nominated such Exchange Director, or (y) the Exchange Board Observer Appointing Member that appointed such Exchange Board Observer, as applicable.

- (f) Each Exchange Director Nominating Member shall retain the right to nominate an Exchange Director (subject to the MEMX LLC Exchange Director Nomination Rotation and the MX2 LLC Exchange Director Nomination Rotation, as applicable) for so long as such Exchange Director Nominating Member owns at least 2,500,000 Class A Units (subject to adjustment in the event of any Unit split, Unit combination, reorganization, reclassification, recapitalization or the like).
  - (i) Such Exchange Director Nominating Member's right to nominate an Exchange Director shall terminate immediately following a Transfer of Units by such former Exchange Director Nominating Member, which, after giving effect to such Transfer, results in such former Exchange Director Nominating Member owning a number of Units that is less than the threshold specified in the preceding sentence.
  - (ii) The Exchange Director slot which is vacant as a result of operation of this Section 8.18(f) shall be subject to Section 8.18(c)(vii).
  - (iii) If an Exchange Director Nominating Member Transfers a number of Units equal to at least 2,500,000 Class A Units (subject to adjustment in the event of any Unit split, Unit combination, reorganization, reclassification, recapitalization or the like) to a Nominating Transferee, the Board may determine to grant such Nominating Transferee the right to nominate one (1) Exchange Director; provided, however, that such Nominating Transferee's nominee for Exchange Director shall not be appointed to serve on any Exchange Board unless such nominee's appointment is approved by the Board.
- (g) As of the Effective Date each Exchange Board Observer Appointing Member shall have the right, but not the obligation, to appoint one (1) observer to each Exchange Board (each such observer, an "Exchange Board Observer") for so long as such Exchange Board Observer Appointing Member holds an aggregate number of Class A Units, Class C Units and/or Class D Units equal to at least 1,250,000 (subject to adjustment in the event of any Unit split, Unit combination, reorganization, reclassification, recapitalization or the like). If an Exchange Board Observer Appointing Member has appointed an Exchange Board Observer and, thereafter, ceases to hold an aggregate number of Class A Units, Class C Units and/or Class D Units equal to at least 1,250,000 (subject to adjustment in the event of any Unit split, Unit combination, reorganization, reclassification, recapitalization or the like), such former Exchange Board Observer Appointing Member shall no longer have the right to appoint an Exchange Board Observer and the Exchange

Board Observer appointed by such former Exchange Board Observer Appointing Member shall automatically and immediately be deemed removed from his or her position as such.

- (i) Notwithstanding the foregoing, when an Exchange Director Nominating Member reaches its turn in the MEMX LLC Exchange Director Nomination Rotation or the MX2 LLC Exchange Director Nomination Rotation, as applicable, and its Exchange Director nominee is elected to the applicable Exchange Board and is seated as such, its Exchange Board Observer shall be automatically and immediately deemed removed from his or her position as such with respect to such Exchange Board and, for as long as its Exchange Director nominee is serving as an Exchange Director on such Exchange Board, such Exchange Director Nominating Member shall have no right to nominate an Exchange Board Observer to such Exchange Board. Following the removal of the Exchange Director nominee of such Exchange Director Nominating Member from such Exchange Board, such Exchange Director Nominating Member shall once again have the right to appoint an Exchange Board Observer.
- (ii) For the sake of clarity, an Exchange Director Nominating Member which has waived its right to nominate an Exchange Director to an Exchange Board may nominate an Exchange Board Observer to such Exchange Board.
- (iii) If a Member Transfers an aggregate number of Class A Units, Class C Units and/or Class D Units equal to at least 1,250,000 (subject to adjustment in the event of any Unit split, Unit combination, reorganization, reclassification, recapitalization or the like) to a third party, the Board may determine to grant such Transferee the right to appoint one (1) Exchange Board Observer; provided, however, that such Transferee's designee for Exchange Board Observer shall not be appointed as an Exchange Board Observer to any Exchange Board unless such designee's appointment is approved by the Board.
- (iv) Following a Combination of two Exchange Board Observer Appointing Members, the surviving Affiliated group shall have the right to nominate one (1) Exchange Board Observer to an Exchange and such surviving Affiliated group shall cause the removal of one (1) Exchange Board Observer from such Exchange no later than one (1) Business Day following the consummation of the Combination and thereafter such surviving Affiliated group shall have the right to nominate one (1) Exchange Board Observer to an Exchange. If such Affiliated group does not remove such Exchange Board Observer, the Company shall remove one (1) Exchange Board Observer to an Exchange on the following Business Day.
- (h) If there are Company Subsidiaries other than the Regulated Securities Exchange Subsidiaries that are not member-managed limited liability companies, then, to the extent permitted by Applicable Law, the governing documents of each such Company Subsidiary shall include a prohibition on taking any action that if taken by the Company would require the approval of the Board, without first obtaining such approval.

(i) The Company shall ensure that the governance of any Company Subsidiary, including, as appropriate, its constitutive documents, is conducted in a manner consistent, to the maximum extent possible and permitted by Applicable Law, with the provisions of this Article VIII, including as applicable (i) the necessity for obtaining any Board approvals as set forth in this Agreement, and (ii) each Nominating Member having a right to nominate one (1) member to the board of directors or an equivalent governing body, if any, of each Company Subsidiary, unless otherwise approved by the Board by Supermajority Board Vote; provided that each Regulated Securities Exchange Subsidiary shall be governed by an Exchange Board (which shall be constituted as set forth in the Regulated Securities Exchange Subsidiary LLC Agreement), as and when required pursuant to Section 8.18(a).

#### 8.19 Industry Advisory Board.

- (a) The Board may, upon a determination to do so by Supermajority Board Vote, establish an advisory board with industry representation (the "Industry Advisory Board"). If such Industry Advisory Board is established, it shall be comprised of (i) one representative of each Member which is a Nominating Member, for so long as such Member remains a Nominating Member or is entitled to appoint a Board Observer pursuant to the terms of this Agreement, in each case if such Member desires to appoint a representative to the Industry Advisory Board, and (ii) if so determined by the Board, representatives of such other members of the national securities exchange operated by a Regulated Securities Exchange Subsidiary as determined by the Board (each such representative referred to herein as an "Industry Advisory Board Member"). With respect to Members which appoint an Industry Advisory Board Member pursuant to the immediately preceding clause (i), if such Member no longer has the right to nominate at least one (1) Director hereunder, unless the Board determines otherwise by Supermajority Board Vote, such Member shall no longer have the right to nominate an Industry Advisory Board Member and the Industry Advisory Board Member nominated by such Member shall automatically and immediately be removed from the Industry Advisory Board.
- (b) The Industry Advisory Board shall not be a Board committee, and no overlap of membership between the Board or any Board committee, on the one hand, and the Industry Advisory Board, on the other hand, shall make the Industry Advisory Board a Board committee. The Industry Advisory Board shall provide advice to the Board and the management of the Company and each Regulated Securities Exchange Subsidiary with guidance on technical and operational matters relating to the national securities exchange operated by such Regulated Securities Exchange Subsidiary, and on such other matters as the Board, the Company, or such Regulated Securities Exchange Subsidiary as the case may be, may request from time to time. The Industry Advisory Board shall have an advisory role only and shall not have the authority to act for or on behalf of, or to bind, the Company or any Company Subsidiaries; provided, however, the Board and the Company's management shall consider the advice of the Industry Advisory Board in good faith but shall not be bound thereby. The Board shall not have the authority to delegate any of its powers, including the right to authorize any action required to be approved by the Board pursuant to this Agreement or Applicable Law, to the Industry Advisory Board.

(c) If any Company Subsidiary (including any Regulated Securities Exchange Subsidiary) establishes any committee or advisory board which has functions similar to the contemplated functions of the Industry Advisory Board, each Member which is a Nominating Member, for so long as such Member remains a Nominating Member or is entitled to appoint a Board Observer pursuant to the terms of this Agreement, would have the right, but not the obligation, to appoint a representative to such board or committee.

# ARTICLE IX. PRE-EMPTIVE RIGHTS

# 9.1 Pre-emptive Right.

- (a) <u>Issuance of New Securities</u>. The Company hereby grants to (i) each Class A Member, (ii) each Class C Member, (iii) each Class D Member and (iv) each Converted Common Member (each, a "<u>Pre-emptive Member</u>") the right to purchase its Pro Rata Portion of any New Securities that the Company or any Company Subsidiary may from time to time propose to issue or sell to any party between the date hereof and the consummation of a Qualified Public Offering. The Company shall ensure that the Company Subsidiaries comply with the provisions of this <u>Article IX</u>.
- Definition of New Securities. As used herein, the term "New Securities" shall mean any authorized but unissued Units and any Unit Equivalents convertible into Units, exchangeable or exercisable for Units, or providing a right to subscribe for, purchase or acquire Units, or, in each of the foregoing cases, if such New Securities are issued by a Company Subsidiary any equity interests or Equity Interest Equivalents in such Company Subsidiary; provided, that the term "New Securities" shall not include Units, Unit Equivalents, equity interests or Equity Interest Equivalents issued or sold by the Company or any Company Subsidiary in connection with: (i) a grant to any existing or prospective Directors, Officers or other service providers of the Company pursuant to any incentive plan of the Company or similar equity-based plans or other compensation agreement (including the Incentive Plan); (ii) the conversion or exchange of any validly issued securities of the Company or any Company Subsidiary into Units or other equity interests, or the exercise of any warrants or other rights to acquire Units or other equity interests; (iii) any acquisition by the Company or any Company Subsidiary of any equity interests, assets, properties or business of any Person; (iv) any merger, consolidation or other business combination involving the Company or any Company Subsidiary; (v) the commencement of any Public Offering; (vi) without prejudice to clause (iv) above, any issuance of Units, Unit Equivalents, equity interests or Equity Interest Equivalents in a transaction which results in a Change of Control of the Company or any Company Subsidiary, with respect to which the Board has waived the rights of the Members under this Section 9.1 pursuant to a Supermajority Board Vote; (vii) conversion of Class C Units and/or Class D Units, as applicable, pursuant to Sections 3.10(d), 3.10(e) or 3.11, as applicable; or (viii) to the extent not covered by clauses (i) through (vii) above, Common Units issued in the manner set forth in clauses (A) through (H) of the definition of Exempted Securities.
- (c) <u>Additional Issuance Notices</u>. The Company shall give written notice (an "<u>Issuance Notice</u>") of any proposed issuance or sale described in <u>Section 9.1(a)</u> to the Pre-emptive

Members within five (5) Business Days following any meeting of the Board at which any such offer to sell additional Units is approved. The Issuance Notice shall set forth the material terms and conditions of the proposed issuance or sale, including:

- (i) the maximum number and description of the New Securities proposed to be issued and the percentage of the Company's Units then outstanding on a Fully Diluted Basis (both in the aggregate and with respect to each type, class or series of Units proposed to be issued) that such issuance would represent;
- (ii) the proposed issuance date, which shall be at least twenty (20) Business Days from the date of the Issuance Notice; and
  - (iii) the proposed purchase price per unit of the New Securities.
- (d) The Issuance Notice shall also be accompanied by a current copy of the Members Schedule indicating the Pre-emptive Members' holdings of Units (and, if applicable, the Company's and the Members' (as applicable) ownership interests in any Company Subsidiary) in a manner that enables each Pre-emptive Member to calculate its Pro Rata Portion of any New Securities. For the purposes of this <u>Section 9.1</u>:
  - (i) if the New Securities are being issued by a wholly-owned Company Subsidiary, the Pro Rata Portion of each Pre-emptive Member for the purposes of this Section 9.1 shall be the same as Pro Rata Portion of such Pre-emptive Member;
  - (ii) if the New Securities are being issued by a Company Subsidiary that is not a wholly-owned Subsidiary of the Company, the Pro Rata Portion of each Pre-emptive Member for the purposes of this Section 9.1 shall equal the product of (A) the ownership interest of the Company in such Company Subsidiary (expressed as a percentage) times (B) such Pre-emptive Member's Pro Rata Portion; and
  - (iii) if the New Securities are being issued by a Company Subsidiary and a Pre-emptive Member holds both (A) Class A Units, Class C Units and/or Class D Units, and (B) equity interests directly in such Company Subsidiary, the Pro Rata Portion of such Pre-emptive Member for the purposes of this <u>Section 9.1</u> shall be determined using a fraction, expressed as a percentage, the numerator of which shall be the aggregate direct and indirect ownership interest of such Pre-emptive Member in such Company Subsidiary and the denominator of which shall be the sum of the aggregate direct and indirect ownership interests in such Company Subsidiary of all Members and direct holders of ownership interests in such Company Subsidiary.
- (e) <u>Exercise of Pre-emptive Rights</u>. Each Pre-emptive Member shall for a period of twenty (20) Business Days following the receipt of an Issuance Notice (the "<u>Exercise Period</u>") have the right to elect irrevocably to purchase all or any portion of its Pro Rata Portion of any New Securities at the respective purchase prices set forth in the Issuance Notice by delivering a written notice to the Company (an "<u>Acceptance Notice</u>") specifying the number of New Securities it desires

to purchase. The delivery of an Acceptance Notice by a Pre-emptive Member shall be a binding and irrevocable offer by such Member to purchase the New Securities described therein. The failure of a Pre-emptive Member to deliver an Acceptance Notice by the end of the Exercise Period shall constitute a waiver of its rights under this <u>Section 9.1</u> with respect to the purchase of such New Securities, but shall not affect its rights with respect to any future issuances or sales of New Securities.

- (f) Over-allotment. No later than ten (10) Business Days following the expiration of the Exercise Period, the Company shall notify each Pre-emptive Member in writing of the number of New Securities that each Pre-emptive Member has agreed to purchase (including, for the avoidance of doubt, where such number is zero) (the "Over-allotment Notice"). Each Pre-emptive Member exercising its rights to purchase its Pro Rata Portion of the New Securities in full (an "Exercising Member") shall have a right of over-allotment such that if any other Pre-emptive Member has failed to exercise its right under this Section 9.1 to purchase its full Pro Rata Portion of the New Securities (each, a "Non-Exercising Member"), such Exercising Member may purchase its Pro Rata Portion of such Non-Exercising Member's allotment by giving written notice to the Company within five (5) Business Days of receipt of the Over-allotment Notice (the "Over-allotment Exercise Period").
- Completion of Proposed Issuance. Following the expiration of the Exercise (g) Period or, if applicable, the Over-allotment Exercise Period, the Company or the applicable Company Subsidiary shall be free to complete the proposed issuance or sale of New Securities described in the Issuance Notice with respect to which Pre-emptive Members declined to exercise the pre-emptive right set forth in this Section 9.1 on terms no less favorable to the Company or such Company Subsidiary, as applicable, than those set forth in the Issuance Notice (except that the amount of New Securities to be issued or sold by the Company or such Company Subsidiary, as applicable, may be reduced); provided, that: (i) such issuance or sale is closed within twenty (20) Business Days after the expiration of the Exercise Period or, if applicable, the Over-allotment Exercise Period (subject to the extension of such twenty (20) Business Day period for a reasonable time not to exceed forty (40) Business Days to the extent reasonably necessary to obtain any thirdparty approvals); and (ii) for the avoidance of doubt, the price at which the New Securities are sold is at least equal to or higher than the purchase price described in the Issuance Notice. In the event the Company or the applicable Company Subsidiary has not sold such New Securities within such time period, the Company or such Company Subsidiary shall not thereafter issue or sell any New Securities without first again offering such securities to the Members in accordance with the procedures set forth in this Section 9.1.
- (h) <u>Closing of the Issuance</u>. The closing of any purchase by any Pre-emptive Member shall be consummated concurrently with the consummation of the issuance or sale described in the Issuance Notice. Upon the issuance or sale of any New Securities in accordance with this <u>Section 9.1</u>, the Company or the applicable Company Subsidiary shall deliver the New Securities free and clear of any liens (other than those arising hereunder and those attributable to the actions of the purchasers thereof), and the Company or such Company Subsidiary, as applicable, shall so represent and warrant to the purchasers thereof, and further represent and warrant to such purchasers that such New Securities shall be, upon issuance thereof to the Exercising Members and

after payment therefor, duly authorized, validly issued, fully paid and non-assessable. The Company or the applicable Company Subsidiary, in the discretion of the Board, may deliver to each Exercising Member certificates evidencing the New Securities. Each Exercising Member shall deliver to the Company or the applicable Company Subsidiary, as applicable, the purchase price for the New Securities purchased by it by certified or bank check or wire transfer of immediately available funds. Each party to the purchase and sale of New Securities shall take all such other actions as may be reasonably necessary to consummate the purchase and sale including, without limitation, entering into such additional agreements as may be necessary or appropriate.

- Emergency Funding. If by Supermajority Board Vote, the Board (i) approves (i) the expedited issuance of New Securities for emergency purposes in order for the Company to continue the operation of its business, and (ii) approves that such issuance is imminently needed such that compliance with Sections 9.1 (a)-(h) would be impracticable (an "Emergency Funding"), notwithstanding any provision hereof to the contrary, the Company may solicit additional immediate funding from the Members and, in lieu of complying with the other provisions of this Section 9.1, the Company shall instead give notice to the Members within twenty (20) calendar days after the issuance of such New Securities (the "Post-Closing Preemptive Rights Notice"). Such Post-Closing Preemptive Rights Notice shall comply with the requirements of an Issuance Notice and the other provisions of this Section 9.1 (other than the requirement to provide notice in advance of funding, which shall be superseded by this Section 9.1(i)). Each Member shall have twenty (20) Business Days from the date of the Post-Closing Preemptive Rights Notice to elect to (but, for the avoidance of doubt, shall not be obligated to) purchase the New Securities from the Company that would, if purchased by such Member, maintain such Member's Pro Rata Portion in effect immediately prior to the Emergency Funding.
- (j) <u>Limitations on Ownership</u>. Notwithstanding the foregoing, a Member's ability to purchase New Securities pursuant to this <u>Section 9.1</u> shall be subject to the limitations set forth in <u>Section 3.5</u>.

# ARTICLE X. TRANSFER

#### 10.1 General Restrictions on Transfer.

- (a) (i) Each Member acknowledges and agrees that, until the consummation of a Qualified Public Offering, such Member (or any Permitted Transferee of such Member) shall not Transfer any Units or Unit Equivalents except as permitted pursuant to Section 3.8, this Section 10.1, Sections 10.2 or 10.6, or in accordance with the procedures described in Sections 10.3, 10.4 or 10.5 as applicable.
  - (ii) Reserved.
  - (iii) Reserved.

- (iv) No Transfer of Units or Unit Equivalents to a Person not already a Member of the Company shall be deemed completed until the prospective Transferee is admitted as a Member of the Company in accordance with Section 4.1 hereof.
- (b) Notwithstanding any other provision of this Agreement (including <u>Section 10.2</u>), prior to the consummation of a Qualified Public Offering, each Member agrees that it shall not, directly or indirectly, Transfer any of its Units or Unit Equivalents, and the Company agrees that it shall not issue any Units or Unit Equivalents:
  - (i) except as permitted under the Securities Act and other applicable federal or state securities or blue sky laws, and then, with respect to a Transfer of Units or Unit Equivalents, if requested by the Company, only upon delivery to the Company of an opinion of counsel in form and substance satisfactory to the Company to the effect that such Transfer may be effected without registration under the Securities Act;
  - (ii) if such Transfer or issuance would cause the Company to fail to qualify for the safe harbor of Treasury Regulation Section 1.7704-1(h)(1)(ii), including the look-through rule in Treasury Regulation Section 1.7704-1(h)(3);
  - (iii) if such Transfer or issuance would affect the Company's existence or qualification as a limited liability company under the Delaware Act;
  - (iv) if such Transfer or issuance would cause the Company to lose its status as a partnership for federal income tax purposes;
  - (v) if such Transfer or issuance would cause the Company or any of the Company Subsidiaries to be required to register as an investment company under the Investment Company Act of 1940, as amended; or
  - (vi) if such Transfer or issuance would cause the assets of the Company or any of the Company Subsidiaries to be deemed "Plan Assets" as defined under the Employee Retirement Income Security Act of 1974 or its accompanying regulations or result in any "prohibited transaction" thereunder involving the Company or any Company Subsidiary.

In any event, the Board may refuse the Transfer to any Person if the Board has received an opinion of counsel to the effect that such Transfer would have a material adverse effect on the Company as a result of any regulatory or other restrictions imposed by any Governmental Authority.

(c) Any Transfer or attempted Transfer of any Units or Unit Equivalents in violation of any provisions of this Agreement shall be null and void, no such Transfer shall be recorded on the Company's books and the purported Transferee in any such Transfer shall not be treated (and the purported Transferor shall continue be treated) as the owner of such Units or Unit

Equivalents for all purposes of this Agreement, including without limitation, voting, payment of dividends and distributions with respect to such Units whether upon liquidation or otherwise.

- (d) For the avoidance of doubt, any Transfer of Units or Unit Equivalents permitted by Section 10.2 or made in accordance with the procedures described in Section 10.3, Section 10.4, Section 10.5 or 10.6, as applicable, and purporting to be a sale, transfer, assignment or other disposal of the entire Membership Interest represented by such Units or Unit Equivalents, inclusive of all the rights and benefits applicable to such Membership Interest as described in the definition of the term "Membership Interest," shall be deemed a sale, transfer, assignment or other disposal of such Membership Interest in its entirety as intended by the parties to such Transfer, and shall not be deemed a sale, transfer, assignment or other disposal of any less than all of the rights and benefits described in the definition of the term "Membership Interest," unless otherwise explicitly agreed to by the parties to such Transfer.
- 10.2 <u>Permitted Transfers</u>. The provisions of <u>Sections 10.1(a)</u>, <u>10.3</u>, <u>10.4</u> and <u>10.5</u> shall not apply to any of the following Transfers by any Member of any of its Units or Unit Equivalents:
- With respect to any Member that is an entity, to (i) any Affiliate of such Member, provided, however, that in the event that Units are Transferred to an Affiliate of such Member and such Affiliate subsequently becomes a non-Affiliate of such Member, then, absent a Supermajority Board Vote to allow such non-Affiliate to retain its Units, such Units must be Transferred back to the Member from which the Units were received at such time, (ii) to any Person which acquires substantially all of the assets of such Member, so long as such Member has, immediately prior to (and following) such acquisition, material assets and/or operations other than the Units, (iii) to any Person which, through a merger, consolidation, recapitalization, sale of equity interests or other transaction or series of transactions involving such Member, owns in the surviving entity after the closing of such transaction(s) a majority of the outstanding equity securities in such Member when it did not own a majority of the equity securities in such Member immediately prior to such transaction, so long as such Member or the other Affiliates of such Member involved in such transaction(s) and which such Person Controls after such closing had material assets and/or operations other than Units immediately prior to (and following) such closing or (iv) to the extent not covered by any of the clauses (i) through (iii) above, to any Transferee the Transfer of Interests to which is required under Applicable Law or by a Governmental Authority;
- (b) With respect to any Member that is an individual, by will or by the laws of intestate succession, to such Member's executors, administrators, testamentary trustees, legatees or beneficiaries; <u>provided</u> that any Member who Transfers Units shall remain bound by the provisions of <u>Section 11.1</u>; or
  - (c) Pursuant to a Public Offering.
- (d) The provisions of <u>Section 10.2(a)</u> constitute an individual agreement between the Company, on the one hand, and each applicable Member, on the other hand, and do not constitute an agreement among the Members. Accordingly, only the Company shall have the right

to enforce Section 10.2(a) against any Member, and no Member shall have the right to enforce Section 10.2(a) against any other Member.

# 10.3 Right of First Offer.

- (a) Offered Units. At any time prior to the consummation of a Qualified Public Offering, and subject to the terms and conditions specified in Section 10.1, Section 10.2, this Section 10.3 and Section 10.5, if any Class A Member, Class C Member, Class D Member or Common Member (the "Offering Member") desires to Transfer all or a portion of its Units (or applicable Unit Equivalents), the Company, first, and each other Member constituting a ROFO Rightholder, second, shall have a right of first offer with respect to such Units (or applicable Unit Equivalents) (such Units or Unit Equivalents, the "Offered Units"). As used herein, the term "ROFO Rightholders" shall mean all Class A Members, Class C Members, Class D Members and Common Members, other than the Offering Member, holding greater than two percent (2%) of the total number of outstanding Units (or applicable Unit Equivalents).
- (b) Offering; Exceptions. Each time the Offering Member desires to Transfer any of its Units (or applicable Unit Equivalents) (other than Transfers that (i) are permitted by Section 10.2 or (ii) are required to be made by a Member pursuant to Section 10.4), the Offering Member shall first make an offering of the Offered Units to the Company, first, and the ROFO Rightholders, second, all in accordance with the following provisions of this Section 10.3, prior to Transferring such Offered Units to any third party Transferee.

# (c) <u>Offer Notice</u>.

- (i) The Offering Member shall give written notice (the "Offering Member Notice") to the Company and the ROFO Rightholders specifying:
  - (A) the number and series of Offered Units proposed to be Transferred by the Offering Member; and
  - (B) the purchase price per Offered Unit (which shall be payable solely in cash) and the other material terms and conditions of the Transfer; provided that if the Offered Units include Units of more than one series, (1) the Offering Member Notice shall state the number of Units of each series which constitute the Offered Units and (2) the Offering Member may, at its discretion, provide for a different price per Offered Unit with respect to each such series, in which case, the Offering Member Notice shall state each such price.
- (ii) The Offering Member Notice shall constitute the Offering Member's offer to Transfer the Offered Units to the Company and the ROFO Rightholders, which offer shall be irrevocable until the end of the ROFO Rightholder Option Period described in Section 10.3(d)(iii).

- (iii) By delivering the Offering Member Notice, the Offering Member represents and warrants to the Company and each ROFO Rightholder that:
  - (A) the Offering Member has full right, title and interest in and to the Offered Units;
  - (B) the Offering Member has all the necessary power and authority and has taken all necessary action to Transfer such Offered Units as contemplated by this <u>Section 10.3</u>; and
  - (C) the Offered Units are free and clear of any and all liens other than those arising as a result of or under the terms of this Agreement.

# (d) Exercise of Right of First Offer.

- Upon receipt of the Offering Member Notice, the Company and each (i) ROFO Rightholder shall have the right to purchase the Offered Units in the following order of priority: first, the Company shall, subject to approval by a Supermajority Board Vote, have the right to purchase all or any portion of the Offered Units in accordance with the procedures set forth in Section 10.3(d)(ii), and thereafter, the ROFO Rightholders shall have the right to purchase the Offered Units, in accordance with the procedures set forth in Section 10.3(d)(iii), to the extent the Company does not exercise its right in full. Notwithstanding the foregoing, the Company and the ROFO Rightholders may only exercise their right to purchase the Offered Units if, after giving effect to all elections made under this Section 10.3(d), no less than all of the Offered Units shall be purchased by the Company and/or the ROFO Rightholders, subject to the limitations set forth in Section 3.5 through Section 3.7. Notwithstanding the foregoing, in the event that the procedures set forth in this Section 10.3 would result in a Member's ownership exceeding the limitations set forth in Section 3.5, the amount of any such excess shall be allocated among the other Purchasing Rightholders that do not and will not exceed the limitations set forth in Section 3.5 after the exercise of their respective rights under this Section 10.3, provided, however, that if as a result of the reallocation among the Purchasing Rightholders, all of the Purchasing Rightholders' ownership would exceed the limitations set forth in Section 3.5, then any remaining excess may be offered to a third party on the same terms.
- (ii) The initial right of the Company to purchase any Offered Units shall be exercisable with the delivery of a written notice (the "Company ROFO Exercise Notice") by the Company to the Offering Member and the ROFO Rightholders within ten (10) Business Days of receipt of the Offering Member Notice (the "Company Option Period"), stating the number (including where such number is zero) and type of Offered Units the Company elects irrevocably to purchase on the terms and respective purchase prices set forth in the Offering Member Notice. The Company ROFO Exercise Notice shall be binding upon delivery and irrevocable by the Company. Notwithstanding anything to the contrary contained herein, in no event shall the Company exercise its right to acquire Units purchased pursuant to this Section 10.3(d)(ii) and assign such right to any other Person.

- than all of the Offered Units, the ROFO Rightholders shall have the right to purchase the remaining Offered Units not selected by the Company. For a period of thirty (30) Business Days following the receipt of a Company ROFO Exercise Notice in which the Company has elected to purchase less than all the Offered Units (such period, the "ROFO Rightholder Option Period"), each ROFO Rightholder shall have the right to elect irrevocably to purchase all or none of its Pro Rata Portion of the remaining Offered Units by delivering a written notice to the Company and the Offering Member (a "Member ROFO Exercise Notice") specifying its desire to purchase its Pro Rata Portion of the remaining Offered Units, on the terms and respective purchase prices set forth in the Offering Member Notice. In addition, each ROFO Rightholder shall include in its Member ROFO Exercise Notice the number of remaining Offered Units that it wishes to purchase if any other ROFO Rightholders do not exercise their rights to purchase their entire Pro Rata Portions of the remaining Offered Units. Any Member ROFO Exercise Notice shall be binding upon delivery and irrevocable by the ROFO Rightholder.
- (iv) The failure of the Company or any ROFO Rightholder to deliver a Company ROFO Exercise Notice or Member ROFO Exercise Notice, respectively, by the end of the Company Option Period or ROFO Rightholder Option Period, respectively, shall constitute a waiver of their respective rights of first offer under this <u>Section 10.3</u> with respect to the Transfer of Offered Units, but shall not affect their respective rights with respect to any future Transfers.
- (e) <u>Allocation of Offered Units</u>. Upon the expiration of the ROFO Rightholder Option Period, the Offered Units not selected for purchase by the Company pursuant to <u>Section 10.3(d)(ii)</u> shall be allocated for purchase among the ROFO Rightholders as follows:
  - (i) First, to each ROFO Rightholder having elected to purchase all or less than all of its entire Pro Rata Portion of such Units, such ROFO Rightholder's Pro Rata Portion of such Units; and
  - (ii) Second, the balance, if any, not allocated under clause (i) above (and not purchased by the Company pursuant to Section 10.3(d)(ii)), shall be allocated to those ROFO Rightholders who set forth in their Member ROFO Exercise Notices a number of Offered Units that exceeded their respective Pro Rata Portions (the "Purchasing Rightholders"), in an amount, with respect to each such Purchasing Rightholder, that is equal to the lesser of:
    - (A) the number of Offered Units that such Purchasing Rightholder elected to purchase in excess of its Pro Rata Portion; or
    - (B) the product of (x) the number of Offered Units not allocated under clause (i) (and not purchased by the Company pursuant to Section 10.3(d)(ii)), multiplied by (y) a fraction, the numerator of which is the number of Offered Units that such Purchasing Rightholder was permitted to purchase pursuant to clause (i),

and the denominator of which is the aggregate number of Offered Units that all Purchasing Rightholders were permitted to purchase pursuant to clause (i).

The process described in clause (ii) shall be repeated until no Offered Units remain or until such time as all Purchasing Rightholders have been permitted to purchase all Offered Units that they desire to purchase.

- (f) Consummation of Sale. In the event that the Company and/or the ROFO Rightholders shall have, in the aggregate, exercised their respective rights to purchase all and not less than all of the Offered Units, then the Offering Member shall sell such Offered Units to the Company and/or the ROFO Rightholders, and the Company and/or the ROFO Rightholders, as the case may be, shall purchase such Offered Units, within sixty (60) days following the expiration of the ROFO Rightholder Option Period (which period may be extended for a reasonable time not to exceed ninety (90) days to the extent reasonably necessary to obtain required approvals or consents from any Governmental Authority) on the terms set forth in the Offering Member Notice. Each Member shall take all actions as may be reasonably necessary to consummate the sale contemplated by this <u>Section 10.3(f)</u>, including, without limitation, entering into agreements and delivering certificates and instruments and consents as may be deemed necessary or appropriate. At the closing of any sale and purchase pursuant to this Section 10.3(f), the Offering Member shall deliver to the Company and/or the participating ROFO Rightholders certificates (if any) representing the Offered Units to be sold, free and clear of any liens or encumbrances (other than those contained in this Agreement), accompanied by evidence of transfer and all necessary transfer taxes paid and stamps affixed, if necessary, against receipt of the purchase price therefor from the Company and/or such ROFO Rightholders by certified or official bank check or by wire transfer of immediately available funds.
- ROFO Rightholders shall not have collectively elected to purchase all of the Offered Units the Offering Member may Transfer all of such Offered Units that are not so purchased, at a price per Offered Unit of each applicable series not less than specified in the Offering Member Notice and on other terms and conditions which are not materially more favorable in the aggregate to the proposed purchaser than those specified in the Offering Member Notice, but only to the extent that such Transfer occurs within ninety (90) days after expiration of the ROFO Rightholder Option Period. Any Offered Units not Transferred within such 90-day period shall be subject to the provisions of this Section 10.3 upon subsequent Transfer.

#### 10.4 Drag-along Rights.

(a) <u>Participation</u>. At any time prior to the consummation of a Qualified Public Offering, if the Board approves, by Supermajority Board Vote, a transaction that results in a Change of Control (such transaction, a "<u>Drag-along Sale</u>"), the Company shall deliver a written notice to the Members invoking the provisions of this <u>Section 10.4</u>, and each Member shall be obliged (unless expressly prohibited from doing so by Applicable Law (including regulatory restrictions applicable to such Member or its Affiliates)), to participate in such transaction

(including, if necessary, by converting their Unit Equivalents into the Units to be sold in the Dragalong Sale) in the manner set forth in <u>Section 10.4(b)</u>.

# (b) <u>Sale of Units</u>. Subject to compliance with <u>Section 10.4(c)</u>:

- if the Drag-along Sale is structured as a sale of less than all of the Units of the Company on a Fully Diluted Basis to a Third Party Purchaser, then each Member shall sell, with respect to each type, class or series of Units proposed by the Board to be included in the Drag-along Sale, the number of Units and/or Unit Equivalents of such type, class or series equal to the product obtained by multiplying (i) the total number of such type, class or series of Units proposed to be purchased by the Third Party Purchaser by (ii) a fraction (x) the numerator of which is equal to the number of applicable Units on a Fully Diluted Basis that are held by such Member at such time and (y) the denominator of which is equal to the number of applicable Units on a Fully Diluted Basis that are held by all Members at such time; provided that the proceeds of any such Drag-along Sale which are paid to the Members and to which this Section 10.4(b)(i) applies will be allocated among the Members based upon the classes or series of Units included or deemed to be included in the Drag-along Sale by each of the Members as if the proceeds of such Drag-along Sale were paid to the Members pursuant to Section 13.3 of this Agreement in connection with a Distribution and the Units of the Members included or deemed to be included in such Dragalong Sale were the only outstanding Units of the Company at the time of such Distribution; and
- (ii) if the Drag-along Sale is structured as a sale of all or substantially all of the consolidated assets of the Company and the Company Subsidiaries or as a merger, consolidation, recapitalization, or reorganization of the Company or other transaction requiring the consent or approval of the Members, then notwithstanding anything to the contrary in this Agreement, each Member shall vote in favor of the transaction and otherwise consent to and raise no objection to such transaction, and shall take all actions to waive any dissenters', appraisal or other similar rights that it may have in connection with such transaction. The Distribution of the aggregate consideration of such transaction shall be made in accordance with Section 13.3(c).
- (c) <u>Conditions of Sale</u>. The obligations of the Members in respect of a Dragalong Sale under this <u>Section 10.4</u> are subject to the satisfaction of the following conditions:
  - (i) the consideration to be received by each Member shall be the same form and amount of consideration to be received by the Member per Unit of each applicable type, class or series (the Distribution of which shall be made in accordance with Section 10.4(b)) and the terms and conditions of such sale shall, except as otherwise provided in Section 10.4(c)(ii), be the same as those upon which the Member sells its Units (it being understood that (A) the amount of consideration received by any particular Member shall be reduced, on a per-Unit basis, to account for any Tax Advances made to such Member which have not yet been repaid through reduction of a subsequent Distribution to such Member in the manner set forth in Section 7.1, and (B) in case of any such reduction of consideration,

any such Tax Advance shall be deemed repaid and shall not reduce any subsequent Distributions made to such Member);

- (ii) if any Member is given an option as to the form and amount of consideration to be received, the same option shall be given to all Members; <u>provided</u>, that in the event that the consideration to be received by the Members participating in the Dragalong Sale is other than cash or marketable securities, each Class A Member, Class C Member and Class D Member participating in the Drag-Along Sale may, in its respective sole discretion, elect to receive, in lieu of such other consideration, cash consideration equal to the fair market value of such consideration (as such fair market value is determined in good faith by the Board) and otherwise on the same terms and conditions upon the Members participating in the Drag-along Sale not receiving cash consideration;
- (iii) each Member shall execute the applicable purchase agreement, if applicable, and make or provide the same representations, warranties, covenants, indemnities and agreements; provided, that:
  - (A) each Member shall only be obligated to make individual representations and warranties only with respect to its title to and ownership of the applicable Units, authorization, execution and delivery of relevant documents, enforceability of such documents against such Member, but not with respect to any of the foregoing with respect to any other Members or their Units;
  - (B) all representations, warranties, covenants and indemnities shall be made by each Member severally and not jointly and any indemnification obligation shall be *pro rata* based on the consideration received by each Member, in each case in an amount not to exceed the aggregate proceeds received by the applicable Member; and
  - (C) no Member (or Affiliate thereof) shall be required to enter into (1) any non-compete, non-solicitation or no-hire provision, (2) a provision providing for the licensing of intellectual property or the delivery of any products or services, (3) any other provision that is not a customary financial term related directly to the sale of the relevant Member's Units pursuant to this <u>Section 10.4</u> or (4) any provision amending, extending or terminating any commercial agreement or other commercial relationship with the Company or any Company Subsidiary or any Transferee in the Drag-along Sale;

provided, that the fact that, in connection with a Drag-along Sale, one or more members of management of the Company enter into restrictive covenants in favor of the counterparty to the Drag-along Sale or one of its Affiliates (and, if applicable, receive additional, reasonable consideration for entering into such restrictive covenants), shall not constitute a failure to satisfy any of the conditions specified in this Section 10.4(c).

- (d) <u>Cooperation</u>. Each Member shall take all actions as may be reasonably necessary to consummate the Drag-along Sale, including, without limitation, entering into agreements and delivering certificates and instruments, in each case, consistent with the agreements being entered into and the certificates being delivered the other Members, but subject to <u>Section 10.4(c)(iii)</u>.
- (e) <u>Separate Series</u>. The Members acknowledge and agree that for the purposes of determining whether and to what extent each Member is obligated to participate in the Dragalong Sale, (i) Class A-1 Units, Class A-2 Units, Nonvoting Class A-1 Units and Nonvoting Class A-2 Units shall not be considered separate series, (ii) Class C-1 Units and Class C-2 Units shall not be considered separate series, and (iii) Class D-1 Units and Class D-2 Units shall not be considered separate series.
- (f) <u>Individual Agreement</u>. The provisions of this Section 10.4 constitute an individual agreement between the Company, on the one hand, and each applicable Member, on the other hand, and do not constitute an agreement among the Members. Accordingly, only the Company shall have the right to enforce this <u>Section 10.4</u> against any Member, and no Member shall have the right to enforce this <u>Section 10.4</u> against any other Member.

## 10.5 Tag-along Rights.

- (a) <u>Participation</u>. At any time prior to the consummation of a Qualified Public Offering, and subject to the terms and conditions specified in <u>Sections 3.5</u>, 10.1, 10.2, and 10.3, if any Class A Member, Class C Member, Class D Member or Common Member (the "<u>Selling Member</u>") proposes to Transfer (in a single transaction or a series of related transactions) any of the number of its Units which, in the aggregate, amount to at least ten percent (10%) of the Units (or any Unit Equivalents of such Units) then held by such Selling Member to any Person (the "<u>Proposed Transferee</u>"), each other Class A Member, Class C Member, Class D Member and Common Member (each, a "<u>Tag-along Member</u>") shall be permitted to participate in such sale (a "<u>Tag-along Sale</u>") on the terms and conditions set forth in this <u>Section 10.5</u>.
- (b) <u>Application of Transfer Restrictions</u>. The provisions of this <u>Section 10.5</u> shall only apply to the portion of the Transfers in which:
  - (i) the Company and ROFO Rightholders have not exercised their rights under <u>Section 10.3</u> to purchase all of the Offered Units; and
  - (ii) the Company has elected to not exercise its drag-along right under <u>Section 10.4</u> (if applicable).
- (c) <u>Sale Notice</u>. Prior to the consummation of any Transfer of Units (or any Unit Equivalents of such Units) qualifying under <u>Section 10.5(b)</u> and after satisfying its obligations pursuant to <u>Section 10.3</u>, the Selling Member shall deliver to the Company and each Tag-along Member a written notice (a "<u>Sale Notice</u>") of the proposed Tag-along Sale as soon as practicable following the expiration of the ROFO Rightholder Option Period, and in no event later than five (5)

Business Days thereafter. The Sale Notice shall make reference to the Tag-along Members' rights hereunder and shall describe in reasonable detail:

- (i) the aggregate number of Units (or any Unit Equivalents of such Units) the Proposed Transferee has offered to purchase;
  - (ii) the identity of the Proposed Transferee;
- (iii) the proposed date, time and location of the closing of the Tag-along Sale;
- (iv) the purchase price per applicable Unit and the other material terms and conditions of the Transfer; and
- (v) a copy of any form of agreement proposed to be executed in connection therewith.

# (d) Exercise of Tag-along Right.

- (i) The Selling Member and each Tag-along Member timely electing to participate in the Tag-along Sale pursuant to Section 10.5(d)(ii) shall have the right to Transfer in the Tag-along Sale the number of Units (and applicable Unit Equivalents, if any) equal to the product of (x) the aggregate number of Units (and applicable Unit Equivalents) that the Proposed Transferee proposes to buy as stated in the Sale Notice and (y) a fraction (A) the numerator of which is equal to the number of Units on a Fully Diluted Basis then held by the Selling Member or Tag-along Member, as applicable, and (B) the denominator of which is equal to the number of Units on a Fully Diluted Basis then held by the Selling Member and all of the Tag-along Members (such amount the "Tag-along Portion").
- (ii) Each Tag-along Member shall exercise its right to participate in a Tag-along Sale by delivering to the Selling Member a written notice (a "<u>Tag-along Notice</u>") stating its election to do so and specifying the number of Units and/or Unit Equivalents (up to its Tag-along Portion) to be Transferred by it no later than twenty (20) Business Days after receipt of the Sale Notice (the "<u>Tag-along Period</u>").
- (iii) The election of each Tag-along Member set forth in a Tag-along Notice shall be irrevocable, and such Tag-along Member shall be bound and obligated to consummate the Transfer on the terms and conditions set forth in this <u>Section 10.5</u>.

#### (e) Remaining Portions.

(i) If any Tag-along Member declines to exercise its right under <u>Section 10.5(d)(i)</u> or elects to exercise it with respect to less than its full Tag-Along Portion (the aggregate amount of Units resulting from all such unexercised Tag-Along Portions, the "<u>Remaining Portion</u>"), the Selling Member shall promptly deliver a written notice (a "<u>Remaining Portion Notice</u>") to those Tag-along Members who have elected to Transfer

their Tag-Along Portion in full (each, a "Fully Participating Tag-along Member"). The Selling Member and each Fully Participating Tag-along Member (with respect to any Remaining Portion) shall be entitled to Transfer, in addition to any applicable Units or Unit Equivalents already being Transferred, a number of Units (or applicable Unit Equivalents) held by it equal to the product of (x) the Remaining Portion and (y) a fraction (A) the numerator of which is equal to the number of Units (and applicable Unit Equivalents), as the case may be, then held by the Selling Member and each Fully Participating Tag-along Member, as applicable Unit Equivalents), as the case may be, then held by the Selling Member and all Fully Participating Tag-along Members.

- (ii) Each Fully Participating Tag-along Member shall exercise its right to participate in the Transfer described in <u>Section 10.5(e)(i)</u> by delivering to the Selling Member a written notice (a "<u>Remaining Tag-along Notice</u>") stating its election to do so and specifying the number of Units (or applicable Unit Equivalents), as the case may be (up to the amounts it may Transfer pursuant to <u>Section 10.5(e)(i)</u>, to be Transferred by it no later than five (5) Business Days after receipt of the Remaining Portion Notice.
- (iii) The election by each Fully Participating Tag-along Member set forth in a Remaining Tag-along Notice shall be irrevocable, and such Member shall be bound and obligated to consummate the Transfer on the terms and conditions set forth in this <u>Section</u> 10.5.
- (f) <u>Waiver</u>. Each Tag-along Member which does not deliver a Tag-along Notice in compliance with <u>Section 10.5(d)(ii)</u> shall be deemed to have waived all of such Tag-along Member's rights to participate in the Tag-along Sale with respect to the Units (and/or Unit Equivalents) owned by such Tag-along Member.
- (g) <u>Limitations on Ownership</u>. Notwithstanding the foregoing, in the event that the procedures set forth in this <u>Section 10.5</u> would result in a Person acquiring Units in excess of the limitations set forth in <u>Section 3.5</u>, the number of Units to be acquired by such Person from the Selling Member and each Tag-along Member shall be proportionately reduced so as to result in an acquisition of Units that would not exceed the limitations set forth in <u>Section 3.5</u>.

## (h) Conditions of Sale.

(i) Each Member participating in the Tag-along Sale shall receive the same consideration per Unit of each applicable type, class or series, as the case may be, after deduction of such Member's proportionate share of the related expenses in accordance with Section 10.5(j) below (it being understood that (A) the amount of consideration received by any particular Member shall be reduced, on a per-Unit basis, to account for any Tax Advances made to such Member which have not yet been repaid through reduction of a subsequent Distribution to such Member in the manner set forth in Section 7.1, and (B) in case of any such reduction of consideration, any such Tax Advance shall be deemed repaid and shall not reduce any subsequent Distributions made to such Member). If any such

Member is given an option as to the form and amount of consideration to be received, the same option shall be given to all such Members; <u>provided</u>, that in the event that the consideration to be received by the Selling Member is other than cash or marketable securities, each Tag-along Member may, in its respective sole discretion, elect to receive, in lieu of such other consideration, cash consideration equal to the fair market value of such consideration (as such fair market value is determined in good faith by the Board) and otherwise on the same terms and conditions as the Tag-along Members not receiving cash consideration. The proceeds of any Tag-along Sale which are paid to the Members and to which this <u>Section 10.5</u> applies will be allocated among the Members based upon the classes or series of Units included in the Tag-along Sale by each of the Members as if the proceeds of such Tag-along Sale were paid to the Members pursuant to <u>Section 13.3</u> of this Agreement in connection with a Distribution and the Units of the Members included in such Tag-along Sale were the only outstanding Units of the Company at the time of such Distribution.

- (ii) Each Tag-along Member shall make or provide the same representations, warranties, covenants, indemnities and agreements as the Selling Member makes or provides in connection with the Tag-along Sale; <u>provided</u> that:
  - (A) the Selling Member and each such Tag-along Member shall be obligated to make individual representations and warranties only with respect to its title to and ownership of the applicable Units, authorization, execution and delivery of relevant documents, enforceability of such documents against the Selling Member or such Tag-along Member, as applicable, and other matters relating to the Selling Member or such Tag-along Member, as applicable, but not with respect to any of the foregoing with respect to any other Members or their Units;
  - (B) all representations, warranties, covenants and indemnities shall be made by the Selling Member and each Tag-along Member, as applicable, severally and not jointly and any indemnification obligation shall be *pro rata* based on the consideration received by the Selling Member or each such Tag-along Member, as applicable, in each case in an amount not to exceed the aggregate proceeds received by the Selling Member or each such Tag-along Member, as applicable; and
  - (C) no Tag-along Member (or any Affiliate thereof) shall be required to enter into (1) any non-compete, non-solicitation or no-hire provision, (2) a provision providing for the licensing of intellectual property or the delivery of any products or services, or (3) any other provision that is not a customary financial term related directly to the sale of such Tag-along Member's Units pursuant to this Section 10.5.
- (iii) Each holder of then currently exercisable Unit Equivalents with respect to a type, class or series of Units proposed to be Transferred in a Tag-along Sale shall be given an opportunity to convert such Unit Equivalents into the applicable type, class

or series of Units prior to the consummation of the Tag-along Sale and participate in such sale as holders of such type, class or series of Units.

- (i) <u>Cooperation</u>. Each Tag-along Member shall take all actions as may be reasonably necessary to consummate the Tag-along Sale, including, without limitation, entering into agreements and delivering certificates and instruments, in each case, consistent with the agreements being entered into and the certificates being delivered by each Selling Member, but subject to <u>Section 10.5(h)(ii)</u>.
- (j) <u>Expenses</u>. The fees and expenses of the Selling Member incurred in connection with a Tag-along Sale and for the benefit of all Tag-along Members (it being understood that costs incurred by or on behalf of the Selling Member for its sole benefit will not be considered to be for the benefit of all Tag-along Members), to the extent not paid or reimbursed by the Company or the Proposed Transferee, shall be shared by the Selling Member and all the participating Tag-along Members on a *pro rata* basis, based on the consideration received by each such Member; <u>provided</u>, that no Tag-along Member shall be obligated to make any out-of-pocket expenditure prior to the consummation of the Tag-along Sale.
- (k) <u>Consummation of Sale</u>. The Selling Member shall have sixty (60) days following the expiration of the Tag-along Period in which to consummate the Tag-along Sale, on terms not more favorable to the Selling Member than those set forth in the Tag-along Notice (which such sixty (60)-day period may be extended for a reasonable time not to exceed ninety (90) days to the extent reasonably necessary to obtain required approvals or consents from any Governmental Authority). If at the end of such period the Selling Member had not completed the Tag-along Sale, the Selling Member may not then effect a Transfer that is subject to this <u>Section 10.5</u> without again fully complying with the provisions of this <u>Section 10.5</u>.
- (l) <u>Transfers in Violation of the Tag-along Right</u>. If the Selling Member sells or otherwise Transfers to the Proposed Transferee any of its Units in breach of this <u>Section 10.5</u>, then each Tag-along Member shall have the right to sell to the Selling Member, and the Selling Member undertakes to purchase from each Tag-along Member, the number of Units of each applicable type, class or series that such Tag-along Member would have had the right to sell to the Proposed Transferee pursuant to this <u>Section 10.5</u>, for a per Unit amount and form of consideration and upon the terms and conditions on which the Proposed Transferee bought such Units from the Selling Member, but without indemnity being granted by any Tag-along Member to the Selling Member; <u>provided</u>, that nothing contained in this <u>Section 10.5(1)</u> shall preclude any Tag-along Member from seeking alternative remedies against the Selling Member as a result of its breach of this <u>Section 10.5</u>. The Selling Member shall also reimburse each Tag-along Member for any and all reasonable and documented out-of-pocket fees and expenses, including reasonable legal fees and expenses, incurred pursuant to the exercise or the attempted exercise of the Tag-along Member's rights under this Section 10.5(1).
- (m) <u>Separate Series</u>. The Members acknowledge and agree that for the purposes of determining whether and to what extent each Member is permitted to participate in the Tag-along Sale, (i) Class A-1 Units, Class A-2 Units, Nonvoting Class A-1 Units and Nonvoting Class A-2

Units shall not be considered separate series, (ii) Class C-1 Units and Class C-2 Units shall not be considered separate series, and (iii) Class D-1 Units and Class D-2 Units shall not be considered separate series.

# 10.6 Regulatory Hardship Transfers; Surrender Right.

- (a) If (i) a Class A Member, a Class C Member or a Class D Member receives a directive from any Governmental Authority to divest any Units, (ii) a Class A Member, a Class C Member or a Class D Member determines in good faith and based on the advice of counsel (which may be in-house counsel) that (A) its continuing as a Member is legally impermissible, (B) it controls or could be presumed to control the Company for purposes of the BHCA, or the Home Owners' Loan Act of 1933, as amended, (C) as a result of a change in the business of the Company or any Company Subsidiary, it would be prohibited or materially restricted by Applicable Law from holding all or a portion of the Units, (D) its continuing as a Member would significantly and adversely affect its relationship with its applicable regulators or cause such Member significant reputational harm, or (E) the taking of any action by the Company or any Company Subsidiary would or could result in material legal consequences, material regulatory consequences or material reputational consequences to such Member or require such Member to file any application or notice for approval with its regulators or (iii) the SEC requires changes to the ownership or governance structure of any Regulated Securities Exchange Subsidiary or the Company as contemplated herein, and such changes materially and adversely affect the rights and benefits expected with respect thereto as of February 19, 2020 by a Class A Member, as of December 3, 2021 by a Class C Member, or as of March 13, 2023 by a Class D Member, as applicable (in the case of each of the foregoing clauses (i), (ii) and (iii), a "Regulatory Hardship Determination"), such Class A Member, Class C Member or Class D Member, as applicable, shall promptly (but no later than ten (10) Business Days following such determination) notify the Company of such Regulatory Hardship Determination.
- (b) Without limiting the foregoing, subject always to Sections 10.1 and 10.3, a Class A Member, a Class C Member or a Class D Member, as applicable, which has made a Regulatory Hardship Determination may Transfer all or a portion of its Units in accordance therewith; provided, that, in such event (i) Sections 10.1(a)(i), 10.1(a)(ii) and 10.5 shall not apply to such Transfer, and (ii) all of the time periods applicable pursuant to Section 10.3, other than Section 10.3(g), shall be reduced by half.
- (c) A Class A Member, a Class C Member or a Class D Member, as applicable, which has (i) made a Regulatory Hardship Determination or (ii) otherwise determined that continued ownership of Units by such Member would have a materially adverse impact on such Member as a result of applicable regulatory laws, may, at any time thereafter and acting in its sole discretion, voluntarily surrender to the Company any or all of its Units at any time, for consideration of one dollar (\$1.00), provided, that in such event Section 10.1(a)(i), 10.1(a)(ii), Section 10.3 and Section 10.5 and all other restrictions on transfer hereunder shall not apply.

10.7 <u>Transfers of Class B Units</u>. For the avoidance of doubt, without limiting any of the provisions set forth in this <u>Article X</u>, the Class B Units shall only be transferable if such Transfer is in compliance with the terms of the Incentive Plan.

# ARTICLE XI. ADDITIONAL AGREEMENTS OF THE PARTIES

# 11.1 <u>Confidentiality</u>.

- Each party hereto (for the sake of clarity, the Company and each Member) (a) acknowledges that during the term of this Agreement and in connection with this Agreement and the transactions contemplated hereby, it may have access to and become acquainted with trade secrets, proprietary information and confidential information belonging to one or more of the other parties hereto and their respective Affiliates that are not generally known to the public which the party disclosing such information (each such party, the "Disclosing Party") treats as confidential, in any format whatsoever (including oral, written, electronic or any other form or medium) (collectively, "Confidential Information"). In addition, each party to this Agreement receiving Confidential Information hereunder (each such party to this Agreement, the "Receiving Party") acknowledges that: (i) each Disclosing Party has invested, and continues to invest, substantial time, expense and specialized knowledge in developing its Confidential Information; (ii) the Confidential Information provides or may provide the applicable Disclosing Party with a competitive advantage over others in the marketplace; and (iii) the Disclosing Party may be irreparably harmed if the Confidential Information were disclosed. Without limiting the applicability of any other agreement to which any Receiving Party is subject, no Receiving Party shall, directly or indirectly through Affiliates or Representatives that receive Confidential Information hereunder, disclose or use (other than as permitted by any other agreement or document between the Receiving Party or the Disclosing Party or, in case of a Receiving Party that is a Member solely for the purposes of such Member monitoring and analyzing its investment in the Company or performing or having its Affiliates or Representatives (including Representatives of Affiliates) perform its or their respective duties as a Director, Alternate Director, Officer, employee, consultant or other service provider of the Company, as applicable) at any time, including, without limitation, use for personal, commercial or proprietary advantage or profit, either during the applicable Person's association or employment (as applicable) with the Disclosing Party or thereafter, any Confidential Information of which such Receiving Party is or becomes aware. Each Receiving Party in possession of Confidential Information shall take all commercially reasonable steps to safeguard such information and to protect it against disclosure, misuse, espionage, loss and theft.
- (b) Nothing contained in Section 11.1(a) shall prevent any Receiving Party from disclosing Confidential Information: (i) upon the order of any court or administrative agency; (ii) upon the request or demand of any Governmental Authority; (iii) to the extent compelled by legal process or required or requested pursuant to subpoena, interrogatories or other discovery requests; (iv) if the Receiving Party believes in good faith that disclosure of such Confidential Information to a Governmental Authority is necessary, advisable, or appropriate; (v) to the extent necessary in connection with the exercise of any remedy hereunder; (vi) if such Receiving Party is a Member and the Company is the Disclosing Party, to other Members; (vii) to such Receiving Party's

Representatives who, in the reasonable judgment of such Receiving Party, need to know such Confidential Information and are subject to confidentiality obligations or otherwise agree to be bound by the provisions of this Section 11.1 as if a Member or to otherwise keep Confidential Information confidential on terms no less protective of such information as set forth in this Section 11.1; or (viii) to any potential Permitted Transferee in connection with a proposed Transfer of Units from a Receiving Party that is a Member, as long as such Transferee agrees to be bound by the provisions of this Section 11.1 as if a Member; provided, that in the case of clause (i), (ii) or (iii), the applicable Receiving Party shall, if practicable and permitted by Applicable Law, and not in violation of an order of a court of competent jurisdiction or instructions of a prudential regulator, notify the Disclosing Party of the proposed disclosure as far in advance of such disclosure as practicable (but unless prohibited from doing so by any Applicable Law, not make any such disclosure before notifying the Disclosing Party) and use reasonable efforts to ensure that any Confidential Information so disclosed is accorded confidential treatment satisfactory to the Disclosing Party, when and if available.

- (c) The restrictions of Section 11.1(a) shall not apply to Confidential Information that: (i) is or becomes generally available to the public other than as a result of a disclosure by a Receiving Party in violation of this Agreement; (ii) is or becomes available to a Receiving Party on a non-confidential basis prior to its disclosure to the Receiving Party or any of its Representatives in compliance with this Agreement; (iii) is or has been independently developed or conceived by the applicable Receiving Party without use of Confidential Information; or (iv) becomes available to the Receiving Party or any of its Representatives on a non-confidential basis from a source other than the applicable Disclosing Party; provided, that such source is not known by the Receiving Party to be bound by a confidentiality agreement or other duty of confidentiality with the Disclosing Party or any of its Representatives.
- Unless the Receiving Party agrees to receive, and notwithstanding anything to the contrary herein, no party shall disclose Confidential Information constituting material nonpublic information relating to any issuer of public market securities (including any information derived from such material non-public information), that would require such Receiving Party or its Affiliates to abstain from trading in the securities of any such issuer, or information that would in any other manner restrict such Receiving Party or its Affiliates' trading or brokerage activities or their ability to make any particular trade. Each Receiving Party and its Affiliates disclaim any duty with regard to the foregoing. Each Disclosing Party acknowledges and agrees that each Receiving Party and its Affiliates are in the business (such current and future businesses, the "Trading and Investment Business") of trading, investing and dealing in financial instruments and/or providing brokerage services, both for themselves and on behalf of clients and/or customers, and are in other businesses related to the financial services industry. Each Disclosing Party acknowledges that each Receiving Party's Trading and Investment Business competes with such Disclosing Party's businesses, including for customers, accounts, employees, and trading opportunities, and each Disclosing Party acknowledges and agrees that each Receiving Party and its Affiliates are free to continue to compete and operate their Trading and Investment Business without regard to this Agreement or any obligations hereunder. In no event shall this Agreement be deemed or construed

as in any manner restricting the operation of the Trading and Investment Business by any Receiving Party or its Affiliates.

- (e) In no event shall this Agreement or any obligations hereunder be deemed or construed as restricting any Receiving Party's or its Affiliates' development and use of materials, documents, data, strategies, analysis, customer lists, and/or information similar to or competitive with any Confidential Information disclosed by a Disclosing Party hereunder. No Receiving Party nor any of its Affiliates shall have any liability to a Disclosing Party, and such Disclosing Party waives and releases any claims under this Agreement or the obligations hereunder that it might have against a Receiving Party or any of its Affiliates, with respect to its operation of the Trading and Investment Business of the Receiving Party and its Affiliates now or in the future.
- (f) The provisions of this <u>Section 11.1</u> shall survive the dissolution, liquidation, winding up and termination of the Company.

#### 11.2 Other Activities of Members.

- (a) Subject to the terms and conditions of any other written agreement to the contrary, any Member (other than a Member which initially became a Member through receiving Class B Units or other interests in the Company pursuant to an equity incentive plan (including the Incentive Plan)), any Person employed by, related to or in any way Affiliated with any such Member (excluding, for the avoidance of doubt, any Officer or employee of the Company or any Company Subsidiary), Director, Alternate Director, or Board Observer (other than the CEO) (the "Permitted Persons") may:
  - (i) have business interests and engage in business activities in addition to those relating to the Company or any Company Subsidiary; and
  - (ii) engage in any lawful business or trade, profession, employment or activity whatsoever, regardless of whether any such activity actually or potentially competes, directly or indirectly, with the business or activities of the Company or any Company Subsidiary in each case for its own account, or in partnership or participation with, or as an employee, officer, director, stockholder, member, manager, trustee, general or limited partner, agent or representative of, any other Person, and no Permitted Person shall be required to devote its entire time (business or otherwise), or any particular portion of its time (business or otherwise) to the business of the Company or any of Company Subsidiary. Without limiting the generality of the foregoing, but subject to any other written agreement to which a Permitted Person may be party, each Permitted Person may (A) engage in the same or similar activities or lines of business as the Company or any Company Subsidiary or develop or market any products or services that actually or potentially compete, directly or indirectly, with those of the Company or any Company Subsidiary, (B) invest or own any interest in, or develop a business relationship with, any Person engaged in the same or similar activities or lines of business as, or otherwise in competition with, the Company or any Company Subsidiary and (C) do business with any client or customer of the Company or any Company Subsidiary. Neither the Company, any Company Subsidiary, any Permitted

Person nor any Affiliate of any of the foregoing Persons, by virtue of this Agreement, shall have any rights in and to any such independent venture or the income or profits derived therefrom, regardless of whether or not such venture was initially presented to a Permitted Person as a direct or indirect result of its relationship with the Company or any Company Subsidiary.

- (b) No Permitted Person shall have any obligation hereunder to present any business opportunity to the Company or any Company Subsidiary, even if the opportunity is one that the Company or a Company Subsidiary might reasonably have pursued or had the ability or desire to pursue, in each case, if granted the opportunity to do so, and no Permitted Person shall be liable to the Company or any Company Subsidiary, any Member (or any Affiliate thereof) or any other Person for breach of any fiduciary or other duty relating to the Company or any Company Subsidiary (whether imposed by Applicable Law or otherwise), solely by reason of the fact that the Permitted Person pursues or acquires such business opportunity, directs such business opportunity to another Person or fails to present such business opportunity, or information regarding such business opportunity, to the Company or any Company Subsidiary.
- (c) Any repeal or modification of this <u>Section 11.2</u> shall not adversely affect any right or protection of any Permitted Person existing prior to such repeal or modification.
- (d) Subject to the terms of any other written agreement by any Member to the contrary, neither the Company, any Company Subsidiary, nor any other Member shall have any rights by virtue of this Agreement in any business interests or activities of any Member.
- (e) Nothing in this <u>Section 11.2</u> shall limit or otherwise prejudice any contractual rights the Company or any Company Subsidiary may have or obtain against any Person, or any director, officer or employee of any such Person.

### 11.3 Compliance with Law.

- (a) The Company shall implement and maintain policies and procedures reasonably designed to ensure that the Company and the Company Subsidiaries comply with all Applicable Laws relating to international trade, economic sanctions, and export controls including but not limited to those administered by OFAC at 31 C.F.R. Part 500 to End, and the U.S. Export Administration Regulations at 15 C.F.R. Part 730 et. seq. and any other requirement applicable for any import, export controls and economic sanctions laws and regulations.
- (b) The Company shall implement and maintain policies and procedures reasonably designed to prevent violations by the Company and the Company Subsidiaries of the FCPA and other applicable anti-bribery or anti-corruption laws, including but not limited to periodic training, to occur at the Company's discretion, of the Company's and the Company Subsidiaries' independent contractors, consultants and other agents.
- (c) The Company shall implement and maintain policies and procedures reasonably designed to ensure compliance by the Company and the Company Subsidiaries with all

Applicable Laws relating to the prevention of money laundering or terrorist financing of any Governmental Authority applicable to the Company or any Subsidiary of the Company or its property or in respect of its operations, including all applicable financial record-keeping, antimoney laundering program, customer identification program and suspicious activity reporting requirements of the Bank Secrecy Act as amended by the USA PATRIOT Act, and other antimoney laundering regulations at such time they become applicable to the Company or any Company Subsidiary.

- (d) The Company shall implement and maintain policies and procedures reasonably designed to ensure compliance by the Company and the Company Subsidiaries with all Applicable Laws relating to disclosure of material non-public information (i) regarding any Person (whether or not the securities of such Person are publicly traded) with outstanding securities or whose securities are in the process of being offered for sale, or (ii) which the Company or any Company Subsidiary knows or reasonably should know the disclosure of which is prohibited or restricted.
- (e) The Company shall, and shall cause the Company Subsidiaries to, ensure that no such entity takes any action, refrains from taking any action or enters into any transaction where such action, failure to act or transaction would (i) violate any Applicable Law (including, without limitation, applicable data protection or privacy laws) or (ii) reasonably be expected to result in adverse legal, compliance or regulatory consequences for the Company or any Company Subsidiary or any Member. Without limiting the generality of the foregoing, the Company shall, and shall cause all of the Company Subsidiaries to, ensure that they make all material required reports, notifications and disclosures to all applicable Governmental Authorities.
- (f) The Company shall implement and maintain policies and procedures reasonably designed to ensure compliance by the Company and the Company Subsidiaries with all Applicable Laws relating to data protection and data privacy.
- (g) The Company shall, and shall cause all of the other Company Subsidiaries to, comply in good faith in all material respects with all Applicable Laws relating to taxes, including filing, on a timely basis, all returns, declarations, forms, statements, reports, schedules, information returns or similar statements or documents, and any amendments thereof (including any related or supporting information or schedule attached thereto) required to be filed (including electronically) with any Governmental Authority in connection with the determination, assessment or collection of (i) any federal, state, local, provincial or foreign income or franchise taxes, including any interest, penalty, or addition thereto, and (ii) any sales, use, ad valorem, transfer, license, recording, employment (including federal and state income tax withholding, backup withholding, FICA, FUTA or other payroll taxes), environmental, excise, severance, stamp, occupation, premium, prohibited transaction, property, value added, net worth, gross receipts, or any other federal, state, local, provincial or foreign taxes, customs, tariffs, imposts, levies, duties, government fees or other like assessments or charges of any kind, and shall ensure that such tax returns are complete and accurate in all material respects.

- The Company shall comply with the federal securities laws and the rules and regulations promulgated thereunder and shall cooperate with the SEC, each Regulated Securities Exchange Subsidiary, FINRA and any other SROs of which MEMX Execution Services LLC, a Subsidiary of the Company that is registered with FINRA as a broker-dealer, is a member, pursuant to and to the extent of their respective regulatory authority. The Directors, Officers, employees and agents of the Company, by virtue of their acceptance of their respective positions, agree to comply, and shall comply, with the federal securities laws and the rules and regulations promulgated thereunder and shall be deemed to agree to cooperate with (i) the SEC and each Regulated Securities Exchange Subsidiary in respect of the SEC's oversight responsibilities regarding such Regulated Securities Exchange Subsidiary and the self-regulatory functions and responsibilities of such Regulated Securities Exchange Subsidiary, and (ii) FINRA, any other SROs of which MEMX Execution Services LLC is a member, and MEMX Execution Services LLC in respect of FINRA's and any such other SRO's oversight responsibilities regarding MEMX Execution Services LLC, as applicable, and the Company shall take reasonable steps necessary to cause its Directors, Officers, employees and agents to so cooperate. No present or past member, employee, beneficiary, agent, customer, creditor, regulatory authority (or member thereof) or other Person (other than each Regulated Securities Exchange Subsidiary and MEMX Execution Services LLC, which are intended third-party beneficiaries of this Section 11.3(h)) shall have any rights against the Company or any Director, Officer, employee or agent of the Company under this Section 11.3(h).
- (i) The Company hereby represents and warrants to each Bank Member that the Company is not a covered fund (as such term is defined in 12 C.F.R. § 248.10(b)), and not a bank, bank holding company, depository institution or holding company for a depository institution, as such terms are defined in the BHCA. The Company shall not allow itself to become a covered fund, bank, bank holding company, depository institution or holding company for a depository institution (as so defined).
- (j) No Member which initially became a Member as a result of receipt of Class B Units or other equity interests in the Company pursuant to an equity incentive plan (including the Incentive Plan) shall have any right to enforce the provisions, or to recover damages in the event of a breach of any provision, of this <u>Section 11.3</u>.
- (k) All policies referred to in this <u>Section 11.3</u> shall be developed by the Company in consultation with appropriate legal counsel.

## 11.4 Antitrust Practices and Guidelines.

- (a) The Company shall retain respected, regular outside antitrust and competition counsel ("<u>Antitrust Counsel</u>") to advise the Company and the Company Subsidiaries as necessary with respect to antitrust and competition law considerations with respect to the business and operations thereof, including as set forth in Exhibit D.
- (b) Attached as <u>Exhibit D</u> hereto are antitrust guidelines which shall be followed at all times by the Company, each Company Subsidiary, the Members, the Board, each committee of the Board, the board of directors or similar governing body of any Company Subsidiary and any

committee of such a board of directors or similar governing body. The Company shall ensure that the Company Subsidiaries follow such guidelines.

## 11.5 <u>Initial Public Offering</u>

- Initial Public Offering. If at any time the Board desires to cause (i) a Transfer of all or a substantial portion of (x) the assets of the Company or (y) the Units to a newly organized corporation or other business entity (an "IPO Entity"), (ii) a merger or consolidation of the Company into or with a IPO Entity as provided under § 18-209 of the Delaware Act or otherwise, or (iii) another restructuring of all or substantially all the assets or Units of the Company into an IPO Entity, including by way of the conversion of the Company into a Delaware corporation as provided under § 18-216 of the Delaware Act (any such corporation also herein referred to as an "IPO Entity"), in any such case in anticipation of or otherwise in connection with an Initial Public Offering of securities of an IPO Entity or its Affiliate (an "Initial Public Offering"), each Member shall take such steps to effect such Transfer, merger, consolidation, conversion or other restructuring as may be reasonably requested by the Board, including, without limitation, executing and delivering all agreements, instruments and documents as may be reasonably required and Transferring or tendering such Member's Units to an IPO Entity in exchange or in consideration for shares of capital stock or other equity interests of the IPO Entity, determined in accordance with the valuation procedures set forth in Section 11.5(b). The members of the Board and the Company will cooperate in good faith to structure any such Initial Public Offering in a tax-efficient manner.
- (b) Fair Market Value. In connection with a transaction described in Section 11.5(a), the Board shall, in good faith but subject to the following sentence, determine the Fair Market Value of the assets and/or Units Transferred to, merged with or converted into shares of the IPO Entity, the aggregate Fair Market Value of the IPO Entity and the number of shares of capital stock or other equity interests to be issued to each Member in exchange or in consideration therefor. In determining Fair Market Value, (i) the offering price of the Initial Public Offering shall be used by the Board to determine the Fair Market Value of the capital stock or other equity interests of the IPO Entity and (ii) the Distributions that the Members would have received with respect to their Units if the Company were dissolved, its affairs wound up and Distributions made to the Members in accordance with Section 13.3(c) shall determine the Fair Market Value of the Units. In addition, any Units to be converted into or redeemed or exchanged for shares of the IPO Entity shall receive shares of voting common stock with substantially equivalent economic, priority and other rights and privileges as in effect immediately prior to such transaction (disregarding the tax treatment of such transaction).
- (c) Lock-up Agreement. Each Member hereby agrees that in connection with an Initial Public Offering, and upon the request of the managing underwriter in such offering, such Member shall not, without the prior written consent of such managing underwriter, during the period commencing 180 days prior to the effective date of such registration and ending on the date specified by such managing underwriter (such period not to exceed 180 days), (i) offer, pledge, sell, contract to sell, grant any option or contract to purchase, purchase any option or contract to sell, hedge the beneficial ownership of or otherwise dispose of, directly or indirectly, any Units or Unit Equivalents (including any equity securities of the IPO Entity), or (ii) enter into any swap or other

arrangement that transfers to another, in whole or in part, any of the economic consequences of ownership of such securities, whether any such transaction described in clause (i) or (ii) above is to be settled by delivery of Units or Unit Equivalents (including equity securities of the IPO Entity) or such other securities, in cash or otherwise. The foregoing provisions of this Section 11.5(c) shall not apply to sales of securities to be included in such Initial Public Offering or other offering if otherwise permitted, and shall be applicable to the Members only if all Directors and Officers of the Company and all Members owning more than 1% of the Company's outstanding Units (or the IPO Entity's equivalent common equity securities) are subject to the same restrictions. Each Member agrees to execute and deliver such other agreements as may be reasonably requested by the Company or the managing underwriter which are consistent with the foregoing or which are necessary to give further effect thereto. Notwithstanding anything to the contrary contained in this Section 11.5(c), each Member shall be released, pro rata, from any lock-up agreement entered into pursuant to this Section 11.5(c) in the event and to the extent that the managing underwriter or the Company permit any discretionary waiver or termination of the restrictions of any lock-up agreement pertaining to any Director, Officer or holder of greater than one percent (1%) of the Company's outstanding Units (or the IPO Entity's equivalent common equity securities). Notwithstanding the foregoing and for the avoidance of doubt, it is understood and agreed that nothing contained herein shall in any way limit a Member or any Affiliate of a Member from engaging in any brokerage, investment advisory, financial advisory, anti-raid advisory, principaling, merger and advisory, financing, asset management, trading, market making, arbitrage, investment activity, or similar activities conducted in the ordinary course of business of such Member or its Affiliates.

- Offering, the Company shall enter, or shall ensure that the IPO Entity enters, as applicable, into a registration rights agreement, upon commercially reasonable terms, with any Member requesting such agreement with respect to the registration of its Units (or the capital stock into which Units are converted) with customary terms and conditions and in form and substance approved by a the Board; provided that such registration rights agreement shall include (i) customary demand registration rights (including at least three (3) demand registrations), that (A) otherwise apply equally to all Members, (B) apply only after an Initial Public Offering, and (C) are subject to customary minimum thresholds, (ii) customary piggyback registration rights available to all Members on a pro rata basis in accordance with their *pro rata* ownership in the Company or the IPO Entity, as applicable and (iii) an obligation of the Company to pay for all registration expenses, which shall include the fees and expenses of one special counsel for the participating Members.
- (e) The provisions of this <u>Section 11.5</u> constitute an individual agreement between the Company, on the one hand, and each applicable Member, on the other hand, and do not constitute an agreement among the Members. Accordingly, only the Company shall have the right to enforce this <u>Section 11.5</u> against any Member, and no Member shall have the right to enforce this <u>Section 11.5</u> against any other Member.

#### 11.6 Use of Name and Trade Marks.

- (a) Except to the extent required under Applicable Law or unless permitted by the terms of a separate agreement between a Member or an Affiliate of a Member and the Company, no Member shall, without the prior written consent of the Company (which consent may be given or withheld in the Company's sole discretion), (i) use in advertising, publicity or otherwise (including press releases) the name of the Company or any Company Subsidiary or their respective employees, or any Proprietary Identifier owned or used by the Company or any Company Subsidiary, or (ii) represent, directly or indirectly, that any product or any service provided by such Member has been approved, endorsed, recommended or provided by, or in association with, the Company or any Company Subsidiary.
- Except to the extent the Company is permitted to do so pursuant to an (b) agreement between the Company and the Member in question relating to Company's use of such Member's Proprietary Identifiers, the Company shall not, without the prior written consent of such Member for each instance (which consent may be given or withheld in such Member's sole discretion), (i) use in advertising, publicity or otherwise (including press releases) the name of any Member or its Affiliates (other than, if applicable, the Company and any Company Subsidiary) or employees, or any Proprietary Identifier owned or used by a Member or its Affiliates (other than, if applicable, the Company and any Company Subsidiary), (ii) represent, directly or indirectly, that any product or any service provided by the Company or any of Company Subsidiary has been approved, endorsed, recommended or provided by, or in association with, any Member or its Affiliates (other than, if applicable, the Company and any Company Subsidiary) or (iii) disclose to any non-Member that such Member has any interest in the Company or any of its Subsidiaries except as required by Applicable Law or if such information becomes publicly available; provided, that if the Company or any Company Subsidiary is required under Applicable Law to disclose any information about a Member or any of its Affiliates in connection with any regulatory filing or regulatory application of the Company or any Company Subsidiary that is permitted under this Agreement the Company shall provide, or shall ensure that the applicable Company Subsidiary provides, such Member (A) a copy of such proposed disclosure as soon as reasonably practicable, but in no event less than fifteen (15) days prior to making such filing or application, and (B) such other information reasonably requested by such Member in connection with such filing or application, in each case to the extent permitted by Applicable Law.
- c) Except as required by Applicable Law or if such information becomes publicly available, no Member (the "<u>Using Member</u>") shall, without the prior written consent of the applicable other Member in each instance (which consent may be given or withheld in such other Member's sole discretion), disclose to any non-Member that such other Member (other than an Affiliate of such disclosing Member) has any interest in the Company or any of its Subsidiaries, or is a party to this Agreement or any ancillary agreements contemplated hereby; <u>provided</u>, that a Using Member or any of its Affiliates may provide such information if requested by a regulator having jurisdiction over such Using Member or any of its Affiliates and in such event such Using Member shall provide, or shall ensure that the applicable Affiliate thereof provides, such other Member prior notice, as soon as reasonably practicable, of such disclosure to the extent permitted by Applicable Law.

11.7 <u>Certain Regulatory Matters</u>. The Members and the Company agree that if the exercise by a Member (such Member, the "<u>Acting Member</u>") of its rights hereunder (including pursuant to <u>Sections 3.9, 3.10, 8.11</u> or <u>10.6</u>) is reasonably likely to result in adverse legal or regulatory consequences for another Member, including causing any such other Member to (a) hold an ownership interest in the Company greater than the thresholds prescribed in <u>Section 3.6</u>, (b) control the Company for purposes of the BHCA, or the Home Owners' Loan Act of 1933, as amended, or (c) become an affiliated person, within the meaning of Section 2(a)(3) of the Investment Company Act of 1940, as amended, of any Regulated Securities Exchange Subsidiary, the Company shall work with any affected Member and shall use commercially reasonable efforts to eliminate or minimize such adverse legal or regulatory consequences.

# ARTICLE XII. INFORMATION RIGHTS: ACCOUNTING; TAX MATTERS

- 12.1 <u>Information Rights</u>. Subject to <u>Exhibit D</u>, as long as (x) a Member (other than a holder of Class B Units or other equity interests in the Company received pursuant to an equity incentive plan (including the Incentive Plan)) holds an aggregate number of Class A Units, Class C Units, Class D Units and Converted Common Units equal to at least 2,500,000 (subject to adjustment in the event of any Unit split, Unit combination, reorganization, reclassification, recapitalization or the like), or (y) notwithstanding the provisions of clause (x), is subject to or has an Affiliate subject to, the BHCA or the Home Owners' Loan Act of 1933, as amended (each Member to which clause (x) or clause (y) above applies, a "Qualified Member"), the Company shall furnish to such Qualified Member the following:
- (a) within thirty (30) calendar days after the close of each calendar month (other than the last calendar month of a fiscal quarter), a management report for such month in such form and with such substance as determined by the CEO but including, at a minimum an unaudited consolidated balance sheet of the Company and the Company Subsidiaries as of the end of such month, together with related statements of operations, income and cash flow for such month (and the current year to date);
- (b) within thirty (30) calendar days after the close of each fiscal quarter (other than the fourth quarter), an unaudited consolidated balance sheet of the Company and the Company Subsidiaries as of the end of such quarter, together with related statements of operations, income and cash flow for such quarter (and the current year to date) and a management report for such quarter (and the current year to date) in such form and with such substance as determined by the CEO together with a statement showing all amounts credited and debited to each Member's Capital Account and each Member's distributive share, for federal income tax purposes, of income, gains, deductions, losses and credits (or items thereof) arising out of Company operations, as required by Applicable Law;
- (c) as soon as available, and in any event within one hundred twenty (120) days after the end of each Fiscal Year, audited consolidated balance sheets of the Company and the Company Subsidiaries as at the end of each such Fiscal Year and audited consolidated statements of income, cash flows and Members' equity for such Fiscal Year, in each case setting forth in

comparative form the figures for the previous Fiscal Year, accompanied by the certification of independent certified public accountants of recognized national standing selected by the Board, certifying to the effect that, except as set forth therein, such financial statements have been prepared in accordance with GAAP, applied on a basis consistent with prior years, and fairly present in all material respects the financial condition of the Company and Company Subsidiaries as of the dates thereof and the results of their operations and changes in their cash flows and Members' equity for the periods covered thereby;

- (d) any report, presentation, management-prepared update, or other information of general applicability to all Members that is provided to any Member;
- (e) prompt reports of the initiation or settlement of, or material developments in, any action or proceeding involving the Company or any Company Subsidiary;
- (f) the Company's Annual Budget promptly, and in any event within ten (10) calendar days after such Annual Budget is approved by the Board by Supermajority Board Vote;
- to the extent permitted by Applicable Law, notification (which shall be delivered reasonably promptly after an Officer receives written notice or otherwise has actual knowledge of such matter) of (i) any material violation or breach of any Applicable Law or internal compliance policy by the Company or any Company Subsidiary, (ii) any criminal or regulatory investigation or proceeding against the Company or any Company Subsidiary, (iii) any event or occurrence with respect to the Company or any Company Subsidiary that would, or would reasonably be expected to, result in a material violation or breach of any Applicable Law or internal compliance policy or require reporting to a Governmental Authority, or (iv) any other event or occurrence with respect to Company or any Company Subsidiary that would or could reasonably be expected to result in adverse legal or regulatory consequences for any Member or any of its respective Affiliates. Notwithstanding the foregoing, the Qualified Members shall have no right to receive pursuant to this Section 12.1(g) any (A) information, material, data or documents the provision of which to such Qualified Member would, in the reasonable judgment of the Board, result in a waiver of any applicable legal privilege or that is not permitted to be disclosed under Applicable Law or (B) trade secrets of the Company or any Company Subsidiary or any similar materials:
- (h) without limiting the generality of clause (g) above, prompt notice of any event or occurrence with respect to the Company or a Company Subsidiary that would or could reasonably be expected to result in adverse legal or regulatory consequences for such Qualified Member or any of its Affiliates, including any action that could reasonably be expected to result in a violation of any regulation or statute administered by OFAC or of the FCPA or any other applicable anti-bribery or anti-corruption laws; and
- (i) with reasonable promptness, such other information and data as a Qualified Member may from time to time reasonably request; <u>provided</u> that Qualified Members shall have no right to receive pursuant to this <u>Section 12.1(i)</u> any (i) information, material, data or documents the provision of which to such Qualified Member would, in the reasonable judgment of the Board,

result in a waiver of any applicable legal privilege or that is not permitted to be disclosed under Applicable Law (including antitrust and competition laws), (ii) trade secrets of the Company or any Company Subsidiary or any similar materials or (iii) any information which the Company is not permitted to provide to a Qualified Member pursuant to this Agreement or under Applicable Law.

### 12.2 <u>Inspection Rights; Books and Records.</u>

- (a) All corporate, financial and similar records, reports and documents of the Company, including, without limitation, all financial statements, books and records and minutes of proceedings shall be maintained at a location within the United States. Upon reasonable notice from a Qualified Member, the Company shall, and shall cause its Directors, Alternate Directors, Officers and employees to, afford each Qualified Member and its Representatives reasonable access during normal business hours to the corporate, financial and similar records, reports and documents of the Company, including, without limitation, all financial statements, books and records and minutes of proceedings, and to permit each Qualified Member and its Representatives to examine such documents and make copies thereof. The foregoing right of inspection shall be subject to (a) any limitations or restrictions reasonably determined by the Board from time to time and (b) the limitations set forth in Section 12.2(c).
- To the extent Company's corporate, financial and similar records, reports and documents, including all financial statements, books and records and minutes of proceedings, are related to the activities of any Regulated Securities Exchange Subsidiary, such corporate, financial and similar records, reports and documents, including all financial statements, books and records and minutes of proceedings, as well as premises, Directors, Alternate Directors, Officers, employees and agents of the Company shall be deemed to be the corporate, financial and similar records, reports and documents, including all financial statements, books and records and minutes of proceedings, as well as premises, directors, officers, employees or agents, as applicable, of such Regulated Securities Exchange Subsidiary for the purposes of, and subject to oversight pursuant to, the Exchange Act. For so long as the Company shall Control, directly or indirectly, any Regulated Securities Exchange Subsidiary or MEMX Execution Services LLC, the Company's corporate, financial and similar records, reports and documents, including all financial statements, books and records and minutes of proceedings shall be subject at all times to inspection and copying by the SEC and each such Regulated Securities Exchange Subsidiary (and to the extent such records and documents relate to the activities of MEMX Execution Services LLC, FINRA, any other SROs of which MEMX Execution Services LLC is a member, and MEMX Execution Services LLC); provided that such corporate, financial and similar records, reports and documents, including all financial statements, books and records and minutes of proceedings are related to the operation or administration of such Regulated Securities Exchange Subsidiary or MEMX Execution Services LLC, as applicable.
- (c) All books and records of a Regulated Securities Exchange Subsidiary reflecting confidential information pertaining to the self-regulatory function of such Regulated Securities Exchange Subsidiary (including but not limited to disciplinary matters, trading data, trading practices and audit information) that shall come into the possession of the Company, and the information contained in those books and records, shall be retained in confidence by the Company

and the Directors, Alternate Directors, Officers, employees and agents of the Company and shall not be used for any non-regulatory purposes. Notwithstanding the foregoing sentence, nothing in this Agreement shall be interpreted so as to limit or impede the rights of the SEC or each Regulated Securities Exchange Subsidiary to access and examine such confidential information pursuant to the U.S. federal securities laws and the rules and regulations thereunder, or to limit or impede the ability of any Directors, Alternate Directors, Officers, employees or agents of the Company to disclose such confidential information to the SEC or such Regulated Securities Exchange Subsidiary.

### 12.3 Partnership Representative.

- (a) For purposes of this <u>Section 12.3</u>, unless otherwise specified, all references to provisions of the Code shall be to such provisions as enacted by the Bipartisan Budget Act of 2015 as such provisions may be modified from time to time (the "<u>BBA</u>").
- (b) The Board shall identify a Person to act as the Company's designated "partnership representative" within the meaning of Code Section 6223 (the "Partnership Representative"). The Partnership Representative and any "designated individual" within the meaning of Treasury Regulations Section 301.6223-1(b)(3)(ii) shall have sole authority to act on behalf of the Company for purposes of Subchapter C of Chapter 63 of the Code and any comparable provisions of state or local income tax laws (the "Partnership Audit Rules"), subject to advice by the Company's tax advisors and approval by the Board of any material elections or decisions.
- (c) If the Company qualifies to elect pursuant to Code Section 6221(b) (or successor provision) to have Subchapter C of Chapter 63 of the Code not apply to any federal income tax audits and other proceedings, the Partnership Representative shall, upon the instructions of the Board, cause the Company to make such election.
- (d) If any "partnership adjustment" (as defined in Code Section 6241(2)) is determined with respect to the Company, the Partnership Representative shall promptly notify the Board upon the receipt of a notice of final partnership adjustment, and shall take such actions as directed by the Board, including whether to file a petition in tax court, cause the Company to pay the amount of any such adjustment under Code Section 6225, or make the election under Code Section 6226 or take any other action authorized by the Partnership Audit Rules. The Partnership Representative shall promptly notify each Member of any audit or contest relating to a tax return (or other tax matter) of the Company or any of its Subsidiaries and shall use commercially reasonable efforts to keep each Member reasonably informed with respect to such audit or contest.
- (e) If any "partnership adjustment" (as defined in Code Section 6241(2)) is finally determined with respect to the Company and the Partnership Representative has not caused the Company to make the election under Code Section 6226, then (i) the Members shall take such actions requested by the Partnership Representative; (ii) the Partnership Representative shall use commercially reasonable efforts to make any modifications available under Code Section 6225(c)(3), (4) and (5); and (iii) any "imputed underpayment" (as determined in accordance with

Code Section 6225) or partnership adjustment that does not give rise to an imputed underpayment, shall be apportioned among the Members of the Company for the taxable year in which the adjustment is finalized in such manner as may be necessary (as determined by the Partnership Representative in good faith based on the advice of legal or tax counsel to the Company and subject to approval of the Board) so that, to the maximum extent possible, the tax and economic consequences of the partnership adjustment and any associated interest and penalties are borne by the Members based upon their interests in the Company for the reviewed year. Notwithstanding the foregoing, the Partnership Representative and the Company shall not (1) take any action that would cause a Member to be required to file amended tax returns in accordance with Code Section 6225(c)(2)(A) (or any similar provisions under state or local law), without the advance written consent of each such Member or (2) elect the alternative "pull-in" procedure in accordance with Section 6225(c)(2)(B) of the Code (or any similar provisions under state or local law) in accordance with Code Section 6225(c)(2)(B), without the advance written consent of the Members holding at least a majority of the outstanding Units.

- (f) If any Company Subsidiary (i) pays any partnership adjustment under Code Section 6225, (ii) requires the Company to file an amended tax return and pay associated taxes to reduce the amount of a partnership adjustment imposed on such Company Subsidiary, or (iii) makes an election under Code Section 6226, the Partnership Representative shall cause the Company to make the administrative adjustment request provided for in Code Section 6227 consistent with the principles and limitations set forth in Sections 12.13(d)-(e) above for partnership adjustments of the Company, and the Members shall take such actions reasonably requested by the Partnership Representative in furtherance of such administrative adjustment request.
- (g) The obligations of each Member or former Member under this <u>Section 12.3</u> shall survive the transfer or redemption by such Member of its Units and the termination of this Agreement or the dissolution of the Company.
- (h) For the avoidance of any doubt, in the event of a tax audit of a Member relating to income or credits derived from the Company, the Company shall use commercially reasonable efforts to cooperate with such Member in good faith (at the cost and expense of the Member).

### 12.4 Annual Budget.

(a) Prior to the start of any Fiscal Year, the Company shall prepare and deliver to the Board an annual budget setting forth all reasonably anticipated expenses of the Company and the Company Subsidiaries on a consolidated basis during the course of the upcoming Fiscal Year (the "Annual Budget"). If the Board, by Supermajority Board Vote, does not approve a proposed Annual Budget in its entirety prior to the start of a Fiscal Year, (i) those line items of the proposed Annual Budget that have been approved by the Board, if any, shall constitute the Annual Budget with respect to the matters set forth therein and (ii) with respect to those line items of the Annual Budget that have not been approved by the Board, the Annual Budget from the prior Fiscal Year with respect to those line items shall remain in effect until such line items are approved by the Board.

- (b) The Company and the Board shall allocate sufficient funds in each Annual Budget or variance thereto (as reasonably determined by the Board in good faith) in order to permit the Company and the Company Subsidiaries to adequately comply with their legal, regulatory and compliance obligations during such Fiscal Year under Applicable Law and as set forth in this Agreement.
- (c) The Company shall, and shall cause the other Company Subsidiaries to, be managed in accordance with the Annual Budget (as in effect from time to time), and not take actions that are not consistent with the Annual Budget (as in effect from time to time) except as may be approved by the Board by the applicable vote required hereunder for such action. Notwithstanding the foregoing, until February 19, 2023 no approval of the Board shall be required for variances in the aggregate amount of the expenditures set forth in the Annual Budget of less than fifteen percent (15%); provided, that the CEO promptly notifies the Board of any such expenditures that constitute such a variance. Upon the expiration of such period, the Board shall determine, by Supermajority Board Vote, what level of discretion the CEO shall have with respect to variances from the Annual Budget.
- Tax Returns. At the expense of the Company, the Board (or any Officer that it may designate pursuant to Section 8.14) shall endeavor to cause the preparation and timely filing (including extensions) of all tax returns required to be filed by the Company pursuant to the Code as well as all other required tax returns in each jurisdiction in which the Company and the Company Subsidiaries own property or do business. As soon as reasonably possible after the end of each Fiscal Year, the Board or designated Officer shall cause to be delivered to each Person who was a Member at any time during such Fiscal Year, IRS Schedule K-1 to Form 1065 and such other information with respect to the Company as may be necessary for the preparation of such Person's federal, state and local and foreign income tax returns for such Fiscal Year or the payment of such Person's taxes (including estimated taxes or a quarterly or other basis); provided, however, that if the Company is unable to provide such IRS Schedule K-1 to Form 1065 and such other information within ninety (90) days after the end of a Fiscal Year, the Company shall, within such time period, provide the Members with estimated information sufficient to permit the Members to file his, her or its U.S. federal and state income tax returns for such taxable year on an estimated basis, and thereafter the Company shall provide the Members with a final IRS Schedule K-1 to Form 1065 and such other information for such Fiscal Year as soon as such final IRS Schedule K-1 to Form 1065 and such other information is able to be prepared (provided that in no event shall such final IRS Schedule K-1 to Form 1065 and such other information be provided more than one hundred twenty (120) days after the end of a Fiscal Year).
- 12.6 <u>Company Funds</u>. All funds of the Company shall be deposited in its name, or in such name as may be designated by the Board, in such checking, savings or other accounts, or held in its name in the form of such other investments as shall be designated by the Board. The funds of the Company shall not be commingled with the funds of any other Person. All withdrawals of such deposits or liquidations of such investments by the Company shall be made exclusively upon the signature or signatures of such Officer or Officers as the Board may designate.

# ARTICLE XIII. DISSOLUTION AND LIQUIDATION

- 13.1 <u>Events of Dissolution</u>. The Company shall be dissolved and its affairs wound up only upon the occurrence of any of the following events:
- (a) By determination of (i) the Board by Supermajority Board Vote and (ii) the applicable Members pursuant to Section 4.7(i).
- (b) The sale, exchange, involuntary conversion, or other disposition or Transfer of all or substantially all the assets of the Company; or
- (c) The entry of a decree of judicial dissolution under § 18-802 of the Delaware Act.
- 13.2 <u>Effectiveness of Dissolution</u>. Dissolution of the Company shall be effective on the day on which the event described in <u>Section 13.1</u> occurs, but the Company shall not terminate until the winding up of the Company has been completed, the assets of the Company have been distributed as provided in <u>Section 13.3</u> and the Certificate of Formation shall have been cancelled as provided in <u>Section 13.4</u>.
- 13.3 <u>Liquidation</u>. If the Company is dissolved pursuant to <u>Section 13.1</u>, the Company shall be liquidated and its business and affairs wound up in accordance with the Delaware Act and the following provisions:
- (a) <u>Liquidator</u>. The Board, or, if the Board is unable to do so, a Person selected by the holders of a majority of the Units, shall act as liquidator to wind up the Company (the "<u>Liquidator</u>"). The Liquidator shall have full power and authority to sell, assign, and encumber any or all of the Company's assets and to wind up and liquidate the affairs of the Company in an orderly and business-like manner.
- (b) <u>Accounting</u>. As promptly as possible after dissolution and again after final liquidation, the Liquidator shall cause a proper accounting to be made by a recognized firm of certified public accountants of the Company's assets, liabilities and operations through the last day of the calendar month in which the dissolution occurs or the final liquidation is completed, as applicable.
- (c) <u>Distribution of Proceeds</u>. The Liquidator shall liquidate the assets of the Company and distribute the proceeds of such liquidation (the "<u>Liquidation Proceeds</u>") in the following order of priority, unless otherwise required by mandatory provisions of Applicable Law:
  - (i) <u>First</u>, to the payment of all of the Company's debts and liabilities to its creditors (including Members, if applicable) and the expenses of liquidation (including sales commissions incident to any sales of assets of the Company);

(ii) <u>Second</u>, to the establishment of and additions to reserves that are determined by the Board in its sole discretion to be reasonably necessary for any contingent unforeseen liabilities or obligations of the Company;

### (iii) <u>Third</u>:

- (A) with respect to each outstanding Class C Unit, an amount equal to the greater of (i) unreturned Capital Contributions with respect to such Class C Unit; provided that the amount of the unreturned Capital Contributions with respect to each outstanding Class C Unit shall be reduced (but not below zero) by the amount of any previous Distributions made with respect to such Class C Unit and (ii) the amount that such Class C Unit would receive pursuant to Section 13.3(c)(v) if it were converted to a Common Unit immediately prior to the applicable distribution (for the avoidance of doubt, regardless of whether such Class C Unit has converted to a Common Unit and regardless of whether the holder of such Class C Unit has given prior notice to the Company of its election to convert such Class C Unit into a Common Unit); provided further that such amounts will be distributed to the Class C Members pro rata in accordance with their unreturned Capital Contributions; and
- (B) with respect to each outstanding Class D Unit, an amount equal to the greater of (i) unreturned Capital Contributions with respect to such Class D Unit; provided that the amount of the unreturned Capital Contributions with respect to each outstanding Class D Unit shall be reduced (but not below zero) by the amount of any previous Distributions made with respect to such Class D Unit and (ii) the amount that such Class D Unit would receive pursuant to Section 13.3(c)(v) if it were converted to a Common Unit immediately prior to the applicable distribution (for the avoidance of doubt, regardless of whether such Class D Unit has converted to a Common Unit and regardless of whether the holder of such Class D Unit has given prior notice to the Company of its election to convert such Class D Unit into a Common Unit); provided further that such amounts will be distributed to the Class D Members pro rata in accordance with their unreturned Capital Contributions;

<u>provided</u>, that if assets of the Company available for distribution at any applicable time are insufficient to distribute to the Class C Members and Class D Members the full amounts to which the Class C Members and the Class D Members would be entitled pursuant to clauses (A) and (B) immediately above, as applicable, then the applicable amount available for distribution shall be shared *pro rata* among such Class C Members and Class D Members based on the amounts to which such Class C Members and Class D Members would have been entitled pursuant to clauses (A) and (B) immediately above, as applicable.

(iv) <u>Fourth</u>, with respect to each outstanding Class A Unit, an amount equal to the unreturned Capital Contributions; <u>provided</u> that the amount of unreturned Capital Contributions with respect to each outstanding Class A Unit shall be reduced by the amount of any Distributions made with respect to such Class A Unit; <u>provided further</u> that

such amounts will be distributed to the Class A Members *pro rata* in accordance with their unreturned Capital Contributions;

- (v) <u>Fifth</u>, with respect to each outstanding Class A Unit, vested Class B Unit, and Common Unit, an amount equal to the remaining Liquidation Proceeds divided by aggregate number of all Class A Units, vested Class B Units and Common Units outstanding at the time of the Distribution which are entitled to participate in the Distribution (taking into account the computational rules of <u>Section 13.3(d)</u>); <u>provided</u> that no Distributions shall be made pursuant to this <u>Section 13.3(c)(v)</u> with respect to any Converted Common Unit until the amount of aggregate per-Unit Distributions pursuant to <u>Section 7.3</u>, <u>Section 13.3(c)(iv)</u> and this <u>Section 13.3(c)(v)</u> with respect to Class A Units and Common Units which are not Converted Common Units is equal to the amount previously distributed with respect to such Converted Common Unit, and any amount not distributed as a consequence of this proviso shall be re-allocated and distributed pursuant to this <u>Section 13.3(c)(v)</u>, with this proviso applied on an iterative basis; and
- (vi) Notwithstanding Section 13.3(c)(v), in the case of a holder of vested Class B Units, the Liquidation Proceeds shall be distributed to the holder of such vested Class B Unit only to the extent such Liquidation Proceeds exceed the Participation Threshold with respect to such vested Class B Unit. Distributions limited by the foregoing sentence shall be reduced by the applicable Class B Unit's Participation Threshold and shall, in lieu of distribution to the holder of such Class B Unit, be distributed *pro rata* to the Class A Units and the Class B Units not subject to such a limitation.
- (d) Application of Participation Threshold. If immediately prior to a distribution under Section 13.3(c)(v), the Participation Threshold with respect to any vested Class B Unit is greater than or equal to the amount that is determined by dividing (1) the sum of (x) the amount that is to be distributed under Section 13.3(c)(v), as applicable, plus (y) the aggregate remaining Participation Thresholds for all outstanding vested Class B Units (including such Class B Unit) that have a Participation Threshold that is less than that of such Class B Unit by (2) the total number of Units, excluding all vested Class B Units with a Participation Threshold greater than that of such Class B Unit, such Class B Unit shall not participate in such distribution and for purposes of determining the amount distributable with respect to each Unit with respect to such distribution such Class B Unit shall not be treated as outstanding. This clause (d) shall be applied first to the vested Class B Units having the greatest Participation Threshold and then seriatim to the vested Class B Units having the next greatest Participation Threshold until all vested Class B Units have been evaluated in the descending order of their Participation Thresholds.
- (e) Profits Unit Cap. The amount distributable with respect to a Class B Unit on liquidation shall not exceed the excess of (1) the sum of (i) the amounts available for distribution pursuant to Section 13.3(c)(iii), Section 13.3(c)(iv) and Section 13.3(c)(v) (such amount, the "Aggregate Liquidation Proceeds") plus (ii) the amount of distributions made pursuant to Section 7.2 after the date such Class B Unit was granted (but not including any such distributions made with respect to such Class B Unit) less (2) the Aggregate Liquidation Proceeds that would have been so available if the Company had, immediately before such Class B Unit was granted, wound up its

affairs, sold all of its assets for cash equal to their then fair market values, repaid all of its creditors, and distributed all remaining proceeds to the Members in complete liquidation (the excess of (1) over (2), for a particular Class B Unit, the "Profits Unit Cap"). For the sake of clarity, each Class B Unit shall have its own Profits Unit Cap. To avoid double counting, any Profits Unit Caps that are attributable in whole or in part to the same increase in Company value (the excess of clause (1) above over clause (2) above) shall be aggregated, and to the extent the sum of such Profits Unit Caps exceed such increase in Company value, each such Profits Unit Cap shall be reduced appropriately using a pro rata methodology that is consistent with Internal Revenue Service Revenue Procedures 93-27 and 2001-43. For example, if two Class B Units are granted at the same time and are the only Class B Units granted before liquidation of the Company, each Class B Unit shall have a Profits Unit Cap equal to half of the increase in Company value (the excess of clause (1) above over clause (2) above) from the date of grant to the date of liquidation. However, the foregoing Profits Unit Cap limitation shall not be construed to prevent the holders of Class B Units from receiving proceeds from a sale of Class B Units whether pursuant to Article X or otherwise.

- that require the liquidation of the assets of the Company, but subject to the order of priorities set forth in Section 13.3(c), if upon dissolution of the Company the Liquidator determines that an immediate sale of part or all of the Company's assets would be impractical or could cause undue loss to the Members, the Liquidator may defer the liquidation of any assets except those necessary to satisfy Company liabilities and reserves, and may, in its absolute discretion, Distribute to the Members, in lieu of cash, as tenants in common and in accordance with the provisions of Section 13.3(c), undivided interests in such Company assets as the Liquidator deems not suitable for liquidation. Any such Distribution in kind shall be subject to such conditions relating to the disposition and management of such properties as the Liquidator deems reasonable and equitable and to any agreements governing the operating of such properties at such time. For purposes of any such Distribution, any property to be Distributed shall be valued at its Fair Market Value.
- 13.4 <u>Cancellation of Certificate</u>. Upon completion of the Distribution of the assets of the Company as provided in <u>Section 13.3(c)</u> hereof, the Company shall be terminated and the Liquidator shall cause the cancellation of the Certificate of Formation in the State of Delaware and of all qualifications and registrations of the Company as a foreign limited liability company in jurisdictions other than the State of Delaware and shall take such other actions as may be necessary to terminate the Company.
- 13.5 <u>Survival of Rights, Duties and Obligations</u>. Dissolution, liquidation, winding up or termination of the Company for any reason shall not release any party from any Loss which at the time of such dissolution, liquidation, winding up or termination already had accrued to any other party or which thereafter may accrue in respect of any act or omission prior to such dissolution, liquidation, winding up or termination. For the avoidance of doubt, none of the foregoing shall replace, diminish or otherwise adversely affect any Member's right to indemnification pursuant to Section 14.3.
- 13.6 <u>Recourse for Claims</u>. Each Member shall look solely to the assets of the Company for all Distributions with respect to the Company, such Member's Capital Account, and such

Member's share of Net Income, Net Loss and other items of income, gain, loss and deduction, and shall have no recourse therefor (upon dissolution or otherwise) against the Board, the Liquidator or any other Member.

# ARTICLE XIV. EXCULPATION AND INDEMNIFICATION

### 14.1 Exculpation of Covered Persons.

- (a) <u>Covered Persons</u>. As used herein, the term "<u>Covered Person</u>" shall mean (i) each Member, (ii) each officer, director, shareholder, partner, member, Controlling Affiliate, employee, agent or representative of each Member, and each of their Controlling Affiliates, and (iii) (A) each Director, Alternate Director, Officer, Board Observer, employee, manager, agent or representative of the Company or (B) each director, alternate director or board observer (in each case, if and so long as such director, alternate director or board observer is a representative of a Member or an employee of the Company or any Company Subsidiary serving as such director, alternate director or board observer at the request of the Company or any Company Subsidiary), officer, employee, manager, agent or representative of any Company Subsidiary.
- (b) <u>Standard of Care</u>. No Covered Person shall be liable to the Company, any Company Subsidiary or any other Covered Person for any loss, damage or claim incurred by reason of any action taken or omitted to be taken by such Covered Person in good-faith reliance on the provisions of this Agreement, so long as such action or omission does not constitute fraud, bad faith or willful misconduct by such Covered Person.
- good faith upon the records of the Company and upon such information, opinions, reports or statements (including financial statements and information, opinions, reports or statements as to the value or amount of the assets, liabilities, Net Income or Net Losses of the Company or any facts pertinent to the existence and amount of assets from which Distributions might properly be paid) of the following Persons or groups: (i) another Director or Alternate Director; (ii) a Board Observer; (iii) one or more Officers or employees of the Company or any Company Subsidiary; (iv) any attorney, independent accountant, appraiser or other expert or professional employed or engaged by or on behalf of the Company or any Company Subsidiary; or (v) any other Person selected in good faith by or on behalf of the Company or any Company Subsidiary, in each case as to matters that such relying Person reasonably believes to be within such other Person's professional or expert competence. The preceding sentence shall in no way limit any Person's right to rely on information to the extent provided in § 18-406 of the Delaware Act.

### 14.2 Liabilities and Duties of Certain Persons.

- (a) <u>Limitation of Liability; Waiver of Fiduciary Duties.</u>
- (i) Except as set forth in <u>Section 14.2(a)(iii)</u>, this Agreement is not intended to, and does not, create or impose any fiduciary duty on any Member, Director,

Alternate Director or Board Observer in his or her capacity as such. Furthermore, each of the Members and the Company hereby waives any and all fiduciary duties with respect to such Persons, whether existing at law or in equity, that, absent such waiver, may be implied by Applicable Law, and in doing so, acknowledges and agrees that the duties and obligations of each Member, Director, Alternate Director or Board Observer to each other and to the Company are only as expressly set forth in this Agreement. The provisions of this Agreement, to the extent that they restrict the duties and liabilities of a Director, Alternate Director or Board Observer otherwise existing at law or in equity, are agreed by the Members to replace such other duties and liabilities of such Person.

- (ii) Whenever in this Agreement a Member, Director (other than the CEO when serving as a Director), Alternate Director or Board Observer is permitted or required to make a decision (including a decision that is in such Person's "discretion" or under a grant of similar authority or latitude), such Person shall be entitled to consider only such interests and factors as such Person desires, including its own interests, and shall have no duty or obligation to give any consideration to any interest of or factors affecting the Company or any other Person. Whenever in this Agreement such a Person is permitted or required to make a decision in such Person's "good faith," such Person shall act under such express standard and shall not be subject to any other or different standard imposed by this Agreement or any other Applicable Law.
- (iii) Notwithstanding any other provision in this Agreement (including in this Section 14.2(a)), each Director that is also an Officer shall (including, for the sake of clarity when serving as a Director) have all fiduciary duties, including without limitation a duty of loyalty and care as if he or she were a director of a business corporation organized under the General Corporation Law of the State of Delaware.
- (b) <u>Fiduciary Duties of Officers</u>. Each Officer of the Company shall have all fiduciary duties, including without limitation a duty of loyalty and care as if he or she were an officer of a business corporation organized under the General Corporation Law of the State of Delaware.

### 14.3 Indemnification.

(a) <u>Indemnification</u>. To the fullest extent permitted by the Delaware Act, as the same now exists or may hereafter be amended, substituted or replaced (but, in the case of any such amendment, substitution or replacement only to the extent that such amendment, substitution or replacement permits the Company to provide broader indemnification rights than the Delaware Act permitted the Company to provide prior to such amendment, substitution or replacement), the Company shall indemnify, hold harmless, defend, pay and reimburse any Covered Person against any and all losses, claims, damages, judgments, fines or liabilities, including reasonable legal fees or other expenses incurred in investigating or defending against such losses, claims, damages, judgments, fines or liabilities, and any amounts expended in settlement of any claims (collectively, "Losses") to which such Covered Person may become subject by reason of:

- (i) Any act or omission or alleged act or omission performed or omitted to be performed on behalf of the Company, any Member or Affiliate of the foregoing in connection with the business of the Company; or
- The fact that such Covered Person is or was acting in connection with (ii) the business of the Company as a partner, member, stockholder, Controlling Affiliate, manager, Director, Board Observer, Officer, employee or agent of the Company, any Member, or any of their respective Controlling Affiliates, or that such Covered Person is or was serving at the request of the Company as a partner, member, manager, director, board observer, officer, employee or agent of any Person including the Company or any Company Subsidiary; provided that (x) such Covered Person acted in good faith and in a manner believed by such Covered Person to be in, or not opposed to, the best interests of the Company and, with respect to any criminal proceeding, had no reasonable cause to believe his conduct was unlawful, and (y) such Covered Person's conduct did not constitute fraud, bad faith or willful misconduct, in either case as determined by a final, nonappealable order of a court of competent jurisdiction. In connection with the foregoing, the termination of any action, suit or proceeding by judgment, order, settlement, conviction, or upon a plea of nolo contendere or its equivalent, shall not, of itself, create a presumption that the Covered Person did not act in good faith or, with respect to any criminal proceeding, had reasonable cause to believe that such Covered Person's conduct was unlawful, or that the Covered Person's conduct constituted fraud, bad faith or willful misconduct.
- (b) <u>Reimbursement</u>. The Company shall advance the reasonable legal or other expenses (as incurred) of such Covered Person in connection with investigating, preparing to defend or defending any claim, lawsuit or other proceeding relating to any Losses for which such Covered Person may be indemnified pursuant to this <u>Section 14.3</u>; <u>provided</u>, that an advancement shall only be made upon delivery to the Company of an undertaking by or on behalf of the Covered Person to repay all amounts so advanced if it is finally judicially determined that such Covered Person is not entitled to the indemnification provided by this <u>Section 14.3</u>, and in such case, such Covered Person shall promptly reimburse the Company for any reimbursed or advanced expenses.
- shall not be deemed exclusive of any other rights to indemnification to which those seeking indemnification may be entitled under any agreement or otherwise. The provisions of this <u>Section 14.3</u> shall continue to afford protection to each Covered Person regardless of whether such Covered Person remains in the position or capacity pursuant to which such Covered Person became entitled to indemnification under this <u>Section 14.3</u> and shall inure to the benefit of the executors, administrators, legatees and distributees of such Covered Person.
- (d) <u>Insurance</u>. The Company shall purchase, at its expense, insurance to cover Losses covered by the foregoing indemnification provisions and to otherwise cover Losses for any breach or alleged breach by any Covered Person of such Covered Person's duties in such amount and with such deductibles as the Board may determine; <u>provided</u>, that the failure to obtain such insurance shall not affect the right to indemnification of any Covered Person under the indemnification provisions contained herein, including the right to be reimbursed or advanced

expenses or otherwise indemnified for Losses hereunder. If any Covered Person recovers any amounts in respect of any Losses from any insurance coverage, then such Covered Person shall, to the extent that such recovery is duplicative, reimburse the Company for any amounts previously paid to such Covered Person by the Company in respect of such Losses.

- (e) <u>Indemnification Agreement</u>. Upon such individual's election, appointment or designation as a Director or Alternate Director as provided in this Agreement, the Company shall promptly enter into an indemnification agreement on standard terms and conditions with such Director or Alternate Director.
- (f) <u>Funding of Indemnification Obligation</u>. Notwithstanding anything contained herein to the contrary, any indemnity by the Company relating to the matters covered in this <u>Section 14.3</u> shall be provided out of and to the extent of Company assets only, and no Member (unless such Member otherwise agrees in writing) shall have personal liability on account thereof or shall be required to make additional Capital Contributions to help satisfy such indemnity by the Company.
- (g) Indemnitor of First Resort. The Company and each Member hereby acknowledges that one or more of the Covered Persons may have certain rights to indemnification, advancement of expenses and/or insurance provided by an investment fund, institutional investor, financial institution, or other financial intermediary or Person with which such Person is Affiliated or of which such Person is a member, partner or employee (an "Affiliated Institution"). The Company and each Member hereby agrees that, with respect to any such Covered Person, the Company (i) is, relative to each Affiliated Institution, the indemnitor of first resort (i.e., its obligations to the applicable Covered Person under this Agreement are primary and any duplicative, overlapping or corresponding obligations of an Affiliated Institution are secondary), (ii) shall be required to make all advances and other payments under this Agreement, and shall be fully liable therefor, without regard to any rights any Covered Person may have against his or her Affiliated Institution, and (iii) irrevocably waives, relinquishes and releases any such Affiliated Institution from any and all claims against such Affiliated Institution for contribution, subrogation or any other recovery of any kind in respect thereof.
- (h) <u>Savings Clause</u>. If this <u>Section 14.3</u> or any portion hereof shall be invalidated on any ground by any court of competent jurisdiction, then the Company shall nevertheless indemnify and hold harmless each Covered Person pursuant to this <u>Section 14.3</u> to the fullest extent permitted by any applicable portion of this <u>Section 14.3</u> that shall not have been invalidated and to the fullest extent permitted by Applicable Law.
- (i) <u>Amendment</u>. The provisions of this <u>Section 14.3</u> shall be a contract between the Company, on the one hand, and each Covered Person who served in such capacity at any time while this <u>Section 14.3</u> is in effect, on the other hand, pursuant to which the Company and each such Covered Person intend to be legally bound. No amendment, modification or repeal of this <u>Section 14.3</u> that adversely affects the rights of a Covered Person to indemnification for Losses incurred or relating to a state of facts existing prior to such amendment, modification or repeal shall

apply in such a way as to eliminate or reduce such Covered Person's entitlement to indemnification for such Losses without the Covered Person's prior written consent.

14.4 <u>Survival</u>. The provisions of this <u>Article XIV</u> shall survive the dissolution, liquidation, winding up and termination of the Company.

# ARTICLE XV. MISCELLANEOUS

- 15.1 <u>Expenses</u>. Except as otherwise expressly provided herein, all costs and expenses, including fees and disbursements of counsel, financial advisors and accountants, incurred in connection with the preparation and execution of this Agreement, or any amendment or waiver hereof, and the transactions contemplated hereby shall be paid by the party incurring such costs and expenses.
- 15.2 <u>Further Assurances</u>. In connection with this Agreement and the transactions contemplated hereby, the Company and each Member hereby agrees, at the request of the Company or any other Member, to execute and deliver such additional documents, instruments, conveyances and assurances and to take such further actions as may be required to carry out the provisions hereof and give effect to the transactions contemplated hereby. Notwithstanding the foregoing, in no event shall any Member be required to execute a power of attorney pursuant to any provision in this Agreement.
- 15.3 Notices. All notices, requests, consents, claims, demands, waivers and other communications hereunder shall be in writing and shall be deemed to have been given: (a) when delivered by hand (with written confirmation of receipt); (b) when received by the addressee if sent by a nationally recognized overnight courier (receipt requested); (c) on the date sent by e-mail (with confirmation of transmission) if sent during normal business hours of the recipient, and on the next Business Day if sent after normal business hours of the recipient; or (d) on the third day after the date mailed, by certified or registered mail, return receipt requested, postage prepaid. Such communications must be sent to the respective parties at the following addresses (or at such other address for a party as shall be specified in a notice given in accordance with this Section 15.3):

If to the Company:

MEMX Holdings LLC 382 NE 191st Street, Suite 92178 Miami, FL 33179 Attn: Jonathan Kellner

If to a Member, to such Member's most recent email address or mailing address provided to Company by such Member or counsel to such Member, and with a copy to such persons identified by such Member or Member's counsel.

- 15.4 <u>Headings</u>. The headings in this Agreement are inserted for convenience or reference only and are in no way intended to describe, interpret, define, or limit the scope, extent or intent of this Agreement or any provision of this Agreement.
- 15.5 Severability. If any term or provision of this Agreement is held to be invalid, illegal or unenforceable under Applicable Law in any jurisdiction, such invalidity, illegality or unenforceability shall not affect any other term or provision of this Agreement or invalidate or render unenforceable such term or provision in any other jurisdiction. Upon such determination that any term or other provision is invalid, illegal or unenforceable, the parties hereto shall negotiate in good faith to modify this Agreement so as to effect the original intent of the parties as closely as possible in a mutually acceptable manner in order that the transactions contemplated hereby be consummated as originally contemplated to the greatest extent possible.
- 15.6 <u>Entire Agreement</u>. This Agreement, together with the Certificate of Formation and all related Exhibits and Schedules, constitutes the sole and entire agreement of the parties to this Agreement with respect to the subject matter contained herein and therein, and supersedes all prior and contemporaneous understandings, agreements, representations and warranties, both written and oral, with respect to such subject matter.
- 15.7 <u>Successors and Assigns</u>. Subject to the restrictions on Transfers set forth herein, this Agreement shall be binding upon and shall inure to the benefit of the parties hereto and their respective heirs, executors, administrators, successors and assigns.
- 15.8 No Third Party Beneficiaries. Except (a) as provided in Article XIV, which shall be for the benefit of and enforceable by Covered Persons as described therein, and (b) the rights of each Regulated Securities Exchange Subsidiary and MEMX Execution Services LLC expressly set forth in this Agreement, this Agreement is for the sole benefit of the parties hereto (and their respective heirs, executors, administrators, successors and assigns) and nothing herein, express or implied, is intended to or shall confer upon any other Person, including any creditor of the Company, any legal or equitable right, benefit or remedy of any nature whatsoever under or by reason of this Agreement.

### 15.9 Amendments.

(a) Notwithstanding the provisions of this Section 15.9, commencing on the Registration Date, and for so long as the Company shall Control, directly or indirectly, any Registered Securities Exchange Subsidiary, before any amendment to or repeal of any provisions in this Agreement shall be effective, the applicable changes shall be submitted to the Board of Directors of each Registered Securities Exchange Subsidiary for approval, and if approved, the proposed changes to this Agreement shall not be effective until filed with or filed with and approved by the SEC, as the case may be, to the extent required by Applicable Law, it being agreed that if the same must be filed with or filed with and approved by the SEC before the changes may be effective, under Section 19 of the Exchange Act and the rules promulgated under the Exchange Act by the SEC or otherwise, then the proposed changes to this Agreement shall not be effective until filed with or filed with and approved by the SEC, as the case may be.

- (b) No provision of this Agreement may be amended or modified unless such amendment or modification is approved by Supermajority Board Vote. Any such written amendment or modification shall be binding upon the Company and each Member; provided, that (i) an amendment or modification modifying the rights or obligations of any Member in a manner that is disproportionately adverse to (A) such Member relative to the rights of other Members in respect of Units of the same type, class or series or (B) a type, class or series of Units relative to the rights of another type, class or series of Units, shall in each case be effective only with that Member's consent or the consent of the Members holding a majority of the Units in that type, class or series, as applicable, and (ii) an amendment or modification which materially increases an existing liability or obligation on a Member or imposes a new material liability or a new material obligation on a Member shall require the consent of such Member.
- 15.10 Waiver. No waiver by any party or parties of any of the provisions hereof shall be effective unless explicitly set forth in writing and signed by the party or parties so waiving as provided herein, provided that (i) any proposed waiver that disproportionally affects any Member in any material respect (including any adverse change to such Member's governing rights) shall require the consent of each Member so affected; (ii) any proposed waiver that would or could reasonably be expected to result in adverse regulatory consequences to any Class A Member, Class C Member or Class D Member, as applicable, shall require the consent of each Class A Member, Class C Member or Class D Member, as applicable which would, or reasonably could, be so affected; (iii) any waiver pursuant to a section or provision of the Agreement specifying a voting, consent, approval right, threshold, or other requirement, shall require the approval of the Members constituting at least such voting, approval right, consent or approval threshold or otherwise satisfying such requirement; and (iv) any waiver with respect to a Member's limited liability or Capital Contribution obligations (including any requirement to contribute additional capital) shall require the consent of each Member impacted by such a waiver. No waiver by any party shall operate or be construed as a waiver in respect of any failure, breach or default not expressly identified by such written waiver, whether of a similar or different character, and whether occurring before or after that waiver. No failure to exercise, or delay in exercising, any right, remedy, power or privilege arising from this Agreement shall operate or be construed as a waiver thereof, nor shall any single or partial exercise of any right, remedy, power or privilege hereunder preclude any other or further exercise thereof or the exercise of any other right, remedy, power or privilege. For the avoidance of doubt, nothing contained in this Section 15.10 shall diminish any of the explicit and implicit waivers described in this Agreement, including in Section 3.10, Section 4.7(j), Section 4.8, Section 8.5(b), Section 8.5(c), Section 8.11, Section 9.1(e), Section 10.3(d)(iv), Section 10.4(b)(ii) and Section 15.13 hereof.
- 15.11 <u>Governing Law</u>. All issues and questions concerning the application, construction, validity, interpretation and enforcement of this Agreement shall be governed by and construed in accordance with the internal laws of the State of Delaware, without giving effect to any choice or conflict of law provision or rule (whether of the State of Delaware or any other jurisdiction) that would cause the application of laws of any jurisdiction other than those of the State of Delaware.

### 15.12 Submission to Jurisdiction.

- (a) The parties hereby agree that any suit, action or proceeding seeking to enforce any provision of, or based on any matter arising out of or in connection with, this Agreement or the transactions contemplated hereby, whether in contract, tort or otherwise, shall be brought in the United States District Court for the District of Delaware or in the Court of Chancery of the State of Delaware (or, if such court lacks subject matter jurisdiction, in the Superior Court of the State of Delaware), so long as one of such courts shall have subject-matter jurisdiction over such suit, action or proceeding, and that any cause of action arising out of this Agreement shall be deemed to have arisen from a transaction of business in the State of Delaware. Each of the parties hereby irrevocably consents to the jurisdiction of such courts (and of the appropriate appellate courts therefrom) in any such suit, action or proceeding and irrevocably waives, to the fullest extent permitted by law, any objection that it may now or hereafter have to the laying of the venue of any such suit, action or proceeding in any such court or that any such suit, action or proceeding which is brought in any such court has been brought in an inconvenient form. Service of process, summons, notice or other document by registered mail to the address set forth in Section 15.3 shall be effective service of process for any suit, action or other proceeding brought in any such court.
- Notwithstanding the provisions of Sections 15.11 and 15.12(a), the Company and its Directors, Officers, employees and agents, by virtue of their acceptance of such position, shall be deemed to irrevocably submit to the jurisdiction of the United States federal courts, the SEC, and each Registered Securities Exchange Subsidiary, for the purposes of any suit, action or proceeding pursuant to the United States federal securities laws, and the rules or regulations thereunder, arising out of, or relating to, the activities of such Registered Securities Exchange Subsidiary, and by virtue of their acceptance of any such position, shall be deemed to waive, and agree not to assert by way of motion, as a defense or otherwise in any such suit, action or proceeding, any claims that it or they are not personally subject to the jurisdiction of the United States federal courts, the SEC or the applicable Registered Securities Exchange Subsidiary, that the suit, action or proceeding is an inconvenient forum or that the venue of the suit, action or proceeding is improper, or that the subject matter of that suit, action or proceeding may not be enforced in or by such courts or agency. The Company and its Directors, Officers, employees and agents also agree that they shall maintain an agent, in the United States, for the service of process of a claim arising out of, or relating to, the activities of each Registered Securities Exchange Subsidiary.
- 15.13 <u>Waiver of Jury Trial</u>. Each party hereto hereby acknowledges and agrees that any controversy which may arise under this Agreement is likely to involve complicated and difficult issues and, therefore, each such party irrevocably and unconditionally waives any right it may have to a trial by jury in respect of any legal action arising out of or relating to this Agreement or the transactions contemplated hereby.
- 15.14 Equitable Remedies. Each party hereto acknowledges that a breach or threatened breach by such party of any of its obligations under this Agreement may give rise to irreparable harm to the other parties, for which monetary damages may not be an adequate remedy, and hereby agrees that in the event of a breach or a threatened breach by such party of any such obligations, each of the other parties hereto shall, in addition to any and all other rights and remedies that may be available to them in respect of such breach, be entitled to seek equitable relief, including a

temporary restraining order, an injunction, specific performance and any other relief that may be available from a court of competent jurisdiction (without any requirement to post bond).

- 15.15 Attorneys' Fees. In the event that any party hereto institutes any legal suit, action or proceeding, including arbitration, against another party in respect of a matter arising out of or relating to this Agreement, the prevailing party in the suit, action or proceeding shall be entitled to receive, in addition to all other damages to which it may be entitled, the costs incurred by such party in conducting the suit, action or proceeding, including reasonable attorneys' fees and expenses and court costs.
- 15.16 <u>Remedies Cumulative</u>. The rights and remedies under this Agreement are cumulative and are in addition to and not in substitution for any other rights and remedies available at law or in equity or otherwise, except to the extent expressly provided in <u>Section 14.2</u> to the contrary.
- 15.17 <u>Counterparts</u>. This Agreement may be executed in counterparts, each of which shall be deemed an original, but all of which together shall be deemed to be one and the same agreement. A signed copy of this Agreement delivered by facsimile, e-mail or other means of Electronic Transmission shall be deemed to have the same legal effect as delivery of an original signed copy of this Agreement.
- 15.18 Restrictions on Non-U.S. Operations. The Company shall not own (directly or indirectly) any interest in a separate unit (foreign branch or hybrid entity, as defined in Treasury Regulations Section 1.1503(d)-1(b)(3)) for purposes of the dual consolidated loss rules promulgated under Code Section 1503(d). The Company shall not either (a) directly or indirectly (other than through an entity that is treated as a corporation for United States federal income tax purposes) own assets, conduct business or otherwise engage in activities that would cause it to have a permanent establishment outside the United States or a foreign branch, as that term is defined in Treasury Regulations Section 1.367(a)-6T(g)(1) for purposes of the dual consolidated loss rules in Treasury Regulations Section 1.1503(d)-1(b)(4), or (b) cause any Member to file an income tax return or pay income tax in any jurisdiction outside the United States solely as a result of the Company's activities in such jurisdiction. To mitigate this risk, the Company shall consult with tax counsel prior to directly or indirectly engaging in any business or activities that reasonably may create a risk of being treated as a permanent establishment or foreign branch, including for example, maintaining or conducting business through a fixed place of business outside the United States (a place of management, branch, office, or other ownership of real property outside the United States), sending one or more employees to work at a fixed location outside the United States for a significant period of time, engaging in business activities outside the United States that are reflected in a separate set of books and records, or maintaining outside the United States agents or employees with the authority to enter into contracts on behalf of the Company.

[Signature page follows]

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be executed as of the date first written above by their respective officers thereunto duly authorized.

Com	pan	V	•

<b>MEMX</b>	Holdings	LLC

By:		
Name:		
Title:		

# Citadel Securities Principal Investments LLC

By:		
Name:		
Title:		

# Virtu Investments, LLC

By:		
Name:		
Title:		

# **Banc of America Strategic Investments Corporation**

By:		
Name:		
Title:		

**Strategic Investments I, Inc.** 

By:			
Name:			
Title:			

**UBS Americas Inc.** 

By:				
Name	:			
Title:				
UBS A	Americas Inc.			
By:				
Name	:			
Title:		•	•	

**FMR LLC** 

By:			
Name:			
Title:			

## **Class A Member:**

The Charles Sentian Corporation	arles Schwab Corporation
---------------------------------	--------------------------

By:		
Name:		
Title:		

## **Class A Member:**

Datek Or	ıline	Management	Corp.
----------	-------	------------	-------

By:		
Name:		
Title:		

## Goldman Sachs PSI Global Holdings, LLC

By:			
Name:			
Title:	_		

Jane Street Group, LLC

By:			
Name:			
Title:			

# JPMC Strategic Investments I Corporation

By:		
Name:		
Title:		

Class A Member and Class C Member:	
David B. Williams	
Caroline H. Williams	
Caronic II. Williams	

Wells Fargo Central Pacific Holdings, Inc.

By:		
Name:		
Title:		

# Flow Traders U.S. Holding LLC

By:	
Name:	
Title:	

## **Class A Member:**

BLK	SMI.	, LLC

By:	
Name:	
Title:	

# **Class A Member and Class C Member:**

# Manikay Global Opportunities 2, LP

By:			
Name:			_
Title:			

# **Class A Member and Class C Member:**

Citicorp North America, Inc.

By:			
Name:			
Title:			

	<b>Optiver</b>	<b>PSI</b>	<b>B</b> 1	LL	C
--	----------------	------------	------------	----	---

By:		
Name:		
Title:		

<b>IMC</b>	Strategic	<b>Holdings</b>	LLC
------------	-----------	-----------------	-----

By:			
Name:			
Title:			

CTC	Venture	<b>Capital</b>	, LLC

By:			
Name:			
Title:			

Belvedere Stra	ategic	Capi	ital	LP
----------------	--------	------	------	----

By:		
Name:		
Title:		

BSC MEMX, a Series of Belvedere Strategic Capital SPV Master LLC

By:	
Name:	
Title:	

## **EXHIBIT A- Form of Adherence Agreement**

# ADHERENCE AGREEMENT TO THE EIGHTH AMENDED AND RESTATED LIMITED LIABILITY COMPANY AGREEMENT OF MEMX HOLDINGS LLC

This is an ADHERENCE AGREEMENT (the " <u>Adherence Agreement</u> ") dated as of, 2 to the Eighth Amended and Restated Limited Liability Company Agreement of MEMX Holdings LLC (the " <u>Company</u> "), dated as of [], 2023 (as the same
may be amended, restated, supplemented or otherwise modified from time to time, the "Agreement"), by and among the Members party thereto (individually, a "Member," and collectively, the "Members").
Capitalized terms used herein and not otherwise defined herein shall have the meanings assigned to such terms in the Agreement.
WHEREAS, [Section 4.1(b)) of the Agreement provides that additional Persons may become Members of the Company by execution and delivery of an instrument in the form of this Adherence Agreement] [Section 4.5 of the Agreement provides that Member's heir may become a Member of the Company by execution and delivery of an instrument in the form of this Adherence Agreement]. The undersigned new Member of the Company (the "New Member") is executing this Adherence Agreement in accordance with the requirements of the Agreement to become a Member under the Agreement.
Accordingly, the Company and the New Member agree as follows:
SECTION 1. In accordance with [Section 4.1(b)][Section 4.5] of the Agreement, the New Member by its signature below becomes a Member under the Agreement with the same force and effect as if originally named therein as a Member, and the New Member hereby agrees to be bound by and subject to all of the terms and conditions specified in the Agreement as a Member thereunder. Each reference to a "Member" in the Agreement shall be deemed to include the New Member. The Agreement is hereby incorporated herein by reference.
[SECTION 2. The Company shall issue to the New Member, in exchange for] <sup>1</sup>
SECTION 3. The New Member represents and warrants to the Company and the other Members that this Adherence Agreement and the Agreement constitute the legal, valid and
<sup>1</sup> NTD to Adherence Agreement Draft: Insert the number and type of Units being issued to the New Member and the description of the capital contribution being made by such New Member, if and as applicable.

binding obligation of the New Member, enforceable against the New Member in accordance with its terms.

SECTION 4. This Adherence Agreement may be executed in counterparts, each of which shall constitute an original, but all of which when taken together shall constitute a single contract. This Adherence Agreement shall become effective when the Company shall have received counterparts of this Adherence Agreement that, when taken together, bear the signatures of the New Member and the Company.

SECTION 5. Except as expressly supplemented hereby, the Agreement shall remain in full force and effect.

SECTION 6. All issues and questions concerning the application, construction, validity, interpretation and enforcement of this Adherence Agreement shall be governed by and construed in accordance with the internal laws of the State of Delaware, without giving effect to any choice or conflict of law provision or rule (whether of the State of Delaware or any other jurisdiction) that would cause the application of laws of any jurisdiction other than those of the State of Delaware.

SECTION 7. All communications and notices hereunder shall be in writing and given as provided in <u>Section 15.3</u> of the Agreement.

IN WITNESS WHEREOF, the New Member and the Company have duly executed this Adherence Agreement as of the day and year first above written.

## MEMX HOLDINGS LLC

By:
Name:
Title:
NEW MEMBER]
By:
Name:
Title:
New Member Address for Notices:
Attention:
E-mail:

## **EXHIBIT B – Directors and Observers Schedule**

## **Directors and Alternate Directors**

CEO Director: Jonathan Kellner

Nominating Member	Director	Alternate Director
Citadel Securities Principal Investments LLC	Matt Culek	Donnie Phillips
Virtu Investments, LLC	Douglas Cifu	Andrew Smith
Banc of America Strategic Investments Corporation	Pankil Patel	Mitsu Narang
Strategic Investments I, Inc.	Zheng Wang	Christopher Larkin
FMR, LLC	Derrick Chan	Christopher Gwozdz
Goldman Sachs PSI Global Holdings, LLC	Raj Mahajan	Kevin Maslowe
Jane Street Group, LLC	Brian Nigito	Andrew Upward
JPMC Strategic Investments I Corporation	Christopher Berthe	Matthew Cramer
BLK SMI, LLC	Hubert De Jesus	Samara Cohen
The Charles Schwab Corporation	Jeffrey Starr	Jason Clague
Citicorp North America, Inc.	Michael Masone	Siris Singh
UBS Americas Inc.	Todd Lopez	Hyder Jaffrey

Wells Fargo Central Pacific Holdings, Inc.	Inessa Even	Niall O'Brien
Optiver PSI B1 LLC	Rutger Brinkhuis	Liam Smith

#### **EXHIBIT C – Board Matters**

Notwithstanding the provisions of this Exhibit C, approval by Supermajority Board Vote of any Registered Securities Exchange Subsidiary engaging in, or causing, any of the transactions or taking any of the actions either (x) set forth below or (y) otherwise requiring a Supermajority Board Vote pursuant to any other provision of this Agreement shall not be required if such Registered Securities Exchange Subsidiary engaging in or causing any applicable transaction or taking any applicable action is required in order for such Registered Securities Exchange Subsidiary to fulfill its regulatory functions or responsibilities or to oversee the structure of the market that such Registered Securities Exchange Subsidiary regulates, in each case as determined by the Exchange Board of such Regulated Securities Exchange Subsidiary upon receipt of an opinion of counsel (which may be in-house counsel and need not be in writing) to such effect. In addition, the following matters shall constitute Supermajority Board Matters and require a Supermajority Board Vote:

- 1. Subject to <u>Section 15.9</u>, any amendment of the Certificate of Formation or this Agreement or any constitutive document of any Company Subsidiary.
- 2. Admission of new Members (including the valuation of the Company at the time the applicable Person becomes a Member) and the amount of such Member's Capital Contribution, in each case, other than in connection with a Transfer of Units by a Member in accordance with the terms of this Agreement, and, if applicable, granting such new Member a right to nominate a Director or Exchange Director or appoint a Board Observer or Exchange Board Observer (unless such nomination or appointment is already permitted by this Agreement and except as contemplated by Section 8.10(b), Section 8.13(b), Section 8.18(f)(iii) and Section 8.18(g)(iii).
- 3. Issuance of any Units or Unit Equivalents beyond those issued and outstanding as of the date hereof (except upon the conversion of any Class A Unit, Class C Unit or Class D Unit in accordance with Section 3.10(d), Section 3.10(e) and Section 3.11) or the issuance of any Units that are senior in any respect to such Units.
- 4. Issuance of any securities by any Company Subsidiary beyond those issued and outstanding as of the Effective Date.
- 5. Any exchange, reclassification, cancellation (whether by merger, consolidation or otherwise) or modification of the terms of any Units, Unit Equivalents or other ownership interests in the Company or any Company Subsidiary.
- 6. Acquisition of any assets by the Company or any Company Subsidiary the value of which equals or exceeds, individually or in the aggregate (with respect to assets acquired in the same transaction or a series of related transactions), one million dollars (\$1,000,000) (or a higher amount approved by the Board by Supermajority

- Board Vote), unless such acquisition was previously included in the Annual Budget or annual business plan approved by Supermajority Board Vote.
- 7. Any investment by the Company or any Company Subsidiary in any other entity the value of which equals or exceeds, individually or in the aggregate (with respect to a series of related transactions), three million dollars (\$3,000,000) (or a higher amount approved by the Board by Supermajority Board Vote).
- 8. Any transaction that would result in a Change of Control.
- 9. Decision to commence an Initial Public Offering (including the conversion of the Company or a Company Subsidiary into an IPO Entity in connection with such Initial Public Offering and the determination of which exchange the applicable securities will be listed on).
- 10. Entry by the Company or a Company Subsidiary into, or termination by the Company or a Company Subsidiary of, any material strategic alliance, joint venture or partnership.
- 11. Any action that results in a liquidation or dissolution of the Company or any Company Subsidiary.
- 12. Purchase or redemption of any Units or other equity securities of the Company or any Company Subsidiary, other than repurchases in the ordinary course of business of Units or other equity securities held by employees of the Company or any Company Subsidiary in connection with the termination of their employment and in accordance with the Incentive Plan or any other equity incentive plan adopted by the Board.
- 13. Increasing or decreasing the size of the Board except as otherwise provided in the Agreement.
- 14. Approval of any Company Related Party Transactions, any amendments to the terms of such Company Related Party Transactions and any determinations that particular transactions (or particular categories of transactions) do not constitute Company Related Party Transactions.
- 15. Approval of the Annual Budget and/or annual (or multi-annual, as applicable) business plan of the Company and the Company Subsidiaries, or a material modification to any such Annual Budget or business plan, or material deviation in excess of fifteen percent (15%) from any such Annual Budget or business plan.
- 16. Effecting the grant of an exclusive license to any intellectual property of the Company or any Company Subsidiary.

- 17. Establishing, creating or entering into any plan, arrangement or agreement that would constitute an equity incentive plan or similar bonus plan, or amend the terms of any such equity incentive plan (including the Incentive Plan) or similar bonus plan.
- 18. Entering into any award agreement pursuant to any equity incentive plan (including the Incentive Plan) with more favorable terms than set forth in the applicable equity incentive plan.
- 19. Exercising the right of first offer pursuant to <u>Section 10.3</u> on behalf of the Company.
- 20. A waiver of the Members' preemptive rights in connection with a Change of Control as set forth in Section 9.1.
- 21. Determination of value of any property contributed to the Company by a Member.
- 22. Incurrence of any indebtedness (excluding trade debt incurred in the ordinary course of business), by the Company or any Company Subsidiary greater than one million dollars (\$1,000,000), individually or in the aggregate, unless such indebtedness was previously included in an Annual Budget or annual business plan approved by Supermajority Board Vote.
- 23. Making any, or entering into any commitment for, capital expenditures of the Company for in excess of five million dollars (\$5,000,000) for all commitments in the aggregate, unless such capital expenditures were previously included in an Annual Budget or annual business plan approved by Supermajority Board Vote.
- 24. Lending money to, or guarantying any obligation of, any Person (excluding trade receivables accrued in the ordinary course of business).
- 25. Establishing or eliminating a committee of the Board or of the board of directors or a similar governing body of any Company Subsidiary, or materially amending the governing documents of a committee of the Board, or of the board of directors or a similar governing body of any Company Subsidiary.
- 26. Establishing or eliminating the Industry Advisory Board, materially amending the governing documents of such Industry Advisory Board, or permitting a Member to continue to have the right to nominate an Industry Advisory Board Member following such Member's loss of the right to nominate at least one (1) Director.
- 27. Declaration, allocation or distribution of profits or capital (including establishment of any dividend policy) by the Company (or by a Company Subsidiary other than to the Company), other than Tax Advances.

- 28. Appointment, determination of compensation (salary and bonus, whether cash or equity) and termination of the CEO.
- 29. In the event that Units are Transferred to an Affiliate of a Member pursuant to Section 10.2 and such Affiliate subsequently becomes a non-Affiliate of such Member, allowing such non-Affiliate to retain its Units rather than Transferring them back to the Member from which the Units were received.
- 30. Approving Emergency Funding.
- 31. Formation of any new Company Subsidiary or ownership of any Subsidiary that is not wholly-owned by the Company.
- 32. Change in legal entity form of the Company or any Company Subsidiary (including, by way of example only, conversion into an IPO Entity).
- 33. Commencing, settling or compromising any material legal proceedings, investigations or governmental action brought by or against the Company; provided, however, that no Supermajority Board Vote shall be required to settle or compromise any legal proceeding, investigation or governmental action if the amount payable by the Company does not exceed one million dollars (\$1,000,000) (or a higher amount approved by the Board by Supermajority Board Vote) and such settlement or compromise does not result in any material ongoing obligations of the Company.
- 34. Any action which would subject the Company or a Company Subsidiary to additional material regulatory requirements.
- 35. Taking of any action by the Company or any Company Subsidiary that would or could result in material legal consequences, material regulatory consequences, or material reputational consequences to any Member or require any Member to file any application or notice for approval with its regulators.
- 36. Making or changing any material tax election, including any election pursuant to the Partnership Audit Rules or any election to change the classification of the Company as a partnership for U.S. federal income tax purposes and/or converting the Company to a corporation, approving any tax returns, determining the apportionment of any imputed underpayment within the meaning of Code Section 6225 to the Members, or settling or compromising any tax claim or liability or any investigation or dispute with a Governmental Authority with respect to any taxes.
- 37. Acquisition, disposition or entering into a new business or business line or a material change in the scope of the business or a current line of business by or of the Company or any Company Subsidiary.

- 38. Waiver of the limitations set forth in <u>Sections 3.5(a)(i)</u> and <u>3.5(a)(iii)</u>, in accordance with <u>Section 3.5(b)(ii)</u>.
- 39. Any matter subject to determination by Supermajority Board Vote pursuant to Section 1.4 of Exhibit G.
- 40. Entering into any agreement or commitment to do effectuate any Supermajority Board Matter on behalf of the Company or any Company Subsidiary.

#### **EXHIBIT D – Antitrust Guidelines**

Organizations such as the Company, the Company Subsidiaries and the Members, which may be current ("actual") or potential ("future") competitors, are subject to close scrutiny under the U.S. antitrust laws and parallel laws of other jurisdictions. Among other types of conduct prohibited by the antitrust laws, the laws generally prohibit concerted practices and express or implied agreements among market competitors that are likely to have an anticompetitive impact. This prohibition generally extends to cover exchanges of competitively sensitive information, because such exchanges may facilitate anticompetitive agreements. Such conduct -- even if inadvertent or undertaken in connection with a legitimate objective -- can give rise to substantial criminal and/or administrative penalties and civil liabilities (including for individuals) under U.S., European and other antitrust laws. The Members, the Board, each committee of the Board, and the board of directors or similar governing body of any Company Subsidiary, if any, must exercise diligence, procedurally and substantively, to ensure that their activities are conducted within all parameters established by these laws, both as a matter of appearance and in actual fact.

The following guidelines are designed to assist the Company, each Company Subsidiary, the Members, the Board and each governing body of any Company Subsidiary, each committee of the Board or such governing body and the Industry Advisory Board (each, a "Covered Body") in guiding their conduct relating to their discussions and exchanges and the activities of the Company and each Company Subsidiary, as applicable. In addition, these guidelines should be made available to, and followed by all individuals representing the Company, any Company Subsidiary, the Members and other persons who participate in meetings, discussions and exchanges regarding the Company, any Company Subsidiary and their respective businesses. Furthermore, these guidelines are not intended to be exhaustive; rather they are intended to provide and serve as general guidance. Antitrust Counsel should be consulted in the event of any questions concerning the meaning or implementation of these guidelines or their application in a particular instance.

#### **Practices to Follow**

- Ensure that all individuals appointed by each Covered Body and their advisors are provided a copy of these guidelines and informed of the importance of complying with them.
- Prepare an agenda to the extent practicable for each formal or informal meeting of a Covered Body (whether in person or by phone) (each such meeting, a "Meeting"), strictly limited to matters required for the furtherance of the lawful, procompetitive business of the Company or any Company Subsidiary, as applicable, and distribute the agenda to all participants prior to the applicable meeting. Agendas can be set forth in emails and, when electronic calendar invites are used for purposes of setting a meeting, an agenda can be included in the calendar invite or as the topic of such calendar invite. Counsel should review the

- agenda in advance of each such Meeting to ensure it does not contain anything objectionable on antitrust grounds.
- To the extent appropriate, strictly limit discussion at each Meeting to the agenda items. At no time should discussions diverge from matters related to the furtherance of the Company's business.
- To the extent practicable, have minutes taken of each Meeting that accurately reflect the subjects discussed, decisions made or actions taken, and the persons in attendance, and promptly distribute the minutes to all participants for review. Do not maintain unofficial minutes of Meetings. To the extent possible, have counsel present at any such meeting.
- Matters bearing on competitive interactions between or among the Members, or other competitively sensitive information of the Members, should not be discussed between or among the Members or at any Meetings.
- To the extent practicable, confine group discussions of all matters to formal, scheduled Meetings rather than engaging in side or splinter sessions before or after scheduled Meetings. If discussions at a Meeting stray into areas giving rise to antitrust concerns, participants should interrupt the discussion and register their objections.
- Consult with Antitrust Counsel on all antitrust questions, or matters that may give rise to an antitrust question, relating to the activities of the Company or any Company Subsidiary.
- Orally advise Antitrust Counsel or the CEO of any possible violations of these guidelines as soon as practicable. Avoid written discussion (including emails and texts) concerning potential antitrust violations.

#### Conduct to Avoid

Notwithstanding the foregoing, absent permission from counsel, the matters identified below should not be discussed or be the subject of informational exchanges by or among the Members or the Directors whether seriously or in jest, and no express or implied agreements should be formed with respect to any such matters at any Meeting, at any other gathering incidental to a Meeting, or by any other means:

Member-specific commercially or competitively sensitive information, including pricing policies or patterns, price levels, price differentials or similar matters, sales or trading levels or inventories, or access to particular markets or services (including planned or anticipated changes in any of the foregoing), and customer information, including current or prospective customer lists and terms and conditions of arrangements with customers.

- Territorial restrictions, allocations of customers, restrictions on types of products or services offered by Members, or any other kind of market division or any suggestions of any of the foregoing.
- Present or future prices of services or products other than those provided by the Company or any Company Subsidiary, including pricing practices and strategies, methods, timing, or implementation of price changes or what constitutes a fair, appropriate or "rational" price or profit margin.
- Volumes of transactions or other commercial activities, services or products of individual Members.
- Current or future business strategies and marketing plans of individual Members.
- Compensation information or policies of individual Members and their respective employees.
- Competitively sensitive business practices of individual Members, including but not limited to the following:
  - o recent, intended or proposed trades, including information on pricing and pricing terms (e.g., discounts, commissions, bid-offer spreads, fees), volume of orders, client identities, client orders/requests, and intent to offer/execute;
  - o current or recent information on costs, margins and profitability of trading positions;
  - o trading strategies of individual Members and their respective clients; and
  - o pricing strategies and policies of individual Members, including both general and client-specific strategies and policies, and terms with brokers or other intermediaries.
- Past, present or expected negotiations with customers or suppliers.
- Any refusal to deal with actual or potential competitors, customers or suppliers, either through the present planned collaboration or otherwise.
- Any other discussion or exchange of information that in actuality or even in appearance, could constitute an agreement to lessen competition among the Members or that might have the effect of excluding actual or potential competitors, customers or suppliers from any market or influencing the business conduct of any Person with respect thereto.

Refer any questions about whether a particular piece of information is competitively sensitive to Antitrust Counsel. In addition, any exchange of information between a Member, on the one hand, and the Company or any Company Subsidiary, on the other hand, which may contain commercially or competitively sensitive information should be reviewed with Antitrust Counsel.

# **EXHIBIT E – Form of Restricted Economic Election Notice**

Effective Date:
In accordance with Section 3.9 of the Eighth Amended and Restated Limited Liability Company Agreement (as may be amended from time to time) (the "Agreement") of MEMX Holdings LLC (the "Company"), the undersigned Member hereby provides an irrevocable notice to the Company of its election, effective as of the date set forth above, to be treated for purposes of the Agreement as a Restricted Economic Member. Capitalized terms used but not defined in this election form shall have the meanings set forth in the Agreement.
The maximum percentage of the aggregate economic interests attributable to the Units that such Member may own (such Restricted Economic Member's "Maximum Aggregate Economic Interest") shall be:
Maximum Aggregate Economic Interest:%
The undersigned Member acknowledges completing this form independently, after consultation with its own advisors to the extent such Member deems necessary.

[Signature Page Follows]

IN WITNESS WHEREOF, the undersigned, intending to be legally bound hereby, have duly executed this Restricted Economic Election as of the date set forth below.

	MEMBER:	
	[Insert Name of Member]	
	By:	
	Name: Title:	
Accepted:		
MEMX Holdings LLC		
By:		
Name:		
Title:		

# **EXHIBIT F – Restricted Voting Election Notice**

Effective Date:			
In accordance with Section 3.10 of the Eighth Amended and Restated Limited Liability Company Agreement (as may be amended from time to time) (the "Agreement") of MEMX Holdings LLC (the "Company"), the undersigned Member hereby provides an irrevocable notice to the Company of its election, effective as of the date set forth above, to be treated for purposes of the Agreement as a Restricted Voting Member; provided, that the undersigned Member may provide an amended notice to the Company of its election to specify (a) a Maximum Voting Class A Voting Percentage that is greater than its then-applicable Maximum Voting Class A Voting Percentage, or (b) Maximum Class C-1/D-1 Voting Percentage that is greater than its then-applicable Maximum Class C-1/D-1 Voting Percentage, as applicable, in each case solely ir order to give effect to the voting power associated with a purchase of additional Units, but no such election is permissible in connection with the receipt of additional Units by means of any subdivision, distribution in kind or other circumstance not constituting such a purchase of additional Units; provided further, that in no event shall any change by a Restricted Voting Member to its Maximum Voting Class A Voting Percentage or Maximum Class C-1/D-1 Voting Percentage result in or allow Units that were or are held or controlled by such Member to be or become voting Units, including through conversion or transfer.			
Capitalized terms used but not defined in this election form shall have the meanings set forth in the Agreement.			
The maximum percentage of the aggregate voting interests attributable to the Voting Class A Units, Class C-1 Units and/or Class D-1 Units that such Member may own shall be:			
Maximum Voting Class A Voting Percentage:%			
Maximum Class C-1/D-1 Voting Percentage:%			
The undersigned Member acknowledges completing this form independently, after			

[Signature Page Follows]

consultation with its own advisors to the extent such Member deems necessary.

IN WITNESS WHEREOF, the undersigned, intending to be legally bound hereby, have duly executed this Restricted Voting Election as of the date set forth below.

	MEMBER:	
	[Insert Name of Member]	
	By: Name:	
	Title:	
Accepted:		
MEMX Holdings LLC		
By:		
Name:		
Title:		

## **EXHIBIT G - Conversion Rights of Class C Units and Class D Units**

The conversion rights of the Class C Units and Class D Units shall be as follows, subject, in all cases, to <u>Section 3.10(i)</u> and (x) with respect to the Class C Units, <u>Section 3.11(c)</u> and, (y) with respect to the Class D Units, <u>Section 3.11(d)</u>:

- 1.1 Conversion Right; Conversion Ratio.
- (a) Each Class C Unit shall be convertible, at the option of the holder thereof, at any time and from time to time, and without the payment of additional consideration by the holder thereof, into such number of fully paid and nonassessable Common Units that equal the quotient of the Class C Unit Original Purchase Price, divided by the applicable Class C Unit Conversion Price in effect at such time. The "Class C Unit Conversion Price" shall initially equal the Class C Unit Original Purchase Price. The Class C Unit Conversion Price shall be subject to adjustment as provided in this Exhibit G.
- (b) Each Class D Unit shall be convertible, at the option of the holder thereof, at any time and from time to time, and without the payment of additional consideration by the holder thereof, into such number of fully paid and nonassessable Common Units that equal the quotient of the Class D Unit Original Purchase Price, <u>divided</u> by the applicable Class D Unit Conversion Price in effect at such time. The "<u>Class D Unit Conversion Price</u>" shall initially equal the Class D Unit Original Purchase Price. The Class D Unit Conversion Price shall be subject to adjustment as provided in this <u>Exhibit G</u>.
- 1.2 <u>Fractional Units</u>. No fractional Common Units shall be issued upon the conversion of Class C Units or Class D Units pursuant to <u>Section 1.1</u>. In lieu of any fractional Common Units to which the holder would otherwise be entitled, the Company shall pay cash equal to such fraction <u>multiplied</u> by the fair market value of one (1) Common Unit as determined in good faith by the Board.

## 1.3 Mechanics of an Optional Conversion.

(a) Notice of Conversion. In order for a holder of Class C Units or Class D Units, as applicable, to voluntarily convert such Class C Units or Class D Units into Common Units pursuant to Section 1.1, such holder shall (i) provide the Company with written notice that such holder elects to convert all or any number of such holder's Class C Units or Class D Units and, if applicable, any event on which such conversion is contingent, and, (ii) if certificates for the applicable Class C Units or Class D Units have theretofore been issued, surrender such certificate or certificates (or, if such holder alleges that such certificate has been lost, stolen or destroyed, a lost certificate affidavit and an agreement reasonably acceptable to the Company to indemnify the Company against any claim that may be made against the Company on account of the alleged loss, theft or destruction of such certificate) to the Company at the place designated in such notice (each, a "Conversion Notice"). If certificates have been theretofore issued for the applicable Class C Units or Class D Units and if required by the

Company, certificates surrendered for conversion shall be endorsed or accompanied by a written instrument or instruments of transfer, in form reasonably satisfactory to the Company, duly executed by the registered holder or his, her or its attorney duly authorized in writing. The close of business on the date of receipt by the Company of such notice (or the date of such contingent event specified therein) and, if applicable, certificates (or lost certificate affidavit) shall be the time of conversion (the "Conversion Time"), and the Common Units issuable upon conversion of the applicable Class C Units or Class D Units shall be deemed to be outstanding of record as of such date. The Company shall, as soon as practicable after the Conversion Time, (A) issue and deliver to the holder of the applicable Class C Units or Class D Units an updated version of the Members Schedule reflecting the issuance of the Common Units (together with a certificate or certificates for the number of full Common Units issuable upon such conversion in accordance with the provisions hereof, if certificates for Common Units have been issued); provided that failure to deliver any or all of the foregoing shall not affect such conversion, which shall be deemed to be effective notwithstanding any such failure, and (B) where applicable, pay in cash such amount as provided in Section 1.2 in lieu of any fraction of a Common Unit otherwise issuable upon such conversion.

- (b) <u>Effect of Conversion</u>. All Class C Units or Class D Units, as applicable, which shall have been surrendered for conversion as herein provided shall no longer be deemed to be outstanding and all rights with respect to such surrendered Class C Units or Class D Units shall immediately cease and terminate at the Conversion Time, except only the right of the holders thereof to receive Common Units in exchange therefor and to receive payment in lieu of any fraction of a Common Unit otherwise issuable upon such conversion as provided in <u>Section 1.2</u>.
- (c) <u>Taxes</u>. The Company shall pay any and all transfer and other similar taxes that may be payable in respect of any issuance or delivery of Common Units upon conversion of Class C Units or Class D Units, as applicable, pursuant to this <u>Exhibit G</u>. The Company shall not, however, be required to pay any tax which may be payable in respect of any transfer involved in the issuance and delivery of Common Units in a name other than that in which the Class C Units or Class D Units, as applicable, so converted were registered, and no such issuance or delivery shall be made unless and until the Person requesting such issuance has paid to the Company the amount of any such tax, or has established, to the reasonable satisfaction of the Company, that such tax has been paid.
- 1.4 <u>Adjustment to Class C Unit Conversion Price and Class D Unit Conversion Price for Diluting Issues.</u>
- (a) <u>Special Definitions</u>. The following terms shall have the following meanings for purposes of this <u>Exhibit G</u>:
- (i) "Option" shall mean rights, options or warrants to subscribe for, purchase or otherwise acquire Common Units or Convertible Securities.

- (ii) "<u>Class C Unit Original Issue Date</u>" shall mean the date on which the first Class C Unit was issued.
- (iii) "Class D Unit Original Issue Date" shall mean the date on which the first Class D Unit was issued.
- (iv) "<u>Convertible Securities</u>" means any Units or Unit Equivalents directly or indirectly convertible into or exchangeable for Common Units, but excluding Options.
  - (v) "Additional Issued Common Units" means all Common Units issued (or, pursuant to Section 1.4(c) below, deemed to be issued) by the Company (1) with respect to Class C Units, on or after the Class C Unit Original Issue Date, and (2) with respect to Class D Units, on or after the Class D Unit Original Issue Date, other than the following ("Exempted Securities"):
    - (A) Common Units, Options or Convertible Securities issued by reason of a Distribution on the Class C Units or Class D Units, as applicable;
    - (B) Common Units, Options or Convertible Securities issued by reason of a Distribution, stock split, split-up or other distribution on Common Units that is covered by Section 1.5, 1.6, 1.7 or 1.8;
    - (C) Common Units actually issued upon conversion of Class C Units or Class D Units, as applicable;
    - (D) Common Units actually issued upon exercise of any Convertible Security, provided such issuance is pursuant to the terms of such Convertible Security and for which adjustment has already been made pursuant to the provisions of this <u>Exhibit G</u> or for which adjustment is not required pursuant to the provisions of this <u>Exhibit G</u>;
    - (E) Common Units issued pursuant to a Qualified Public Offering;
    - (F) Common Units issued in connection with a bona fide business acquisition by the Company, whether by merger, consolidation, sale of assets, sale or exchange of stock or other ownership interests, or otherwise, in each case to the extent approved by the Board by Supermajority Board Vote;
    - (G) Common Units issued pursuant to any equipment leasing arrangement or debt financing arrangement from a bank or similar institution, which arrangement is approved by Supermajority Board Vote and is primarily for non-equity financing purposes; and

- (H) Common Units issued to persons or entities with which the Company has business relationships, provided such issuances are approved by Supermajority Board Vote and are primarily for non-equity financing purposes.
- (b) <u>No Adjustment of Conversion Price</u>. Notwithstanding anything to the contrary in this <u>Exhibit G</u>:
- (i) no adjustment in the Class C Unit Conversion Price shall be made as the result of the issuance or deemed issuance of Additional Issued Common Units if the Company receives written notice from the holders of a majority of the Class C Units then outstanding (it being understood that (A) with respect to an adjustment to Class C Unit Conversion Price which affects Class C-1 Units only, such majority shall not include Class C-2 Units then outstanding and (B) with respect to an adjustment to Class C Unit Conversion Price which affects Class C-2 Units only, such majority shall not include Class C-1 Units then outstanding), agreeing that no such adjustment shall be made as the result of the issuance or deemed issuance of such Additional Issued Common Units. Any such waiver shall bind all future holders of Class C Units.
  - (ii) no adjustment in the Class D Unit Conversion Price shall be made as the result of the issuance or deemed issuance of Additional Issued Common Units if the Company receives written notice from the holders of a majority of the Class D Units then outstanding (it being understood that (A) with respect to an adjustment to Class D Unit Conversion Price which affects Class D-1 Units only, such majority shall not include Class D-2 Units then outstanding and (B) with respect to an adjustment to Class D Unit Conversion Price which affects Class D-2 Units only, such majority shall not include Class D-1 Units then outstanding), agreeing that no such adjustment shall be made as the result of the issuance or deemed issuance of such Additional Issued Common Units. Any such waiver shall bind all future holders of Class D Units.

## (c) <u>Deemed Issue of Additional Issued Common Units.</u>

(i) If the Company at any time or from time to time after the (A) with respect to the Class C Units, Class C Unit Original Issue Date, and (B) with respect to the Class D Units, the Class D Unit Original Issue Date, shall issue any Options or Convertible Securities (excluding Options or Convertible Securities which are themselves Exempted Securities) or shall fix a record date for the determination of holders of any Class of equity securities in the Company entitled to receive any such Options or Convertible Securities, then the maximum number of Common Units (as set forth in the instrument relating thereto, assuming the satisfaction of any conditions to exercisability, convertibility or exchangeability but without regard to any provision contained therein for a subsequent adjustment of such number) issuable upon the exercise of such Options or, in the case of Convertible Securities and Options therefor, the conversion or exchange of such Convertible Securities, shall be deemed to be Additional Issued Common Units issued as of the time of such issue or, if such a record date shall have been fixed, as of the close of business on such record date.

If the terms of any Option or Convertible Security (excluding Options or Convertible Securities which are themselves Exempted Securities), the issuance of which resulted in an adjustment to the Class C Unit Conversion Price or the Class D Unit Conversion Price, as and where applicable, pursuant to the terms of Section 1.4(d), are revised as a result of an amendment to such terms or any other adjustment pursuant to the provisions of such Option or Convertible Security (but excluding automatic adjustments to such terms pursuant to anti-dilution or similar provisions of such Option or Convertible Security) to provide for either (1) any increase or decrease in the number of Units issuable upon the exercise, conversion and/or exchange of any such Option or Convertible Security, or (2) any increase or decrease in the consideration payable to the Company upon such exercise, conversion and/or exchange, then, effective upon such increase or decrease becoming effective, the Class C Unit Conversion Price or the Class D Unit Conversion Price, as applicable, computed upon the original issue of such Option or Convertible Security (or upon the occurrence of a record date with respect thereto) shall be readjusted to the Class C Unit Conversion Price or the Class D Conversion Price, as applicable, which would have been obtained had such revised terms been in effect upon the original date of issuance of such Option or Convertible Security. Notwithstanding the foregoing, no readjustment pursuant to this clause (c) shall have the effect of increasing (A) the Class C Unit Conversion Price to an amount which exceeds the lower of (1) the Class C Unit Conversion Price in effect immediately prior to the original adjustment made as a result of the issuance of such Option or Convertible Security, or (2) the Class C Unit Conversion Price that would have resulted from any issuances of Additional Issued Common Units (other than deemed issuances of Additional Issued Common Units as a result of the issuance of such Option or Convertible Security) between the original adjustment date and such readjustment date, or (B) the Class D Unit Conversion Price to an amount which exceeds the lower of (1) the Class D Unit Conversion Price in effect immediately prior to the original adjustment made as a result of the issuance of such Option or Convertible Security, or (2) the Class D Unit Conversion Price that would have resulted from any issuances of Additional Issued Common Units (other than deemed issuances of Additional Issued Common Units as a result of the issuance of such Option or Convertible Security) between the original adjustment date and such readjustment date.

(iii) (A) If the terms of any Option or Convertible Security (excluding Options or Convertible Securities which are themselves Exempted Securities), the issuance of which did not result in an adjustment to the Class C Unit Conversion Price pursuant to the terms of Section 1.4(d) (either because the consideration per Unit (determined pursuant to Section 1.4(e)) of the Additional Issued Common Units subject thereto was equal to or greater than the Class C Unit Conversion Price then in effect, or because such Option or Convertible Security was issued before the Class C Unit Original Issue Date), are revised after the Class C Unit Original Issue Date as a result of an amendment to such terms or any other adjustment pursuant to the provisions of such Option or Convertible Security (but excluding automatic adjustments to such terms pursuant to anti-dilution or similar provisions of such Option or Convertible Security) to provide for either (1) any increase in the number of Common Units issuable upon the exercise, conversion or exchange of any such Option or Convertible Security, or (2) any decrease in the consideration payable to the Company upon such exercise, conversion

or exchange, then such Option or Convertible Security, as so amended or adjusted, and the Additional Issued Common Units subject thereto (determined in the manner provided in Section 1.4(c)(i)) shall be deemed to have been issued effective upon such increase or decrease becoming effective.

- (B) If the terms of any Option or Convertible Security (excluding Options or Convertible Securities which are themselves Exempted Securities), the issuance of which did not result in an adjustment to the Class D Unit Conversion Price pursuant to the terms of Section 1.4(d) (either because the consideration per Unit (determined pursuant to Section 1.4(e)) of the Additional Issued Common Units subject thereto was equal to or greater than the Class D Unit Conversion Price then in effect, or because such Option or Convertible Security was issued before the Class D Unit Original Issue Date), are revised after the Class D Unit Original Issue Date as a result of an amendment to such terms or any other adjustment pursuant to the provisions of such Option or Convertible Security (but excluding automatic adjustments to such terms pursuant to antidilution or similar provisions of such Option or Convertible Security) to provide for either (1) any increase in the number of Common Units issuable upon the exercise, conversion or exchange of any such Option or Convertible Security, or (2) any decrease in the consideration payable to the Company upon such exercise, conversion or exchange, then such Option or Convertible Security, as so amended or adjusted, and the Additional Issued Common Units subject thereto (determined in the manner provided in Section 1.4(c)(i)) shall be deemed to have been issued effective upon such increase or decrease becoming effective.
- (iv) Upon the expiration or termination of any unexercised Option or unconverted or unexchanged Convertible Security (or portion thereof) which resulted (either upon its original issuance or upon a revision of its terms) in an adjustment to (A) the Class C Unit Conversion Price pursuant to the terms of Section 1.4(d), such Class C Unit Conversion Price shall be readjusted to such Class C Unit Conversion Price as would have obtained had such Option or Convertible Security (or portion thereof) never been issued, or (B) the Class D Unit Conversion Price shall be readjusted to such Class D Unit Conversion Price as would have obtained had such Option or Convertible Security (or portion thereof) never been issued.
- (v) If the number of Units issuable upon the exercise, conversion and/or exchange of any Option or Convertible Security, or the consideration payable to the Company upon such exercise, conversion and/or exchange, is calculable at the time such Option or Convertible Security is issued or amended but is subject to adjustment based upon subsequent events, any adjustment to the Class C Unit Conversion Price or the Class D Conversion Price, as applicable, provided for in this Section 1.4 shall be effected at the time of such issuance or amendment based on such number of Units or amount of consideration without regard to any provisions for subsequent adjustments (and any subsequent adjustments shall be treated as provided in clauses (ii), and (iii) of this Section 1.4(c)). If the number of Units issuable upon the

exercise, conversion and/or exchange of any Option or Convertible Security, or the consideration payable to the Company upon such exercise, conversion and/or exchange, cannot be calculated at all at the time such Option or Convertible Security is issued or amended, any adjustment to the Class C Unit Conversion Price or the Class D Conversion Price, as applicable, that would result under the terms of this Section 1.4 at the time of such issuance or amendment shall instead be effected at the time such number of Units and/or amount of consideration is first calculable (even if subject to subsequent adjustments), assuming for purposes of calculating such adjustment to the Class C Unit Conversion Price or the Class D Conversion Price, as applicable, that such issuance or amendment took place at the time such calculation can first be made; provided, that such adjustment shall in all cases be made (A) with respect to the Class C Conversion Price, prior to the conversion of any Class C Units, and (B) with respect to Class D Conversion Price, prior to the conversion of any Class D Units, based on the good faith determination of the Board (any subsequent adjustment required once the number or amount is actually known shall be treated as provided in clauses (ii), and (iii) of this Section 1.4(c)).

(d) Adjustment Upon Issuance of Additional Issued Common Units. In the event that the Company shall at any time after the Applicable Original Issue Date issue (or is deemed to issue pursuant to Section 1.4(c)) Additional Issued Common Units without consideration or at a price per Additional Issued Common Unit that is less than the Applicable Conversion Price in effect immediately prior to such issue (as adjusted pursuant to Sections 1.5, 1.6, 1.7, or 1.8, as applicable), then the Applicable Conversion Price shall, concurrently with such issue or deemed issue, be reduced to the price (calculated to the nearest one-hundredth of a cent) determined in accordance with the following formula:

$$CP2 = CP1*(A + B) \div (A + C).$$

For purposes of the foregoing formula, the following definitions shall apply:

- (i) "<u>CP2</u>" shall mean the Applicable Conversion Price in effect immediately after such Additional Issued Common Units;
- (ii) "<u>CP1</u>" shall mean the Applicable Conversion Price in effect immediately prior to such issue of Additional Issued Common Units;
- (iii) "A" shall mean the number of Common Units outstanding immediately prior to such issuance or deemed issuance of Additional Issued Common Units (treating for this purpose as outstanding all Common Units issuable upon exercise of Options outstanding immediately prior to such issuance or deemed issuance or upon conversion or exchange of Convertible Securities (including the Class C Units and/or Class D Units, as and where applicable) outstanding (assuming exercise of any outstanding Options therefor) immediately prior to such issue);
- (iv) "B" shall mean the number of Common Units that would have been issued if such Additional Issued Common Units had been issued or deemed issued at a price per

Common Unit equal to CP1 (determined by dividing the aggregate consideration received by the Company in respect of such issue by CP1); and

- (v) "<u>C</u>" shall mean the number of such Additional Issued Common Units issued in such transaction.
  - (vi) "<u>Applicable Conversion Price</u>" means, as and where applicable, the Class C Conversion Price or the Class D Conversion Price.
  - (vii) "<u>Applicable Original Issue Date</u>" means, as and where applicable, the Class C Original Issue Date or the Class D Original Issue Date.
- (e) <u>Determination of Consideration</u>. For purposes of this <u>Section 1.4</u>, the consideration received by the Company for the issuance of any Additional Issued Common Units shall be computed as follows:
  - (i) <u>Cash and Property</u>. Such consideration shall:
- (A) insofar as it consists of cash, be computed at the aggregate amount of cash received by the Company, excluding amounts paid or payable for accrued interest;
- (B) insofar as it consists of property other than cash, be computed at the fair market value thereof, which (1) if such property consists of securities traded on one (1) or more national securities exchanges, shall be the average of the closing price of the security on such exchange, or across the exchanges, over the twenty (20) trading-day period ending three (3) trading days prior to the issuance of the Additional Issued Common Units, and (2) if such property consists of securities traded over-the-counter, shall be the average of the closing bid or sale prices (whichever is applicable) over the twenty (20) trading-day period ending three (3) trading days prior to the issuance of the Additional Issued Common Units, and (3) for all other property, shall be as determined in good faith by Supermajority Board Vote; and
- (C) in the event Additional Issued Common Units are issued together with other securities or other assets of the Company for consideration which covers both, be the proportion of such consideration so received, computed as provided in clauses (A), and (B) above, that is attributable to such Additional Issued Common Units, as (1) provided in the documentation for the applicable issuance, or (2) if not provided as set forth in clause (1) above, as determined in good faith by Supermajority Board Vote.
- (ii) Options and Convertible Securities. The consideration per Additional Issued Common Unit received by the Company for Additional Issued Common Units deemed to have been issued pursuant to Section 1.4, relating to Options and Convertible Securities, shall be determined by dividing:
- (A) the total amount, if any, received or receivable by the Company as consideration for the issue of such Options or Convertible Securities, <u>plus</u> the minimum aggregate amount of

additional consideration (as set forth in the instruments relating thereto, without regard to any provision contained therein for a subsequent adjustment of such consideration) payable to the Company upon the exercise of such Options or the conversion or exchange of such Convertible Securities, or in the case of Options for Convertible Securities, the exercise of such Options for Convertible Securities and the conversion or exchange of such Convertible Securities, by

- (B) the maximum number of Units (as set forth in the instruments relating thereto, without regard to any provision contained therein for a subsequent adjustment of such number) issuable upon the exercise of such Options or the conversion or exchange of such Convertible Securities, or in the case of Options for Convertible Securities, the exercise of such Options for Convertible Securities and the conversion or exchange of such Convertible Securities.
- (f) <u>Multiple Closing Dates</u>. In the event the Company shall issue on more than one (1) date Additional Issued Common Units that are part of one (1) transaction or a series of related transactions and that would result in an adjustment to the Class C Unit Conversion Price and/or the Class D Unit Conversion Price, as applicable, pursuant to the terms of <u>Section 1.4(d)</u>, and such issuance dates occur within a period of no more than ninety (90) calendar days from the first such issuance to the final such issuance, then, upon the final such issuance, the Class C Unit Conversion Price and/or the Class D Unit Conversion Price, as applicable, shall be readjusted to give effect to all such issuances as if they occurred on the date of the first such issuance (and without giving effect to any additional adjustments as a result of any such subsequent issuances within such period).

## 1.5 Adjustment for Stock Splits and Combinations.

- (a) If the Company shall at any time or from time to time after the Class C Unit Original Issue Date effect a subdivision of the outstanding Class C Units, the Class C Unit Conversion Price in effect immediately before that subdivision shall be proportionately decreased so that the number of Units issuable on conversion thereof shall be increased in proportion to such increase in the aggregate number of Units outstanding. If the Company shall at any time or from time to time after the Class C Unit Original Issue Date combine the outstanding Class C Units, the Class C Unit Conversion Price in effect immediately before the combination shall be proportionately increased so that the number of Units issuable on conversion thereof shall be decreased in proportion to such decrease in the aggregate number of Units outstanding. Any adjustment under this section shall become effective at the close of business on the date the subdivision or combination becomes effective.
- (b) If the Company shall at any time or from time to time after the Class D Unit Original Issue Date effect a subdivision of the outstanding Class D Units, the Class D Unit Conversion Price in effect immediately before that subdivision shall be proportionately decreased so that the number of Units issuable on conversion thereof shall be increased in proportion to such increase in the aggregate number of Units outstanding. If the Company shall at any time or from time to time after the Class D Unit Original Issue Date combine the outstanding Class D Units, the Class D Unit Conversion Price in effect immediately before the

combination shall be proportionately increased so that the number of Units issuable on conversion thereof shall be decreased in proportion to such decrease in the aggregate number of Units outstanding. Any adjustment under this section shall become effective at the close of business on the date the subdivision or combination becomes effective.

- 1.6 Adjustment for Certain Distributions. In the event the Company at any time or from time to time (i) with respect to Class C Units after the Class C Unit Original Issue Date, and (ii) with respect to Class D Units after the Class D Unit Original Issue Date, shall (1) make or issue, or (2) fix a record date for the determination of the holders of Common Units entitled to receive, a Distribution payable on the Common Units in Common Units, then and in each such event (x) the Class C Unit Conversion Price in effect immediately before such event shall be decreased as of the time of such issuance or, in the event such a record date shall have been fixed, as of the close of business on such record date, in amount equal to the product of the Class C Unit Conversion Price then in effect, or (y) the Class D Unit Conversion Price in effect immediately before such event shall be decreased as of the time of such issuance or, in the event such a record date shall have been fixed, as of the close of business on such record date, in amount equal to the product of the Class D Unit Conversion Price then in effect, as applicable, in each case multiplied by a fraction:
- (a) the numerator of which shall be the total number of Common Units outstanding immediately prior to the time of such issuance or the close of business on such record date, and
- (b) the denominator of which shall be the total number of Common Units outstanding immediately prior to the time of such issuance or the close of business on such record date, <u>plus</u> the number of Common Units issuable in payment of such Distribution.

Notwithstanding the foregoing, (I) if such record date shall have been fixed and such Distribution is not fully made on the date fixed therefor, the Class C Unit Conversion Price and the Class D Conversion Price shall be recomputed accordingly as of the close of business on such record date and thereafter the Class C Unit Conversion Price and the Class D Conversion Price shall be adjusted pursuant to this subsection as of the time of actual payment of such Distributions, and (II) no such adjustment shall be made (a) with respect to the Class C Conversion Price if the holders of Class C Units simultaneously receive a Distribution of Common Units in a number equal to the number of Common Units on the date of such event, or (b) with respect to the Class D Conversion Price if the holders of Class D Units simultaneously receive a Distribution of Common Units in a number equal to the number of Common Units as they would have received if all outstanding Class D Units had been converted into Common Units on the date of such event.

1.7 <u>Adjustments for Other Distributions</u>. In the event the Company at any time or from time to time (a) with respect to the Class C Units, after the Class C Unit Original Issue Date, and (b) with respect to the Class D Units, after the Class D Unit Original Issue Date shall (i) make or issue, or (ii) fix a record date for the determination of the holders of Common

Units entitled to receive, a Distribution payable in Units or Unit Equivalents of the Company (other than a Distribution of Common Units) or in other property and the provisions of Section 1.6 do not apply to such Distribution, then and in each such event (A) the holders of Class C Units shall receive, simultaneously with the Distribution to the holders of Common Units, a Distribution of such Units, Unit Equivalents or other property in an amount equal to the amount of such Units, Unit Equivalents or other property as they would have received if all outstanding Class C Units had been converted into Common Units on the date of such event and (B) the holders of Class D Units shall receive, simultaneously with the Distribution to the holders of Common Units, a Distribution of such Units, Unit Equivalents or other property in an amount equal to the amount of such Units, Unit Equivalents or other property as they would have received if all outstanding Class D Units had been converted into Common Units on the date of such event.

## 1.8 Adjustment for Merger or Reorganization, etc.

- If there shall occur any reorganization, recapitalization, reclassification, (a) consolidation or Change of Control involving the Company in which the Common Units (but not the Class C Units) are converted into or exchanged for securities, cash or other property, then, following any such reorganization, recapitalization, reclassification, consolidation or Change of Control, as applicable, each Class C Unit shall thereafter be convertible in lieu of the Common Units into which it was convertible prior to such event into the kind and amount of securities, cash or other property which a holder of the number of Common Units of the Company issuable upon conversion of one (1) Class C Unit immediately prior to such reorganization, recapitalization, reclassification, consolidation or Change of Control, as applicable, would have been entitled to receive pursuant to such transaction; and, in such case, appropriate adjustment (as determined in good faith by the Board) shall be made in the application of the provisions in this Exhibit G with respect to the rights and interests thereafter of the holders of the Class C Units, to the end that the provisions set forth in this Exhibit G (including provisions with respect to changes in and other adjustments of the Class C Unit Conversion Price) shall thereafter be applicable, as nearly as reasonably may be, in relation to any securities or other property thereafter deliverable upon the conversion of the Class C Units. This Section 1.8(a) shall fully apply to successive reorganizations, recapitalizations, reclassifications, consolidations or Changes of Control, as applicable, involving the Company in which the Common Units (but not the Class C Units) are converted into or exchanged for securities, cash or other property.
- (b) If there shall occur any reorganization, recapitalization, reclassification, consolidation or Change of Control involving the Company in which the Common Units (but not the Class D Units) are converted into or exchanged for securities, cash or other property, then, following any such reorganization, recapitalization, reclassification, consolidation or Change of Control, as applicable, each Class D Unit shall thereafter be convertible in lieu of the Common Units into which it was convertible prior to such event into the kind and amount of securities, cash or other property which a holder of the number of Common Units of the Company issuable upon conversion of one (1) Class D Unit immediately prior to such reorganization, recapitalization, reclassification, consolidation or Change of Control, as applicable, would have

been entitled to receive pursuant to such transaction; and, in such case, appropriate adjustment (as determined in good faith by the Board) shall be made in the application of the provisions in this Exhibit G with respect to the rights and interests thereafter of the holders of the Class D Units, to the end that the provisions set forth in this Exhibit G (including provisions with respect to changes in and other adjustments of the Class D Unit Conversion Price) shall thereafter be applicable, as nearly as reasonably may be, in relation to any securities or other property thereafter deliverable upon the conversion of the Class D Units. This Section 1.8(b) shall fully apply to successive reorganizations, recapitalizations, reclassifications, consolidations or Changes of Control, as applicable, involving the Company in which the Common Units (but not the Class D Units) are converted into or exchanged for securities, cash or other property.

Certificate as to Adjustments. Upon the occurrence of each adjustment or readjustment of the Class C Unit Conversion Price and/or Class D Unit Conversion Price, as applicable, pursuant to this Exhibit G, the Company at its expense shall, as promptly as reasonably practicable but in any event not later than twenty (20) calendar days thereafter, compute such adjustment or readjustment in accordance with the terms hereof and furnish to each holder of Class C Units and/or Class D Units, as applicable, a certificate setting forth such adjustment or readjustment, including the kind and amount of securities, cash or other property into which the Class C Units and/or Class D Units, as applicable, are convertible and showing in detail the facts upon which such adjustment or readjustment is based. The Company shall, as promptly as reasonably practicable after the written request at any time of any holder of Class C Units and/or Class D Units, as applicable (but in any event not later than twenty (20) calendar days thereafter), furnish or cause to be furnished to such holder a certificate setting forth (a) (i) the Class C Unit Conversion Price then in effect, and (ii) the number of Common Units and the amount, if any, of other securities, cash or property which such holder would receive upon the conversion of the Class C Units held by such holder, as applicable, and/or (b) (i) the Class D Unit Conversion Price then in effect, and (ii) the number of Common Units and the amount, if any, of other securities, cash or property which such holder would receive upon the conversion of the Class D Units held by such holder, as applicable.

## 1.10 <u>Notice of Record Date</u>. In the event:

- (a) the Company shall take a record of the holders of the Common Units (or other securities at the time issuable upon conversion of the Class C Units and Class D Units) for the purpose of entitling or enabling them to receive any Distribution, or to receive any right to subscribe for or purchase any securities of any class or any other securities, or to receive any other security;
- (b) of any capital reorganization of the Company, any reclassification of the Common Units of the Company, or any transaction resulting in a Change of Control involving the Company; or
- (c) of the voluntary or involuntary dissolution, liquidation or windingup of the Company,

then, and in each such case, the Company shall send or cause to be sent to the holders of Class C Units and/or Class D Units a notice specifying, as the case may be, (i) the record date for such Distribution or right, and the amount and character of such Distribution or right, or (ii) the effective date on which such reorganization, reclassification, dissolution, liquidation or winding-up is proposed to take place, and the time, if any is to be fixed, as of which the holders of the Common Units (or such other Units or Unit Equivalents at the time issuable upon the conversion of the Class C Units and Class D Units) shall be entitled to exchange their Common Units (or such other equity securities of the Company or Unit Equivalents) for securities or other property deliverable upon such reorganization, reclassification, Change of Control transaction involving the Company, dissolution, liquidation or winding-up, and the amount per Unit and character of such exchange applicable to the Class C Units and Class D Units, other equity securities of the Company or Unit Equivalents, as applicable.

1.11 <u>Section References</u>. Unless otherwise noted, references to Sections in this Exhibit G shall be to the sections of this Exhibit G.

### **EXHIBIT H- Nomination Waiver**

Effective	e Date:								
_	_	 	_	 	 _		_	_	

In accordance with Section 8.11 of the Eighth Amended and Restated Limited Liability Company Agreement (as may be amended from time to time) (the "Agreement") of MEMX Holdings LLC (the "Company"), the undersigned Member hereby provides [a revocable] [an irrevocable] notice to the Company of its election, effective as of the date set forth above, to waive its right to nominate a Director to the Board. Capitalized terms used but not defined in this election form shall have the meanings set forth in the Agreement.

[Until such election is revoked in accordance with the Agreement,] such Member shall not have the right to nominate a Director to the Board pursuant to <u>Article VIII</u>.

The undersigned Member acknowledges completing this form independently, after consultation with its own advisors to the extent such Member deems necessary.

[Signature Page Follows]

IN WITNESS WHEREOF, the undersigned, intending to be legally bound hereby, have duly executed this Nomination Waiver as of the date set forth below.

	MEMBER:	
	[Insert Name of Member]	
	By: Name: Title:	_
Accepted:		_
MEMX Holdings LLC		
By:		
Name:		
Fitle:		

**EXHIBIT I-1 – MEMX LLC Exchange Director Nomination Rotation** 

Term	<b>Exchange Director Nominating Member</b>
Initial Term	Morgan Stanley
Initial Term	Schwab
Second (2 <sup>nd</sup> ) annual term	Schwab
Second (2 <sup>nd</sup> ) annual term	Virtu
Third (3 <sup>rd</sup> ) annual term	Virtu
Third (3 <sup>rd</sup> ) annual term	Morgan Stanley

The MEMX LLC Exchange Director Nomination Rotation set forth above shall be repeated annually thereafter.

**EXHIBIT I-2 – MX2 LLC Exchange Director Nomination Rotation** 

Term	<b>Exchange Director Nominating Member</b>
Initial term (MX2 LLC Stub Period and the first (1st) annual term)	Schwab
Initial term (MX2 LLC Stub Period and the first (1st) annual term)	Virtu
Second (2 <sup>nd</sup> ) annual term	Virtu
Second (2 <sup>nd</sup> ) annual term	Morgan Stanley
Third (3 <sup>rd</sup> ) annual term	Morgan Stanley
Third (3 <sup>rd</sup> ) annual term	Schwab

The MX2 LLC Exchange Director Nomination Rotation set forth above shall be repeated annually thereafter.